MALE INTIMATE PARTNER ABUSE: DRAWING UPON THREE THEORETICAL PERSPECTIVES

by

Monica Solinas-Saunders

BA in Economics, University of Cagliari, 1997

MA in Sociology, University of Pittsburgh, 2002

Submitted to the Graduate Faculty of

School of Arts and Sciences in partial fulfillment

of the requirements for the degree of

Doctor of Philosophy

University of Pittsburgh

UNIVERSITY OF PITTSBURGH

SCHOOL OF ARTS AND SCIENCES

This dissertation was presented

by

Monica Solinas-Saunders

It was defended on

July 26th, 2007

and approved by

John Marx, PhD, Professor, Department of Sociology

Salvatore Babones, PhD, Assistant Professor, Department of Sociology

Magda Stouthamer-Loeber, PhD, Professor, Western Psychiatric Institute and Clinic

Rolf Loeber, PhD, Professor, Western Psychiatric Institute and Clinic

Dissertation Advisor: Phyllis Coontz, PhD, Associate Professor, Graduate School of Public and International Affairs Copyright © by Monica Solinas-Saunders

MALE INTIMATE PARTNER ABUSE: DRAWING UPON THREE THEORETICAL PERSPECTIVES

Monica Solinas-Saunders, PhD

University of Pittsburgh, 2007

This study addresses controversies in the literature of risk factors of abuse perpetrated by men against female intimate partners. Drawing upon the three theoretical perspectives dominant in the literature (family violence, mainstream feminist, and life-course perspectives), I approach the analysis from two directions. First, each perspective suggests risk factors in a particular category of influence; the family violence perspective suggests a focus on family influences, mainstream feminism suggests socio-cultural influences, and life-course theorists suggest a focus on individual/personal factors. Second, the family violence perspective and the life-course also suggested investigating influences organized by age group. I use a sample of 506 boys from the Pittsburgh Youth Study, a longitudinal study consisting of male individuals from public schools of Pittsburgh, Pennsylvania. More specifically, I focus on the 329 participants in the sample who reported being involved in a romantic relationship by age 23-25, age at which intimate partner abuse was investigated. The findings of the descriptive analysis show that intimate partner abuse is strongly correlated to the variables experience of corporal punishment in the family of origin, attitudes toward women, and delinquency (especially age 19-22). The multivariate analysis confirms that minority males, those who entered the relationship at younger age, those who experienced frequent corporal punishment in the family of origin (age 13-15), and those who had been antisocial during pre-adulthood (age 19-22) were more likely to abuse an intimate female partner during adulthood (age 23-25). The findings on race, age at entering the relationship, and delinquency add new evidence to existing controversies in the empirical literature. In the case of corporal punishment, the results address a gap in the literature.

TABLE OF CONTENTS

1.0 CHAPTER I – INTRODUCTION	1
1.1 Introduction to the dissertation	2
1.2 Organization of the dissertation	6
2.0 HISTORICAL BACKGROUND	9
2.1 Introduction to chapter 2	10
2.2 Early perspectives on abuse against intimates	10
2.3 Violence against women in the family in Colonial America	12
2.4 Early feminist influence: Naming the issue	14
2.5 Policy response to violence against women in the family	15
3.0 CHAPTER III – REVIEW OF THE LITERATURE	18
3.1 Three theoretical perspectives on intimate partner abuse	
in the literature	19
3.1.1 The family violence perspective	19
3.1.2 The mainstream feminist perspective	25
3.1.3 The life-course perspective	30
3.2 Drawing upon the three perspectives: My theoretical model	35
4.0 CHAPTER IV – METHODOLOGY	37
4.1 Introduction to Chapter 4	38
4.2 The Pittsburgh Youth Study data set (PYS)	38
4.2.1 Participants: the "Oldest sample"	39
4.2.2 The interviews	40

4.3 Empirical model	43
4.4 PYS variables employed in the statistical analysis	43
4.5 Summary of the variables employed in the statistical analysis	63
4.6 Hypotheses and plan for the statistical analysis	66
5.0 CHAPTER V - STATISTICAL ANALYSIS	72
5.1 Empirical investigation	73
5.2 Comparing the "current partner-group" to	
the other participants in the sample	73
5.3 Bi-variate associations (between intimate partner abuse	
and each risk factor	78
5.4 Multivariate analysis: Testing hypotheses	81
5.4.1 Part 1: Analysis of risk factors of intimate partner abuse	
by category of influence	82
5.4.2 Part 2: Analysis of risk factors of intimate partner abuse	
by age-group	108
5.4.3 Final model	116
6.0 CHAPTER VI – DISCUSSION	120
6.1 Discussing the findings	121
6.2 Strengths and limitations of the study	129
6.3 Implications for further research	131
	104
APPENDICES	134
APPENDIX A – Supervision/Involvement Questionnaire	135
APPENDIX B – Self-Reported Delinquency Questionnaire	138
APPENDIX D - Bibliography	141

SUMMARY OF TABLES

Table 4.1: Assessment phases for participants of the PYS (N=506)	42
Table 4.2: Distribution of abusive behaviors against intimate partners (N=329)	48
Table 4.3: Prevalence of intimate partner abuse – age 23-25	50
Table 4.4: Distribution of weight among PYS participants	61
Table 4.5: Summary of the variables included in the analysis	65
Table 5.1 Comparison between the "current partner-group" and	
other participants in the sample	75
Table 5.2: Bi-variate associations between the dependent variable	
and each risk factor	79
Table 5.3: Family influence model	84
Table 5.4: Family influence model – Analysis by racial group	88
Table 5.5: Socio-cultural influence model	91
Table 5.6: Socio-cultural influence model – Analysis by racial group	94
Table 5.7: Individual/personal influence model	97
Table 5.8: Individual/personal influence model – Analysis by racial group	100
Table 5.9: Analysis of risk factors of intimate partner abuse by	
category of influence - Logistic regression	103
Table 5.10: Analysis of risk factors of intimate partner abuse by age-group –	
- Logistic regression	111
Table 5.11: Final model – Analysis of risk factors of intimate partner abuse	117

CHAPTER I

INTRODUCTION

1.1 Introduction to the dissertation

In the last ten years, rates of intimate partner abuse have dropped in the United States (BJS, 2006), but the incidence of this social problem is still very high. During the year 2004, there were more than 620,000 non-fatal intimate partner victimizations (BJS, 2006). One third of all these victimizations were serious violent crimes (rapes, sexual assaults, and aggravated assaults) and involved serious injuries (BJS, 2006). The Center for Disease Control and Prevention (2003) estimates that intimate partner abuse causes nearly 2 million injuries and 1,300 deaths nationwide every year.

Victims of intimate partner abuse individually suffer from physical pain, psychological/emotional damage, and social isolation. Additionally, they suffer the economic costs of intimate partner abuse. The Center for Disease Control and Prevention has estimated that the total cost of intimate partner abuse exceeds \$5.8 billion each year, with \$4.1 billion spent for medical and mental health care and about \$1.8 billion in lost productivity (CDC, 2003).

Intimate relationships are supposed to provide love, nurturing, care, support, and protection; however, research has shown that in reality these relationships are not always peaceful and secure (Brush, 1990). In particular, women are at much higher risk to be victimized by intimate partners than by others. Findings from the National Violence against Women Survey (NVAWS) show that 64% of the women who reported experiencing rape, physical assault, and/or stalking since age 18 were victimized by a current or former husband, cohabiting partner, or date (Tjaden and Thoennes, 2000).

Although men and women are both capable of becoming violent/abusive within intimate relationships, males are more likely than females to inflict greater injuries or harm (Brush, 1990; Buzawa and Buzawa, 2003; Straus, 1991; Frieze, 2005). Males are also more likely than females

to kill an intimate partner; in fact, a study conducted by the BJS analyzed data from 1976 to 2002 and found that when the violence occurred among family members or intimate partners, about two thirds of murders were perpetrated by men (BJS, 2005). Research has consistently shown that males and females differ not only in their rates of abuse/violence within intimate contexts, but they also differ in terms of motivations for abusive/violent behavior. In particular, while females, most often, use violence to protect themselves and their children from violent outbursts, males use violence to control others (Miller, 2005).

The purpose of this dissertation is simply to understand why some males are more prone than others to perpetrate abuse against their intimate female partners in adulthood. In other words, this dissertation attempts to investigate specific risk factors of intimate partner abuse that make some males more likely than others to abuse their intimate female partners. For the purpose of this research, intimate partner abuse is defined as the use of physical, sexual, emotional and psychological abuse perpetrated by a man against his female partner.

Although a huge body of literature on violence/aggression within the family and intimate relationships has developed in the last three decades, research on risk factors of intimate partner abuse still shows many contrasting results. Recently, major controversies on the causes of male abusive/violent behaviors against their intimate female partners have been highlighted by theorists of the three dominant perspectives in the literature: the family violence, the mainstream feminist, and the life-course perspectives. By focusing on the complex set of interactions among family members/intimate partners, family violence theorists point out that research is needed to investigate whether interpersonal conflict within the family/intimate relationships, experience of corporal punishment in the family of origin, and a low level of parental bonding during adolescence influence an individual's propensity to perpetrate abuse against intimate partners

(Straus, 1991; Straus and Yodanis, 1996; Mills, 2003). In contrast, mainstream feminists focus on the association between intimate partner abuse and cultural values that justify and legitimize male dominance and control over women and suggest that there is a need for research examining whether having negative attitudes toward women influences males' behaviors within intimate contexts (Dobash and Dobash, 1998). While some research based on field work suggests that having a negative attitude toward women tends to affect a male's likelihood of abusing his female partners, there is little quantitative/empirical work in this area (Johnson and Sigler, 2000). Finally, life-course theorists suggest that risk factors of violence within intimate contexts have to be found within the predispositions of the individual, that is, within his/her personal characteristics and behaviors; more specifically, life-course and developmental theorists point out that research is needed to investigate whether academic competence, delinquency, and number of sexual partners are likely to influence an individual's propensity to abuse intimate partners. The three theoretical perspectives identify three categories of influence: family, sociocultural, and individual/personal influences¹. In addition, all three perspectives identify three demographic variables as risk factors of intimate partner abuse: race, socio-economic status, and age at entering the relationship. Proponents of the three perspectives also agree that alcohol use/abuse should be investigated as a risk factor for intimate partner abuse².

Both the family violence and the life-course perspectives also suggest that it is important to investigate causes of intimate partner abuse over the course of development, taking into consideration different age-groups. Four age-groups (or developmental phases) are taken into consideration for the analysis: *Early adolescence* (age 13-15) is the phase in which an individual seeks his own "identity" and a role within his social network, neighborhood, school, or circle of

¹ In considering different categories of influence I follow Brofenbrenner (1979).

² The three demographic variables (race, socio-economic status, and age at entering the relationship) and the variable alcohol use will be included in the statistical analysis as control variables.

friends. *Late adolescence* (age 16-18) is the phase in which an individual feels and behaves more independently, following hormonal changes and sexual drive. *Pre-adulthood* (age 19-22) is the phase is which an individual attempts to loosen the ties with family members and starts to become more financially independent; *Adulthood* (age 23-25) is the phase in which an individual starts feeling less "self-absorbed", looks for a more steady romantic relationship and attempts to settle down³.

I approach the empirical investigation from two directions. First, I explore whether risk factors considered by category of influence (family, socio-cultural, and individual/personal influences) affect the likelihood of being abusive against an intimate partner in adulthood. Second, I investigate which risk factors organized by age-group (adolescence, late adolescence, pre-adulthood) influence the likelihood of being abusive later in life, during adulthood.

A statistical analysis of secondary data from the Pittsburgh Youth Study (PYS) will be developed for this dissertation. The PYS is an ongoing longitudinal study of the causes and correlates of delinquency, psychopathologies, substance abuse, early sexual behaviors, and intimate partner abuse in urban males. The PYS sampled 3 cohorts from Pittsburgh's Public Schools with the aim of investigating the cohort effect in the participants' development and behaviors. However, this research only focuses on one of the three cohorts, more specifically on a group of 506 boys followed from seventh grade until age 25. Furthermore, only a sub-sample of 329 participants who reported being involved in a romantic relationship by age 23-25, were included in the statistical analysis. The PYS uniquely provides an opportunity to investigate and identify risk factors of violence/abuse by men against their intimate female partner; indeed, this study includes an enormous number of variables reflecting different aspects of their lives. Hence, most of the risk factors suggested by the literature are represented in the data set. Furthermore,

³ The developmental phases were inspired by Erik Erikson's work on stages of development (1994).

because the PYS is a longitudinal study, it also allows study of risk factors of intimate partner abuse by age-group.

This dissertation offers a new approach to the study of risk factors of intimate partner abuse. First, it investigates risk factors that have been the focus of major controversies in the literature, either because they have been neglected in empirical research or because they have so far produced contrasting results. Second, this study offers two different types of analysis of risk factors of intimate partner abuse, one that takes into consideration factors by category of influence (family, socio-cultural , and individual/personal influences), and another that investigates factors over the course of development, by age-group. The findings of this study will be useful to researchers and policy makers interested in the design and implementation of prevention programs that will have a long-term effect in reducing the incidence of intimate partner abuse.

1.2 Organization of the dissertation

This dissertation is organized in six Chapters. Chapter 2 provides a historical overview of intimate partner abuse. Although intimate partner abuse has been recognized as a social problem only in the last two decades, violence against women in the family has a long history. For this reason, the historical antecedents to the problem need to be described in this dissertation. Chapter 3 reviews the literature with emphasis on the controversies around the causes of intimate partner abuse in existing research. The focus here is on three perspectives: family violence, mainstream feminist, and life-course. The family violence perspective suggests the importance of investigating the causes of intimate partner abuse within the family and within intimate contexts. In contrast, the mainstream feminist perspective blames intimate partner abuse on the social

structure and cultural norms that allow men to exercise power and control over women (in society in general, and in intimate relationships in particular). Finally, the life-course perspective suggests that causes and correlates of intimate partner abuse might be found among the predispositions of the individual and his personal characteristics. The three perspectives provide the conceptual framework for the empirical analysis of this study, including family, sociocultural and individual influences. Chapter 4 describes the methodology used for this study. All the variables examined in this research come from the PYS data set. The analysis conducted for this dissertation focuses on the lives of 329 males from the public schools of Pittsburgh, Pennsylvania. The 329 participants are part of a cohort from the PYS (N= 506). The first interviews were conducted in 1987, when the boys were on average 13 years old (7th grade) and the last interviews were conducted in the year 2000, when they were 25 years old on average. Four groups of variables from the PYS data set are employed for the statistical analysis: 1. Experiences in the family of origin and intimate relationships (such as intimate partner abuse, interpersonal conflict with a romantic partner, experience of corporal punishment in the family of origin, and bonding to parents/caretakers during early adolescence); 2. Socio-cultural background (such as attitudes toward women and relationships with women); 3. Individual/personal characteristics and experiences (such as, academic competence, delinquency, and number of sexual partners); and 4. Demographics (such as race, age at beginning of the romantic relationship, and socio-economic status) and alcohol use. A descriptive analysis of these measures is provided in this Chapter. This Chapter also specifies the hypotheses at the multivariate level that are used as a guide for the empirical investigation. Chapter 5 provides a detailed description of the statistical techniques used for the multivariate analysis and presents the findings. Bivariate logistic regressions are used to estimate the participants' likelihood to

become abusive in adulthood based on the risk factors identified in the literature, first considered by category of influence (family, socio-cultural, and individual/personal influences) and then considered by age-group (age 13-15 or early adolescence; age 16-18 or late adolescence; and age 19-22 or early adulthood). Chapter 6 discusses the findings and implications for future research and policy. Furthermore, this Chapter specifies the limitations of the analysis and of the data set.

CHAPTER II

HISTORICAL BACKGROUND

2.1 Introduction to Chapter 2

Although abuse of women in the family became a "visible" social problem only with the Feminist Movement in the 1970s, historians have noted that it has existed for a long time. In previous eras, the issue was considered a private matter and was not included in the public discourse. In this Chapter, I attempt to trace the origins of intimate partner abuse through the lens of the history of violence against women in the family.

2.2 Early perspectives on abuse against intimates

Some historians claim that the origin of males' desire to dominate and control women is rooted in the discovery, in prehistoric times, of the link between sex and procreation (Pagelow, 1984). In particular, the perceived need to limit women's freedom and control their behavior stems from men's anxiety about the ambiguity of paternity, and their desire to ensure that their mates' children are their own (Pagelow, 1984).

In ancient civilizations violence against women in the family was justified by women's inferior legal status vis-à-vis that of men. Men had the authority to rule the house and therefore to discipline their wives. Women had no power within the family and no voice in the public sphere; hence, they were left at the mercy of the husbands (Afoldy, 1987).

The basis for this legal control over women was first formalized in the Roman law of marriage by Romulus (around the 750 B.C.). More specifically, the law stated that women were subject to the ownership of their father who was invested with the paternal power (or *patria potestas*). Fathers were entitled to decide when their daughter had to marry and also had the legal power to choose her husband, even against the young girl's will (Afoldy, 1987). After marriage the husband was given by hand (or *manus*) all the rights on the girl (*puella*), becoming by law

the father of the family (or *pater familias*). Although women had no alternative to marriage, the law specified all the rights that men had over the family and the obligations that women had toward their husbands (Pagelow, 1984; Dobash and Dobash, 1978). With the *patria potestas*, wife and children became the personal property of the man; husbands had the authority to rule their house, which also included the right to rape, beat, torture, or kill their wives (Johnson and Sigler, 2000). The legal control over women also specified that a husband could decide to put his wife to death for drinking his wine or for infidelity (Afoldy, 1987; Pagelow, 1984). However, women were not allowed to turn against their husbands for the same reasons. This double standard was specified in the Roman law as a means to protect the future of the family and the inheritance of the male children (Pagelow, 1984; Dobash and Dobash, 1978).

According to Pagelow (1984), many changes occurred over time in the Roman Empire that improved both the legal and the social statuses of women (at least upper class women). Due to the massive number of deaths among roman soldiers during the Punic and the Civil wars (264-146 B.C.) women gained a new position in society and in their families (Afoldy, 1987). The role of the *mater familias (or matrona)* became stronger and women began participating more actively both socially and intellectually. However, when the Christians gained power (around the 300 A.C.), they demanded the subordination of women to men. Women had to be silent and obedient and they had to accept their husbands' authority (Dobash and Dobash, 1978; Pagelow, 1984). Christian leaders considered wife beating perfectly legitimate (Daniels, 1997). As Buzawa and Buzawa state, "Christianity [...] simply affirmed male-dominated family structures that were already in existence" (2003:57).

In Great Britain, the English Common Law explicitly provided a husband with the right and the responsibility to punish his wife and to keep her in a subservient position. The William

Blackstone's "rule of thumb" stated that, a husband had the right to legally beat his wife with a rod no thicker than his thumb (Daniels, 1995; Johnson and Sigler, 2000; Pleck, 1987; Pagelow, 1984; Dobash and Dobash, 1978). The English Common Law specified that women had no civil rights and that after marriage they became the 'chattel' of their husband (his personal property). Women had no power over their children and in case of parental separation, the father was granted the custody rights of his children (Pagelow, 1984).

2.3 Violence against women in the family in Colonial America

Colonial America inherited principles embedded in European custom and law. The first European settlers carried to the United States their cultural beliefs and rules of social organization, which also applied to the family. The Puritans who escaped persecution in England and moved to Plymouth and Massachusetts Bay were looking for a better life, and the family was considered their all new religious "experiment" (Daniels, 1997). According to Pleck (1987), families were entitled to create a sort of new social order that would have allowed the Puritans to survive in the new world. With the Massachusetts Body of Laws and Liberties, enacted in 1641, the Puritans were the first society to expressly define abuse within the family as an illegal act (Buzawa and Buzawa, 2003; Daniels, 1997; Pleck, 1987).

Throughout the 1700s, many English Common Laws were modified to better fit the conditions of the new world, so that women could have more freedom in the commercial areas (Pagelow, 1984). However, after a new wave of conservatism in 1776 and the American Revolution, women lost many of the privileges that they had previously gained (Pagelow, 1984).

Throughout the 19th century in the US, wife abuse was disciplined differently in different jurisdictions, although it was common throughout the nation to tolerate wife beating as long as

no permanent injury had been inflicted. For instance, in North Carolina the "curtain law" (1873) stated that "if no permanent injury has been inflicted, it is better to draw the curtain, shut out the public gaze, and leave the parties to forget and forgive" (Browne, 1987:167; Johnson and Sigler, 2000:167). Like in England, courts in the US relied on Blackstone's "rule of thumb" (Browne, 1987). The first states to reject the use of this misogynistic practice were Alabama and Massachusetts in 1871 (Pagelow, 1984; Dobash and Dobash, 1979); these were followed by Mississippi, which also rescinded men's privilege to beat their wives; and in Maryland (1882), Delaware (1881), and Oregon (1886) wife beating became punishable at the whipping post (Buzawa and Buzawa, 2003). However, in most court cases involving wife beating, judges were reluctant to blame the violence entirely on the men; in other words, the abuse was often justified and explained by the wife's behavior (Pleck, 1987; Daniels, 1997). If the woman had behaved well, the man would have not been forced to use violence against her (Daniels, 1997).

Traditionally, violence among family members was considered as a private matter and the State was reluctant to intervene because of the legal principles around the public/private sphere. The perception that abuse within the family is not of public interest changed when, at the turn of the century (1900s), concerns about immigrant communities arose in the US. Indeed, racism and classism of European American groups shaped the way wife beating was viewed in society. Some authors speculate that the fear that abuse within the family would make crime rates rise in the communities, made white middle and upper class social leaders to become interested in criminalizing the issue (Daniels, 1997). Buzawa and Buzawa claim that, "when official punishment was deployed, it was far more extensively used against African Americans, immigrants, vagrants, and other groups without political, economic or social power" (2003: 65). Perhaps, economic expansion, growing population, and increasing immigration from other parts

of the world created the need for redefining social order, and the family had to be part of this definition (Daniels, 1997). However, things changed again during the Great Depression, when sympathy toward the poor, unemployed and exhausted men was often used to excuse violence against wives (Daniels, 1997).

2.4 Early feminist influence: Naming the issue

As Pleck (1987) points out, toward the end of the 1800s, several women activists became outspoken on the issue of wife-beating. For instance, Elizabeth Cady Stanton and Susan B. Anthony in 1870 asked for the death penalty for rape. Later, Lucy Stone pushed for the women's right to file for legal separation from their husbands in case of criminal assault; the proposal was presented three times – in 1879, 1883, and 1891 – but it always failed to pass (Daniels, 1997). Apparently, the proposals were seen as 'too radical' for the times (Pleck, 1987; Ferraro, 1989; Daniels, 1997).

It was not until the 1960s-70s that the general public, social service providers and legal actors started to see the violence that many women suffered within the family and intimate relationships as problematic. Consequently, violence against women was added to the public policy agenda as an issue of concern. Out of this, new definitions of violence within the family context emerged: domestic violence, woman battering, spouse abuse or wife abuse, and intimate partner violence. All these terms differ in the way they describe this social problem. For instance, the term "domestic violence" implies that the abuse is perpetrated within a family context and includes violence among family members (violence against a spouse, against a child, or a sibling) excluding all the violence that occurs among dating partners, former spouses, or expartners. The term "woman battering" clearly asserts the gendered nature of violence and does

not take into consideration the violence that men, too, might experience within intimate contexts. Instead, the term "spouse abuse" refers to both male and female victims; however, the utility of this term is limited because it applies only to violence occurring within married couples. Similarly, the term "wife abuse" specifies that women are the victims in intimate relationships, but it applies only to violence that occurs only within the context of marriage. The term "intimate partner violence" is gender neutral; "partner" can refer to a man or a woman. This term presents the advantage of including all types of intimate partners: married couples, unmarried but cohabiting, dating partners, ex-lovers, and ex-spouses (Daniels, 1997; Ferraro, 2001; Buzawa and Buzawa, 2003)⁴.

2.5 Policy response to violence against women in the family

As a result of the feminist movement and other advocacy groups, significant changes have occurred within the Criminal Justice System and new policies have been implemented. Major changes were first adopted by police departments. Throughout the 70s family violence intervention programs were introduced in several cities. With these programs, officers were trained to deal with intimate partner abuse calls in the manner of social workers; they were asked to mediate the dispute between the parties, to recommend counseling, or intervene to separate the parties. Thus, trained police officers were discouraged from treating domestic abuse from a law enforcement perspective, and arrest was considered only as last resort (Buzawa and Buzawa, 2003). However, the family specialized units within police departments became less and less popular during the 1980s because of the pressure of advocates and politicians who called for the criminalization of intimate partner abuse. And, in 1984 the U.S. Attorney General's Task Force

⁴ For the purpose of this dissertation the term "intimate partner abuse" will refer to all acts of physical, verbal, sexual, and psychological abuse exercised by a man against his female partner. All the behaviors included in the scale used to assess intimate partner abuse are specified in Chapter 4.

on Family Violence mandated that violence between intimate partners should be considered a crime, and therefore arrest should be considered the best response to it (Ferraro, 1989).

At the beginning of the 1970s, a series of new statutory changes were introduced at both the State and the Federal level. However, it was not until the 1990s that the most important piece of legislation on the issue of intimate partner abuse/family violence was approved by the Federal Government: the Violence Against Women Act (VAWA), 1994, which was implemented as part of the Violence Crime Control and Law Enforcement Act. This legislation was designed to standardize the control of violent crimes perpetrated against women and children (outside the family and within family/intimate contexts). This document addressed a spectrum of related issues, including programs that would prevent the occurrence of intimate partner abuse and child abuse and programs designed for prosecuting the abusers. Besides funding more than 4,000 shelters for women and children who escape abusive situations, the VAWA has also established many rehabilitation programs for abusive men. Unfortunately, not all local governments take advantage of this opportunity, and, in many counties throughout the US, rehabilitation programs remain unavailable. Model legislation is perhaps provided by the State of Ohio, where next to the 22 shelter service programs for abused women and children, 27 rehabilitation programs for abusive men are offered. In Ohio 78% of these programs are funded by state and federal governments (http://www.odvn.org). Like women's shelters, batterers' programs struggled to survive throughout the country prior to the financial and political support provided by the VAWA legislation. Batterers' programs focus on the idea that many violent men can be rehabilitated; indeed, many of them are remorseful and they do want to change (Austin and Dankwort, 1998). When these programs were first introduced in the late 70s, the public expressed great skepticism about both the effectiveness of rehabilitating this group of offenders

and the increased safety for oppressed battered women (Austin and Dankwort, 1998). Today, it is still difficult to assess the effectiveness of these programs because of the large number of criteria used to evaluate them; however, many are the documented cases in which women and children have benefited as a result of abusers' participation in rehabilitation programs (Edleson, 1995). Unfortunately, the vast majority of batterers' participation in rehabilitation programs is still court-mandated (Healey et al., 1998); there are reasons to believe that if the initiative to change does not come from the batterer himself, most efforts are destined to fail. The VAWA was re-authorized in 1999 and more recently in 2006.

The next Chapter focuses on the three dominant perspectives of intimate partner abuse in the literature and on the major controversies around the analysis of risk factors of intimate partner abuse in research.

CHAPTER III

REVIEW OF THE LITERATURE

3.1. Three theoretical perspectives of intimate partner abuse in the literature

This study draws upon three different theoretical perspectives: the family violence, the mainstream feminist, and the life-course perspectives. In this Chapter I will review each perspective and discuss how I use the three perspectives to develop a model for examining why some males use violence against their intimate partners while other males do not.

3.1.1 The family violence perspective

Among the three perspectives, the family violence perspective is the oldest to study intimate partner abuse. Its proponents are responsible for coining the term "family violence" and for drawing attention to the problem. In the family violence perspective, violence between intimates is defined as a system of interactions in which both parties are involved in abusive behaviors (Mills, 2003). This perspective identifies dynamics of intimate relations in which violence triggers violence in a cyclical fashion. Thus, abuse is explained within the functioning or "dysfunctioning" of the family; the family as a whole is to be blamed for the abuse and not the individual. The social isolation that the family experiences – as a private institution – creates more opportunities for violent interactions (Healey et al., 1998).

One of the major contributions of the family violence approach is that it has pioneered the use of survey methods in a field that was dominated by qualitative research (Brush, 1990). In 1975, Murray A. Straus and Richard J. Gelles designed and conducted the National Family Violence Survey (NFVS). The study used a stratified sample of 6,002 families in which two racial groups (African Americans and Hispanics) were over-sampled in order to allow researchers to analyze racial and ethnic differences. Interviews were conducted by using the computer assisted telephone interviewing system (CATI). Respondents were asked about both

victimization and perpetration of violence/aggression, that is, either physical or verbal violence. In particular, participants were interviewed about experiences of violence within the last year and throughout their life-span.

Reporting of the NFVS findings led to major theoretical/academic disputes around the issue of gender-symmetry in perpetration rates of intimate partner abuse. As a matter of fact, this study's findings pointed out that, in the family and in intimate relationships, women and men perpetrate abuse and violence at very similar rates. In particular, in the survey, 27% of men and 24% of women reported some form of abuse by their spouse at some point during their marriage (Straus, Gelles, and Steinmetz, 1980).

In 1985 the study was repeated with the main objective of comparing rates of violence within the family to the findings of the 1975 version. With this second survey researchers have found that, although rates of intimate partner abuse are similar between females and males, men are more likely than women to perpetrate multiple violent acts during a single attack; as a consequence, women tend to be injured more often than men during incidents of intimate partner abuse (perpetrated by men) (Stets and Straus, 1990).

The measure of violence used for the NFVS is known as Conflict Tactic Scale (CTS). This measure focuses on violent acts, such as throwing something at spouse, pushing, grabbing, shoving, slapping, kicking, biting or hitting spouse with a fist, hitting or trying to hit spouse with an object, beating up spouse, threatening spouse with a knife or gun, and using knife or gun against spouse. In 1985 this instrument was revised (CTS2) and questions about the injuries suffered by the victims as result of the violence were added to the questionnaire; participants were also asked whether the victim sought and received medical attention. Although the CTS2 offered researchers the opportunity to correct some of the limitations of the original NFVS

instrument used to assess intimate partner abuse, most scholars who have analyzed data from this survey have continued using the original version (Frieze, 2005).

In the last two decades, family violence researchers have focused on major controversies around the causes of family violence. Thus, they have identified several risk factors of intimate partner abuse that need to be further analyzed and they have provided suggestions to continue and improve research on abuse/violence within intimate contexts (Straus and Yodanis, 1996; Straus, 1991; 2004; Mills, 2003). One of the risk factors suggested by the family violence perspective is *interpersonal conflict*. Although family violence theorists assume that intimate partner abuse happens within a context of disagreement (as specified in the measure of violence applied in their study, the CTS), very few studies have analyzed the role of conflict and disagreement in the relationship in association with intimate partner abuse (Straus and Yodanis, 1996). One can think that the more partners argue and disagree with one another, the more likely they are to use physical or verbal force to resolve their conflict. However, the association between interpersonal conflict and intimate partner abuse is not at all clear in the family violence literature. Couples who argue a lot might not necessarily engage in abusive behaviors.

Drawing on social learning theory, family violence researchers have pointed to a second risk factor: a potential association between *experience of corporal punishment* in the family of origin and the likelihood to develop later aggressive behavior against an intimate partner (Straus, 1991; 2005; Straus and Yodanis, 1996; Mills, 2003). Since the family of origin is the place in which socialization begins, norms and rules of behavior are primarily copied from family members. Thus, many children who grow up in violent families learn that violence is part of family interactions, and that violence within the domestic walls is a private matter, and therefore it is not punished (Simons et al., 1998; Straus, 1991; Mills, 2003). Furthermore, violence

perpetrated by parents provides children with a sort of "scripted pattern of behavior" that explains interpersonal relations (Straus and Hotaling, 1980) and shows that violence is an effective means to control others (Simons et al., 1997). The real problem is that violence perpetrated in the family tends to repeat itself within an inter-gender and intergenerational system of dynamics (Mills, 2003). When a mother hits her child, the same child comes to believe that by using violence he can actually control other people's behavior. Straus and Yodanis suggests that individuals who were frequently punished as children by their own parents will (as adults) have "less skills in managing conflict and, therefore, have more unresolved conflicts with their spouses" (1996:828).

One of the major difficulties in analyzing the link between corporal punishment and adult intimate partner abuse is that most of the available studies are cross-sectional. Retrospective methods of investigation employed in cross-sectional studies represent a limitation in empirical research. Indeed, information on past experiences and behaviors might not be reported correctly, due to forgetfulness or reinterpretation of facts. According to Straus (2004), the association between the two should be explored by using prospective data.

Recent theoretical work by Straus and other family violence researchers has pointed out that research on intimate partner abuse should also investigate the roles of one's *parental bonding during* adolescence as another potential risk factor (Straus, 2004; Mills, 2003). More specifically, borrowing from Hirschi's theory of Social Bonds (Hirschi, 1969), family violence researchers suggest that it is important to investigate whether a lack of attachment and a weakened bonding to one's parents during childhood and adolescence increases an individual's risk of becoming involved in abusive relationships with intimates later in life (Straus, 2004; Mills, 2003).

Family violence research has demonstrated that there is a substantial association between drinking and violence in the family and within intimate contexts (Gelles, 1974; Coleman and Straus, 1983; Kaufman Kantor and Straus, 1987). However, some scholars argue that "alcohol per se is not a necessary or sufficient cause of violence" (Gelles and Cavanaugh, 2005:180). The association between *alcohol use* and violence might be mediated by social factors like for instance socio-economic status, culture, attitudes toward violence, and the perception that alcohol triggers violence (Gelles and Cavanaugh, 2005).

Family violence researchers have also suggested several demographic risk factors, one of which is the individual's *socio-economic status*. Although family violence studies have found that lower income families are at higher risk for becoming involved in abusive relationships (Straus and Hotaling, 1980), Mills (2003) suggests that intimate partner abuse exists across all social classes and that the association between socio-economic status and experience of intimate partner abuse needs further analysis.

Some family violence theorists have suggested that <u>age</u> might be associated with intimate partner abuse. Mills (2003) points out that there is evidence that younger couples tend to be at higher risk for intimate partner abuse.

In their studies, family violence researchers have found no racial differences in perpetration rates of intimate partner abuse. However, Straus (1991) points out that the relationship between <u>race</u> and intimate partner abuse is very complex and therefore needs further examination in research.

In spite of all the contributions offered by the family violence perspective, there are several limitations of this theoretical approach to the study of intimate partner abuse. The major limitation is related to the definition of violence between intimates itself. For example, Straus

and his collaborators have primarily focused on the traditional notion of the family. This leaves out other kinds of families and relationships, such as single parents and their children, homosexual couples, dating partners, former partners, and casual partners.

Other limitations are related to the CTS used to assess the violence/abuse; in particular critics have pointed out the following shortcomings. According to Brush (1990), questions on intimate partner abuse in the CTS are asked only within a context of conflict or disagreement, even though it is known that a lot of violence does not rise from disagreements. Another limitation of the CTS specified in the literature is that it does not include sexual violence (Brush, 1990; Kimmel, 2002; Frieze, 2005), nor does it consider forms of violence and control like vandalism against property and violence against pets (Brush, 1990). The CTS only includes violence by current spouses or cohabiting partners, but does not include violence by former spouses or dating partners (Kimmel, 2002). Another problem with the CTS is that outcomes of violent behaviors are not included in the questions (Brush, 1990). The CTS questions are rather short and – typical of survey methods – they do not allow the interviewee to provide detailed information about the situation(s) in which intimate partner abuse occurred (Brush, 1990; Frieze, 2005). Also, the CTS questions ignore details about who initiated the violence and whether the violence was actually perpetrated in self-defense (Brush, 1990). And, according to Browne (1993), the CTS does not include questions about the abuser's intention to hurt. Dobash and Dobash (1998) also indicate that the CTS does not allow the researcher to examine instances of sexual harassment or partner stalking: these non-criminal acts are also important to our understanding of the type of situations in which violence/abuse has occurred. Finally, the results produced by the CTS differ completely from the results of clinical samples (such as shelter samples and emergency room samples) (Dobash and Dobash, 1992).

3.1.2 The mainstream feminist perspective

The feminist movement played an important role in bringing attention to family violence by highlighting the problems of battered women. Out of this grew what is now referred to as the Mainstream Feminist perspective. This refers to violence between intimate partners as a gendered issue that affects mostly women. This perspective attributes intimate partner abuse to differences in power and authority between males and females (Healey et al., 1998). In our patriarchal social systems, men learn to be dominant and women to be subordinate (Pence and Paymar, 1993). Feminists argue that the major cause of intimate partner abuse is patriarchy. "Patriarchy is a system of social stratification, which means that it uses a wide array of social control policies and practices to ratify male power and to keep girls and women subordinate to men" (Chesney-Lind, 2006:9). In patriarchal systems women's oppression is perpetuated because male violence against women is accepted, reproduced within everyday interactions and, therefore, institutionalized. Although the expression of masculinity as power and control over others is common to all men, differences among all individuals are to be found within the environment in which they develop social ties and seek the approval of others (Dobash and Dobash, 1998).

In contrast to the family violence perspective, which considers intimate partner abuse as a family dynamic problem, mainstream feminist scholars focus on intimate partner abuse as a gendered social problem. In particular the feminist approach shows that when women are involved, they are almost never the primary aggressor. Instead, women get involved in violence as a defense mechanism (Browne, 1987; Raphael, 2000; Miller, 2005). For instance, in a recent study on female offenders arrested for intimate partner abuse, Susan Miller (2005) points out that in society at large the "general violent woman" is an anomaly. The author identifies three groups

of female abusers: 1. Those who act in self defense to protect themselves and their children from violent partners (65% of the women in the sample). 2. Those who become aggressive out of frustration as a consequence of abuse suffered in the past or because of disagreement over delicate issues, such as, child custody, divorce, infidelity, and economic constraints (30% of the women who participated to Miller's work); and 3. Those who could be seen as generally violent – although, they are in most cases affected by the use of heavy drugs and abuse of alcohol (5% of the sample in Miller's study). Similar estimates were provided by Browne in an earlier study on women convicted for murdering their abusers (Browne, 1987).

Data from community samples confirm that women are, by far, the most vulnerable partners in intimate relationships in our society, representing 90% to 95% of all the victims of assault within intimate contexts reported to the Criminal Justice System (Dobash, Dobash, Wilson, and Daly, 2005). Estimates on intimate partner abuse from the National Crime Victimization Survey (NCVS) show that in 1998, about one million violent crimes were committed against persons by their current or former spouses, boyfriends, or girlfriends in the US (Rennison and Welchans, 2001). Among the victims, about 876,340 (85%) were women. Between the 1993 and 1998, violence against intimate partners made up 22% of the total violent crimes against women and 3% of the violent crimes against men (Rennison and Welchans, 2001). Findings from the Bureau of Justice Statistics indicate that, in 1992, about 28% of female victims of homicide (1,414 women) were killed by their husbands, ex-husbands, or boyfriends (Bachman and Saltzman 1995). In contrast, about 3% of male victims of homicide (637 men) were killed by their wife, ex-wife, or girlfriend. Also, estimates from the NVAWS (Tjaden and Thoennes, 2000) point out that 7.7% of women and .3% of men reported being raped by a current or former partner at some point during their life; 22.1% of women and 7.4% of men in

the survey reported being physically assaulted by an intimate partner during their life-time; and .5% of women and .2% of men in the survey reported being stalked at some point by an intimate partner or a former partner. In summary, the NVAWS indicates that women are at higher risk of victimization than men when violence within intimate contexts is explored. Indeed, in the study, women were significantly more likely to be raped, physically assaulted, or stalked by a current partner or former partner during their life-time (and also during the 12 months prior to the interviews) than their male counterparts. In addition, women were more likely than men to be injured during an assault perpetrated by an intimate partner (39% of women reported being injured during the last partner rape their experienced compared to 16.1% of men) (Tjaden and Thoennes, 1998, 2000).

Like the family violence researchers, feminist researchers have identified several risk factors of intimate partner abuse. First, they have pointed out that it is important to investigate *men's attitudes toward women and relationships with women*. Drawing upon social control theory (Hirschi, 1969), the mainstream feminist perspective assumes the existence of a system of common moral values – accepted by all members of society (in particular by men) – which are then internalized into a system of beliefs. It is through these moral values that males learn from young age what is the role of women in society, how to interact with women, and how to manage intimacy with them. However, within a social system, not all individuals behave the same and it is difficult to imagine that all men within a society view intimate partner abuse in the same way (Dobash and Dobash, 1998). It is in interaction with other factors (such as experiences in one's social network) that we need to consider the role of culture and the influence that social values have on the individual. When a boy witnesses violence against women in the family (or within

his social network), he perceives that women are subordinate to men, have fewer rights, and, therefore, intimate partner abuse is a normal part of inter-gender relationships (Dobash and Dobash, 1998). Unfortunately, it is difficult to operationalize this concept in quantitative/empirical research (Johnson and Sigler, 2000). The main problem is that existing data do not allow researchers to measure cultural values and beliefs.

Feminist literature also suggests that intimate partner abuse can be triggered by the *use/abuse of alcohol* (Dobash and Dobash, 1998; Raphael, 2000). For some men, drinking might be a way to escape the emotional burden of social problems like unemployment, discrimination, and exposure to violence (for instance, witnessing violence in one's family of origin) (Raphael, 2000). However, the use of alcohol is also part of many individuals' recreational activity, which is more accepted and encouraged in certain social networks. This form of recreation most often excludes female partners and keeps men away from the family, even when they are needed (Dobash and Dobash, 1998). Their constant attendance to drinking groups tends to increase the level of disagreement with their partner, often related to issues of jealousy and infidelity (Dobash and Dobash, 1998).

Proponents of the feminist theory have pointed out that studies on intimate partner abuse should also take into consideration the influence of demographic factors like race and class (Frieze, 2005; Raphael, 2000; Sokoloff and Pratt, 2005).

In the analysis of <u>race</u> as a risk factor of intimate partner abuse, empirical work has provided us with contrasting results. For instance, estimates from the NCVS show that while intimate partner abuse occurs across all racial categories, it is more prevalent among African American couples. According to Rennison and Welchans (2000), between 1993 and 1998, African American women, and were victimized by intimate partners at a rate 35% higher than

that of white women, and 2.5 times higher than that of women from other racial groups. For men the racial gap is even bigger, with African American men experiencing intimate partner abuse at a rate 65% higher than that of white men and 2.5 times higher compared to the rate of men in other racial groups. Using the NCVS data, for the years 1992 and 1993, Bachman and Saltzman (1995) found that race is not an important factor in the analysis of intimate partner abuse.

According to feminists, race and class issues together often create a nexus of opportunities for violence and degradation (Raphael, 2000; Sokoloff and Pratt, 2005). Empirical research is needed to investigate how race and *socio-economic status* affect one's likelihood to become abusive in intimate relationships. Using qualitative methods of research, mainstream feminist scholars have pointed out that African American men from low-income disadvantaged communities tend to be at higher risk of perpetration of abuse, because they seek control over someone within intimate relationships; their frustration is driven by their lack of achievement in a society that marginalizes them and makes them feel responsible for their inability to be the breadwinners (Raphael, 2000; Carbado, 1999).

A final demographic factor suggested by mainstream feminist scholars is <u>age at entering</u> <u>the relationship</u>. Although a few studies have reported that younger people tend to engage in abusive behaviors more often than older people (Frieze, 2005; Rennison, 2001), research is needed to further understand this association.

The mainstream feminist perspective has been criticized by proponents of the family violence perspective for several limitations, one of which is that studies of intimate partner abuse focus exclusively on violence perpetrated by males, mostly against female victims. The literature is rich with examples of "intellectual attacks" against mainstream feminist scholars, who are often accused of writing within an advocacy framework rather than within a scientific mode

(Straus, 1991), ignoring empirical evidence and neglecting the fact that abuse can become mutual within everyday interactions (Dutton and Nicholls, 2005).

3.1.3 The life-course perspective

In contrast to both the family violence perspective and mainstream feminism, the life course perspective studies behavior over time. The life-course perspective draws upon developmental theories in order to examine how psychopathologies, conduct problems and delinquent/antisocial behaviors tend to develop, change or stabilize, and desist over time. The focus is on change and stability in the expression of certain behaviors from childhood to adulthood. This perspective refers to intimate partner abuse as a gender neutral phenomenon and it focuses on the idea that causes of aggression and violence perpetrated against intimate partners must be investigated within the predispositions of the individual. More precisely, the life-course states that behavioral traits and features of the individual that manifest early in life (during childhood/adolescence) might associate to a certain tendency to become abusive against intimates during adulthood.

The life-course uses prospective measures of investigation within a framework of analysis provided by longitudinal studies. An example of a longitudinal study of the analysis of precursors of intimate partner abuse is the work by Moffitt and Caspi (1999). Their research was drawn from a longitudinal exploratory work that has been studied for over 20 years as part of the Dunedin Multidisciplinary Health and Development Study. Participants to the Dunedin study were interviewed about intimate partner abuse when they reached age 21 years. The instrument used to assess the occurrence of violence/abuse between intimate partners combined items from the CTS and the Domestic Conflict Scale identified by Margolin in 1990. In this research Moffitt

and Caspi found that perpetration rates of abuse were very similar between men and women; in particular, among the participants of the Dunedin study, 27% of the females and 34% of the males reported being physically abused by their partners. About 37% of women and 22% of men said that they had been abusive against an intimate partner. These findings show similarities between female and male participants both in the victimization and perpetration rates of intimate partner abuse. In this study, female victims were 10 times more likely than other female participants to become abusive against their male partners, and male perpetrators were 19 times more likely to be abused than other participants. Additionally, the authors have found that the two major risk factors of intimate partner violence among the participants were mental health problems and adolescent delinquency.

Another life-course study on intimate partner abuse is the research of Gorman-Smith et al. (2001). Using a sample of 141 African Americans and Latino males from the Chicago Youth Development Study (CYDS), the authors found an association between street crime during adolescence and intimate partner abuse. Similar issues were addressed in a longitudinal study conducted by Woodward et al. (2000). Using a sample of 201 males and 294 females from the Christchurch Health and Development Study (CHDS) conducted in Christchurch, New Zealand, the authors examined the association between adolescent onset of delinquent behaviors and adult perpetration and victimization of intimate partner abuse. Participants were interviewed about their experience of intimate partner abuse when they reached approximately age 21 years. Three groups of individuals were compared in the study: those with childhood onset delinquent/antisocial behavior problems; those with adolescent limited delinquent/antisocial behavior problems; and those who did not have a history of delinquent/antisocial behavior

problems. The study found an association between very early onset in delinquent/antisocial behavior and adult intimate partner abuse.

The life-course perspective also suggests that there might be an association between <u>number of sexual partners</u> and intimate partner abuse. In a 1994 study conducted in the United Kingdom, Farrington found that individuals at higher risk of partner abuse tended to be sexually promiscuous. Research is needed to understand the role that sexual number of sexual partners plays in explaining the nature and the causes of adult intimate partner abuse.

Although life-course studies have not found racial differences in the analysis of intimate partner abuse, it is important to further investigate whether <u>race</u> is associated to partner abuse (Moffit and Caspi, 1999; Gorman-Smith et al., 2000; Woodward et al., 2002)

Similarly to other theorists, proponents of the life-course perspective have pointed out that a risk factor for intimate partner abuse is one's *socio-economic status*. According to Magdol et al. (1997) individuals involved in abusive relationships were more likely to be unemployed. And, differently from other researchers, life-course theorists indicate that poverty in the family of origin can be a risk factor for abusive behaviors against intimates later in life (Fagan and Wexler, 1987; Farrington, 1994). However, one can speculate that coming from a poor family also makes an individual at higher risk of being poor himself during adulthood.

Life-course researchers have also found that there is an association between <u>age</u> and intimate partner abuse (Cate et al., 1982; Gorman-Smith et al., 2001); however, differently from proponents of the family violence perspective and the mainstream feminist perspective, lifecourse theorists have pointed out that younger couples tend to be less committed toward one another and therefore they are at a lower risk of becoming violent than older couples (Cate et al.,

1982; Gorman-Smith et al., 2001). Studies have indicated contrasting results and the association between age and intimate partner abuse still needs to be investigated.

Similarly to family violence researchers and mainstream feminists, life-course theorists have pointed out that there is a need for research to further clarify the role of *alcohol abuse* to explain the causes of intimate partner abuse (Magdol et al. 1997).

Like the other perspectives, the life-course perspective has controversies and limitations. The analysis of *delinquent/antisocial behavior* as a risk factor/precursor for adult intimate partner abuse is the focus of major controversies in the life-course literature. For instance, Gorman-Smith et al. point out that although their study's findings show that individuals who reported partner violence showed a propensity to be involved also in street violence during adolescence, "there is not complete overlap between these two types of violence, and it may be useful to gain a better understanding of the potential differences in risk between the two" (2001:290). Also, this study's findings might be affected by the age of the participants, who were between the age of 15 and 19 years when interviewed about intimate partner abuse. Also, the sample is primarily composed of minority groups from a poor urban community of Chicago.

Although Moffit and Caspi (1999) have found a link between delinquent/antisocial behavior at age 15 and adult intimate partner abuse, they also suggest that the association needs further investigation. While Woodward et al. (2002) have pointed out that very early onset in delinquent/antisocial behavior is an important precursor of adult abuse between intimates, participants to the study were interviewed on intimate partner abuse only at age 21, and abuse within intimate contexts might be still part of a general tendency in antisocial behavior rather than a separate problem. Furthermore, all these studies have focused on the association between intimate partner abuse and delinquent/antisocial behavior in the form of violent behavior. By so

doing, researchers have attempted to make a connection between behaviors that manifest at different ages in one's life but that might be of the same type (violent). Conversely, research is needed to investigate the association between delinquent/antisocial behavior (both violent and non-violent) and adult perpetration of intimate partner abuse. Further research is necessary to investigate the association between delinquent/antisocial behavior over the course of development and adult intimate partner abuse.

Another controversial issue on the analysis of risk factors of intimate partner abuse discussed in the life-course literature refers to the link between *low academic competence* and intimate partner abuse. Very few longitudinal studies have focused on this link. Farrington and West (1990) and Farrington (1994; 1997) found that a low academic competence from childhood to adulthood is strongly associated to perpetration of abuse and violence against intimates. From a psychological point of view one can think that individuals who are low in academic competence also lack in verbal skills, which might trigger aggression against a partner during a disagreement. Also, we may think that less competent/intelligent individuals tend to underestimate the consequences of abuse.

A limitation of the life-course perspective is that in the analysis of precursors of intimate partner abuse, this perspective focuses on the characteristics of the individuals and their own behaviors. However, while individual characteristics or individual behaviors by themselves, taken out of context, are relevant, it is important to take into consideration environmental factors and situational/contextual factors as well (Fagan et al., 1981).

3.2 Drawing upon the three perspectives: My theoretical model

While the research in each perspective has been important to understanding the problem of abuse within intimate contexts, none of the three perspectives provides the breadth of evidence needed to identify what leads to a pattern of intimate partner abuse. However, in light of their studies' findings, theorists and researchers that have embraced the three perspectives have highlighted major controversies on the causes of abuse perpetrated by men against their female intimate partners. I draw upon the three perspectives in order to construct a theoretical framework of analysis to investigate why some males become abusive against their intimate female partners during adulthood while others do not.

As suggested by the family violence perspective, I investigate whether certain family factors have an influence on a male's propensity to abuse an intimate partner; these factors are: interpersonal conflict with intimate partner, experience of corporal punishment in the family of origin, and a low level of parental bonding during adolescence. By following feminist research, I investigate the influence that attitudes toward women and relationships with women have on intimate partner abuse. And, as life-course researchers have suggested, I analyze the influence of a low academic competence, delinquency, and number of sexual partners on intimate partner abuse.

Furthermore, following the suggestions provided by family violence and life-course researchers, I investigate the influence that certain factors that manifest early in life (experience of corporal punishment, a low level of parental bonding, delinquency and number of sexual partners) have on adult abusive behaviors. Thus, I identify four different age-groups based on the individual's developmental phases⁵ (age 13-15 or early adolescence; age 16-18 or late

⁵ The 4 developmental phases used for the analysis were specified in Chapter 1.

adolescence; age 19-22 or young adulthood; age 23-25 or adulthood). Also following the three perspectives, I investigate whether demographic factors (race, socio-economic status, and age at the beginning of the romantic relationship) and alcohol use, have an influence on adult abusive behaviors perpetrated against intimate partners.

In my theoretical model, intimate partner abuse is the dependent variable and it is defined as the use of any physical, sexual, verbal and psychological abuse perpetrated by a man against his female intimate partner (either wife or girlfriend). The explanatory variables are interpersonal conflict with an intimate partner, experience of corporal punishment in the family of origin, a low parental bonding during adolescence, attitudes toward women and relationships with women, academic competence, delinquency, and number of female sexual partners. The control variables are race, socio-economic status, age at entering the romantic relationship, and alcohol use.

The next Chapter discusses the methodology; I will provide a detailed description of the sample used to investigate the abuse perpetrated by males against their female intimate partners. After describing the variables employed in the statistical analysis I will also specify the hypotheses that will guide the statistical analysis.

CHAPTER IV

METHODOLOGY

4.1 Introduction to Chapter 4

This research uses data from the Pittsburgh Youth Study (PYS). In this Chapter I will describe the sample used for the statistical analysis. Then, I will discuss my empirical model and the measures from the PYS data set. A descriptive analysis of each variable employed in the analysis will be provided. Finally, the hypotheses that guide this empirical investigation will be specified.

4.2 The Pittsburgh Youth Study data set (PYS)

For the analysis I use data from the Pittsburgh Youth Study (PYS), an ongoing longitudinal study of the causes and correlates of delinquency, psychopathologies, substance abuse, early sexual behavior, and intimate partner abuse in young urban males. The study begun in 1987; it sampled three cohorts from the Pittsburgh Public Schools in order to investigate "cohort effects" in participants' development and behaviors (Loeber et al., 1998). The three cohorts consisted of a group of children from first grade, a group from fourth grade, and a group from seventh grade. The first graders were then named as "Youngest Sample"; the fourth graders were named as "Middle Sample" and the seventh graders were named as "Oldest Sample". It is on the "Oldest Sample" that I focus for this dissertation. More specifically, the multivariate statistical analysis will use only the group of 329 males that reported being involved in a romantic relationship by age 23-25, the age at which intimate partner abuse was investigated in the PYS. However, because the 329 participants are a sub-sample of the "Oldest Sample" selected by the PYS investigators and researchers, I will provide the reader with a complete description of the original "Oldest sample" selection (section 4.2.1).

4.2.1 Participants: the "Oldest Sample"

The oldest sample comprises a total number of 506 participants. However, the sampling process began by contacting 1.011 families randomly selected from a list of male students attending the seventh grade of any public school of the Pittsburgh metropolitan area. In total, 84.7% of the families initially contacted agreed to participate (856 children were then counted in the initial screening). The first assessment allowed the identification of children at "high risk" of delinquency based on a number of antisocial behaviors indicated by the informants. Information about the children's behaviors was gathered from each participant himself, one of his parents (usually the mother), and his teacher. Each informant was asked to complete a questionnaire. Participants were identified as "high risk" only if they engaged in three or more "disruptive behaviors;" these included: attacking someone, running away from home, setting a fire, stealing from a place other than home, truancy, vandalism, stealing from a car, robbery, stealing a bicycle, shoplifting, stealing a car, attacking someone with a weapon, gang fighting, hitting/hurting a teacher, hitting/hurting a parent, joyriding, burglary, being arrested, drinking alcohol, sniffing glue, or using marijuana. In total 30% of the participants in the original sample was identified as "high risk" (n=282); the remaining group was considered "low risk" (n=574). Then, 91% (n=257) of the participants from the "high risk" group and 43% (n=249) of the participants from the "low risk" group were randomly selected; hence, the sample of 506 boys. The participants in the high-risk group were over-sampled at a ratio 2:1-1:0. Weights were then assigned in order to reflect the original random selection from the public school population of Pittsburgh⁶. The weight assigned to the high risk group is .6485 and the weight assigned to the low-risk group is 1.3626 (Loeber et al., 1998; Wei, 1999).

⁶ Pittsburgh is a city of medium size, located in South Western Pennsylvania with an urban population of 369,879 and a county population of 1,281,666 (as per 2000 US Census).

The statistical analysis developed for this study (Chapter 5) specifically uses a subsample comprising only the 329 participants that reported having an intimate partner by age 23-25 years.

4.2.2 The interviews

Participants in the oldest sample were interviewed 16 times since they were age 13 years⁷. The first five interviews were scheduled every six months and thereafter every year. Until the participants were 17-18 years of age, information was also gathered from the participants' parents (the mother) and their teachers. The interviews took place at the participants' homes and the average duration of interviews was between 2 and 3 hours. For the initial interviews, the boys received a fee of \$5. However, the fee increased gradually phase after phase, and in the last assessment participants were paid \$85. The parents also received a fee that increased from \$12.50 at the first interview to \$60 at the last interview.

Each wave of interviews corresponds to a specific phase of the PYS life (assessment phase); from the first assessment in 1987 to the last assessment carried out in the year 2000, 16 phases were developed. Each assessment phase is marked by an alphabetical letter; however, the order of the letters is not exactly loyal to the English alphabet; this choice can be explained by technical decisions made by researchers of the PYS. For instance, the first assessment phase is indicated as S – which stands for "Screening" (in this phase the participants were selected from the random sample identified as either high risk or low risk individuals as explained in the previous section).

⁷ The average cooperation rate for the oldest sample over the 15 follow ups was 89.5%, reflecting a very low attrition.

The table below shows how the assessment phases were identified at each interview, the year when interviews were completed, and the average age of participants at each phase of the interviews.

Assessment Phase	Year	Age (in years)
S	1987	13
А	1988	13.5
В	1988	14
С	1989	14.5
D	1989	15
Ε	1990	15.5
G	1990-1	16.5
Ι	1991-2	17.5
K	1992-3	18.5
М	1993-4	19.5
0	1994-5	20.5
Q	1995-6	21.5
SS	1996-7	22.5
U	1997-8	23.5
W	1998-9	24.5
Ζ	1999-2000	25.5

Table 4.1: Assessment phases for participants of the PYS Oldest Sample (N=506)

Note: As the dates of interviews show, participants were interviewed every 6 months from phase S up to phase E and every year thereafter.

4.3 Empirical models

With this study I investigate risk factors of intimate partner abuse. In my models the dependent variable is an individual's probability of being an adult perpetrator of intimate partner abuse. The explanatory variables are first considered by category of influence corresponding to the three theoretical perspectives: the family violence, the mainstream feminist, and the life-course perspectives, as discussed in the previous Chapter. Accordingly, the family influence includes interpersonal conflict, experience of corporal punishment in the family of origin, and parental bonding during adolescence; the socio-cultural influence includes attitudes toward women and relationships with women, and the individual/personal influence includes academic competence, delinquency, and number of sexual partners. Next, I investigate the effect of participants' behaviors in early adolescence (ages 13-15), late adolescence (ages 16-18); pre-adulthood (ages 19-22). Four control variables are also included in the analyses: race, socio-economic status (SES), age at entering the romantic relationship, and alcohol use.

4.4 PYS variables employed in the statistical analysis

The Dependent Variable: Intimate partner abuse

Data on intimate partner abuse in the PYS were collected for the first time when participants were approximately 21; interviews were repeated every year until participants were 25 (phases Q to Z). However, for this dissertation I focus on data collected when participants were 23 and again when participants were 25 (that is, phases U and Z). This choice was made after exploring the data collected in the last 5 phases of the study. Only in phase U and phase Z were all participants in the oldest sample interviewed on their experiences with an intimate

partner and only in those two phases, had researchers used the same questionnaire; therefore, only data gathered at age 23 and 25 are comparable.

The questionnaire used to assess intimate partner abuse comprises a total of 37 questions; in particular the questionnaire asks about 15 behaviors that are violent or aggressive in nature and 22 behaviors that can be categorized as non-physical/controlling. Violent behaviors in the questionnaire are: physically twisted partner's arm; pushed, grabbed or shoved partner; slapped partner; physically forced sex on partner; shaken partner; thrown partner bodily; pulled partner's hair; thrown something at partner; choked or strangled partner; kicked, bit, or hit partner with a fist; hit or tried to hit partner with an object; beat up partner; burned or scalded partner; threatened partner with knife or gun; used a knife or gun on partner. Non-physical controlling behaviors in the questionnaire are: damaged a household item or part of the home; disposed of an important item of partner, got upset if dinner/housework was not done; purposely damaged partner's clothes or car; insulted or shamed partner in front of others; locked partner in or out of the house; told partner that she could not work or study; tried to stop partner from seeing family and friends; restricted partner's use of car/telephone; made threats to leave the relationships; tried to turn family or friends against partner; ordered partner around; frightened partner; treated partner like she was stupid; given in to partner but planned revenge; ridiculed partner; threatened to hit or throw something on partner; told partner she was ugly or unattractive; become abusive after using drugs or alcohol; thrown, smashed, or hit something in a disagreement; followed partner when she said not to; gone to partner's workplace/home after she said not to⁸.

⁸Eight of the violent behaviors listed in the PYS questionnaire were taken from the Conflict Tactic Scale (CTS), discussed in the previous Chapter (Straus and Gelles, 1986); more precisely, behaviors from the CTS are: thrown something at partner; pushed, grabbed, shoved partner; slapped partner; kicked, bit or hit partner with a fist; hit or tried to hit partner with an object; beat up partner; threatened partner with a knife or gun; and, used a knife or gun against partner. The remaining behaviors listed in the PYS questionnaire were identified by Moffit and collaborators for the Dunedin Study (Magdol et al., 1998).

Although non-physical/controlling behaviors and violent behaviors are different in nature, for the purpose of this research, all the behaviors were summarized into one variable named "intimate partner abuse". The reliability analysis shows that the 37 items included in the scale might actually be part of the same type of behavior (Cronbach's alpha = .87). Prior to the scale's creation, each item of the questionnaire was recoded into a dichotomous variable. In the PYS data set the possible answers to each question on intimate partner abuse behaviors were: Never, Once, Twice, Sometimes, Frequently and Very Frequently. Although the 37 behaviors were all combined into one scale, non-physical controlling behaviors and violent behaviors were recoded differently. More specifically, due to their level of seriousness, violent behaviors in the scale were recoded as 0 if the respondent answered "Never" and 1 for all other possible answers. In this way, even when a participant perpetrated a violent behavior only once, the scale is able to take that specific incident into consideration. Conversely, the scale for non-physical/controlling behaviors only includes those behaviors perpetrated more often in the interactions with a partner. In the scale, each non-physical controlling behavior was re-coded as 0 if the behavior never occurred or it occurred exceptionally (for instance, once or twice during the past year); it was coded as 1 if the behavior occurred sometimes, frequently or very frequently. This re-coding method was used in order to exclude behaviors that are very common between adult individuals involved in intimate relationships, which would explain very little about conflict within intimate contexts.

After all the 37 items were combined, the intimate partner abuse scale was dichotomized into a 0-1 variable for the statistical analysis. In this scale, 0 means that the participant reported never perpetrating any of the 37 behaviors listed in the scale, and 1 means that the participant perpetrated at least 1 of the 37 behaviors listed in the scale.

In the PYS, participants were first asked about their relationship with a current partner. In the study, current partner is a person with whom the participant has had a relationship for at least a month. All the participants who reported not being involved in a relationship with a current partner were interviewed about their relationship with a past partner. Furthermore, those participants who did not have a current partner or a past partner at the time of the interview were then asked about a relationship with a casual intimate partner. However, for the purpose of this dissertation the analysis focuses only on the group of participants who reported having a current intimate partner at the time of the interview.

In the PYS, participants were also interviewed about victimization of abuse at the hand of a partner (current, past, or casual partner). However, the variable victimization of abuse by an intimate partner won't be included in the statistical analyses of this dissertation⁹. This dissertation focuses solely on the analysis of abuse perpetrated by the participant against a female intimate partner.

In the first phase of interviews on intimate partner abuse considered for this research (phase U in the PYS data set), when participants were 23 years – a total of 52% of participants (n=263) reported having a current partner; however, only 46% of participants (n=234) were willing to answer questions about their relationship. In the second phase of interviews on intimate partner abuse considered for this dissertation (phase Z in the PYS data set), when participants were 25 years of age on average – 55% of participants (n=281) agreed to answer questions about their relationship with a current partner. Information gathered from the participants in phase U and phase Z was combined by computing the mean over the completed

⁹ A multivariate statistical analysis performed by using a 2-Stage Least Square regression indicated that the variable victimization of abuse by an intimate partner might be an endogenous factor in the analysis. In other words, the variable participant's victimization of abuse by an intimate partner is correlated to the error term in the model. This result might indicate that the two factors - perpetration of intimate partner abuse against a partner and participant's victimization of abuse by a partner - might be determined simultaneously in the model.

interviews. In this way, if participants were interviewed twice, the final score was computed as a mean of the two scores in each interview. If the participant was interviewed only once (because he skipped one of the two interviews) the score for the completed interview was considered as final score. Thus, only participants who skipped both interviews (at age 23 and age 25) were deleted from the count. In total 329 participants in the sample are taken into consideration for this empirical investigation; they represent 65.1% of the original "Oldest sample" of the PYS; the remaining group (n = 177) include participants who were not interviewed because they did not have a relationship with a current partner by age 23-25, those who refused, participants interviewed on a past or casual relationships, participants who skipped interviews and computer errors (in one case).

Table 4.2 summarizes the number of abusive behaviors perpetrated by the 329 participants of the Oldest sample against their wives/girlfriends between age 23 and 25. This analysis was developed by using the variable intimate partner abuse prior to dichotomization.

Number of behaviorsfrom the intimate partner abuse scalePrevalenceValue percent					
from the muniate partner abuse scale	Trevalence	Value percent			
0 No abuse	217	66.0%			
1 Abusive behavior	41	12.4%			
2 Abusive behaviors	24	7.4%			
3 Abusive behaviors	13	4.1%			
4 Abusive behaviors	10	3.1%			
5 Abusive behaviors	7	2.0%			
6 Abusive behaviors	3	1.0%			
7 Abusive behaviors	5	1.4%			
8 Abusive behaviors	1	.4%			
9 or more Abusive behaviors	8	2.2%			
Total N	329	100.0%			

Table 4.2: Distribution of abusive behaviors against intimate partners (N= 329)

Note: The table shows the number of behaviors perpetrated against wives/girlfriends by the 329 participants from the "Oldest Sample" who reported being involved with an intimate partner by age 23-25. A weight was applied to this descriptive analysis.

Table 4.2 shows that over a half of the 329 participants who were involved with an intimate partner by age 23-25 reported not perpetrating any of the 37 behaviors (either violent or non-physical controlling behaviors) included in the intimate partner abuse scale used for this study. One fourth of participants perpetrated 1 to 3 of the behaviors included in the scale, while approximately 9% of the 329 participants perpetrated between 4 and 10 abusive behaviors. The remaining 2.2% reported perpetrating between 12 and 23 behaviors from the intimate partner abuse scale (23 is the maximum number of behaviors reported by the participants).

A descriptive analysis of the raw data performed by exploring each abusive behavior taken separately has indicated that among the PYS participants involved with an intimate partner by age 23-25 (N= 329), the most common non-physical/controlling behaviors perpetrated against a current partner were "got upset if dinner/housework was not done when it was supposed to be done" and "made threats to leave the relationship"; whereas the most frequently perpetrated violent behaviors were "pushed, grabbed, or shoved partner" and "physically twisted partner's arm".

Table 4.3 shows the prevalence of intimate partner abuse among the 329 participants who were involved in an intimate relationship by age 23-25. The descriptive analysis was developed by using the dichotomized version of the intimate partner abuse variable that will be employed in the multivariate statistical analysis developed in Chapter 5. A weight was applied to the analysis.

n	%	
217	66.0%	
112	34.0%	
329	100.0%	
	112	

Table 4.3 Prevalence of intimate partner abuse - age 23-25

Note: The analysis of prevalence was conducted using the dichotomous scale for intimate partner abuse. Data from phases U and Z were combined. A weight was used for the analysis. In total 329 participants of the original PYS Oldest sample (N=506) reported being involved in an intimate relationship by the time of the interview (age 23-25).

Table 4.3 shows that 34.0% (n = 112) of the 329 participants interviewed on intimate partner abuse against a current partner reported perpetrating at least one of the behaviors included in the intimate partner scale.

Explanatory Variables

I will describe how each explanatory variable was computed. For each variable I will also provide a short summary of the descriptive statistical analyses. I will also provide a comparison between the sub-sample of 329 participants interviewed on their relationship with an intimate partner (by age 23-25) and the original PYS Oldest Sample of 506 males selected by the PYS investigators with statistically accurate sampling methods. It is important to investigate whether the sub-sample reflects the characteristics of the original PYS sample from which it was drawn in all the aspects investigated in this research. Indeed, differences between the sub-sample of 329 participants and the original PYS sample of 506 males might indicate that participants who had a girlfriend/wife by age 23-25 (n=329) differ from the participants who were not involved in an intimate relationship at the time of the interview (phases U and Z) (n=177). Later, in Chapter 5, I

will test the statistical differences between the group of 329 participants interviewed on their relationship with an intimate partner (age 23-25) included in the multivariate analysis and the remaining group of 177 participants from the Oldest Sample who reported not having an intimate current partner at the time of the interview (age 23-25) and that were excluded from the multivariate statistical analysis.

I consider each explanatory variable in conjunction with the category of influence with which it is associated.

- Family influence

Interpersonal conflict: PYS participants were asked how often they and their partner (girlfriend/wife) agreed about something or spent quality time together. Five questions from the Dyadic Adjustment Scale questionnaire (Spanier, 1976) were used to measure the level of interpersonal conflict with an intimate partner among the 329 participants who reported having a wife/girlfriend at the time of the interview (age 23-25). In the PYS the questions were specified as: 1. How often do you and your partner agree about what you want out of life? 2. How often do you and your partner agree about aims, goals, and things believed important? 3. How often do you and your partner agree about amount of time spent together? 4. How often do you and your partner have an interesting chat? 5. How often do you and your partner calmly discuss something? (Loeber et al.,1998). The variable/scale "Interpersonal conflict" employed in the analysis was computed by combining the 5 different variables from the raw data. In each phase the 5 variables were summed after dividing each by the number of possible outcomes/answers¹⁰. Information from each phase (phase U and Z) was combined into one variable to be used in the analysis that was computed by taking the mean of all completed interviews. The scale is a

¹⁰ Question 1, 2 and 3 had five possible answers: 0 never, 1 hardly ever agree, 2 sometimes, 3 almost always agree, 4 always agree. Differently, questions 4 and 5 had six possible answers: 0 never, 1 less than once a month, 2 once or twice a month, 3 once or twice a week, 4 once a day, 5 more often.

continuous variable where the higher the score the lower the level of interpersonal conflict. The N for the scale "interpersonal conflict" is 313, which indicates that 16 of the 329 participants interviewed on their relationship with a current partner at age 23-25 either skipped the interview or refused to answer the question¹¹. The minimum value for the variable interpersonal conflict with intimate partner is 0 and the maximum value is 11.4. A descriptive analysis performed by comparing the responses to the five questions taken separately has indicated that between age 23-25 a total of 10% of participants reported being often in disagreement with their partner about time spent together and 9% of participants had trouble discussing an issue calmly with partner¹².

Experience of corporal punishment in the family of origin: For the analysis I use a construct of the participant's experience of corporal/physical punishment computed by the PYS researchers (Loeber et al., 1998). This construct summarized information gathered from both the participant and his caretaker (usually the mother) about the way she/he usually disciplined the child. The questions addressed to the informants come from the Discipline Scale (Patterson and Stouthamer-Loeber, 1984). Caretakers were asked how often¹³ they slapped, spanked, or hit their son if he did something that he was not allowed doing or that they did not like. Participants were asked to supply the same information about the behavior of each caregiver, so that the replies could be more effectively corroborated. The responses from the informants were averaged and the construct was then dichotomized by applying a cutoff rule¹⁴. In the original PYS construct experience of corporal punishment, 0 means that corporal punishment was applied infrequently and 1 means that it was applied frequently. For the analysis, I have created a new variable by

 $^{^{11}}$ Only participants who reported being involved with a current partner at the time of the interview were asked guestions about interpersonal conflict with an intimate partner (N=329).

 $^{^{12}}$ A weight was applied to the statistical analysis.

¹³ Possible answers were: almost never, sometimes, often, or very often.

¹⁴ The variable was set to 1 if the average was greater than 79.8% and to 0 otherwise.

calculating the mean over all the completed interviews from age 13 through age 15 (phases A through E); the value of this new variable ranges from 0 to 1 (continuous variable).

A descriptive analysis has indicated that, according to the caretakers' reports, one third of participants (approximately 30.0%) received corporal punishment by age 13-15. However, only one fifth of participants (19.5%) reported receiving corporal punishment by a parent/caretaker. When corporal punishment was applied, participants reported being punished more often by their fathers than by their mothers. This was true for both the participants in the Oldest Sample (N=506) and the 329 participants interviewed on their relationship with a current partner by age 23-25. A weight was applied to the analysis.

Parental bonding during adolescence: The PYS also provides detailed information about the participant's involvement with his parent/caretaker. Several constructs were computed by the PYS researchers by combining information gathered through the Supervision/Involvement Questionnaire¹⁵ that was administered to each parent/caretaker and participant in the study (Loeber et al., 1998). For the statistical analysis I have computed a new scale named "Parental Bonding" that combines information from 5 different PYS existing constructs: caretaker's time spent with child, caretaker talks about activities with the child, caretaker's communication with child, boy not involved with caretaker, and boy has bad relationship with primary caretaker. The reliability analysis for this scale shows Cronbach's alpha = .66. Information from interviews completed from age 13 to age 15 are included. A mean of the score of all the completed interviews was computed. The new scale is a continuous variable that ranges from 0 ("weak bonding" with parent/caretaker) to 1 ("better bonding" with parent/caretaker). An analysis of the frequencies computed by using weighted data from each phase (from phase A to phase E) shows

¹⁵ See the Supervision/Involvement Questionnaire in Appendix A. The scale was identified by the PYS researchers' literature review of family factors related to delinquency (Loeber and Stouthamer-Loeber, 1986; Moos and Moos, 1975; and Skinner et al., 1983).

that at age 13 to 15 years one fifth of participants experienced weak or no bonding to parents/caretakers. This was true for both the original "Oldest Sample" of the PYS (N=506) and the sub-sample of 329 participants who were interviewed on their relationship with a current partner (age 23-25)¹⁶.

- Socio-cultural influence

Attitudes toward women and relationships with women: In the PYS participants were asked about their attitudes toward specific issues, like their attitudes toward women and their relationships with women. The two variables selected for this analysis refer to the participant's opinion about getting a girl pregnant and not helping her to bring up the child, and cheating on a girlfriend/wife. The two questions are part of the PYS Perception of Problem Behavior Questionnaire (Loeber, 1982; Loeber et al., 1998). For the analysis, I have combined the two variables into one scale that is intended to express the participant's attitude toward women and relationships with women; this scale is a continuous variable (the higher the score the more negative the participant's attitude toward women/relationships with women). Information from phases U and Z was combined by computing the mean over all the completed interviews. The scale's value ranges from 2 to 8. An analysis of frequencies using weighted data has indicated that, between age 23-25, slightly more than 6% of participants reported that "it is alright to get a girl pregnant and not help her to raise the child", and approximately 10% of participants said that it is alright sometimes to cheat on an intimate partner. This was true for both the original PYS Oldest Sample and the sub-sample of 329 participants interviewed on their relationship with a current intimate partner by age 23-25.

- Individual/Personal influence

¹⁶ Only 328 participants out of the 329 participants in the sub-sample responded to the interview on parental bonding.

Academic competence: Information about the participant's academic competence was gathered in the PYS (Loeber et al., 1998). Three informants were interviewed about each participant's performance on reading, writing, spelling, math, and three other subjects: the teacher, the caretaker and the participant himself. Caretakers were asked questions from the Child Behavior Check List (CBCL; Achenbach, 1978; Achenbach and Edelbrock, 1979, 1983), the teacher from the Teacher Report Form (TRF; Edelbrock and Achenbach, 1984), and the participant from the Youth Self-Report (YSR; Achenbach and Edelbrock, 1987). The scale for the evaluation of each performance consisted of failing, below average, average, and above average. The responses of the three informants were averaged and the construct was then dichotomized by applying a cutoff rule.¹⁷ The scale for academic competence used for the analysis is a continuous variable that was computed by calculating the mean of the construct over all the completed interviews (age 13 to 15, phases S through E). The scale's value ranges from 0 to 1. An analysis of frequencies developed by using weighted data has indicated that one third of participants were considered below grade level in their reading skills and one fourth of participants were considered below grade level in their writing, spelling, and math achievements. This was true for both the "Oldest sample" of 506 participants and the sub-sample of 329 participants interviewed on their relationship with a current intimate partner.

Delinquency: In the PYS delinquency was measured through the Self-Reported Delinquency Questionnaire (SRD) adapted from the National Youth Survey (Elliott et al., 1985). Information about the participants' delinquent behavior was also gathered from caretakers (CBCL) and teachers (TRF). The interviews with the participants were then compared to official records form juvenile and adult courts (arrest and conviction). These interviews were conducted consistently from age 13 years through age 25 years (from Screening – or phases S to Z). In total

¹⁷ The variable was set to 1 if the average was greater than 75.0% and 0 otherwise.

36 items were included in the scale to assess delinquency. A summary of the major questions of the SRD Questionnaire is reported in Appendix B. For the analysis I use a construct that was computed by the PYS researchers across all the phases of the study; the construct is labeled as "Number of Levels of Delinquency" and summarizes all possible outcomes of participants' delinquency (Loeber et al., 1998). This construct identifies five levels of delinquency: 1. Vandalism at home or stealing from home; 2. Minor vandalism outside home, minor fire-setting, minor fraud, stealing from school or elsewhere, or shoplifting; 3. Major vandalism/fire setting, major fraud, stealing a bike/skateboard or something worth \$5-\$50, joyriding, pick pocketing, stealing from a car, holding stolen goods or something worth more than \$50, carrying a weapon or gang fighting; 4. Drug dealing, breaking and entering, auto theft, strong arm, attack or rape; 5. Two or more delinquent behaviors from level 4 of delinquency.

For the statistical analyses I have created a new variable by computing the mean over all the completed interviews from age 13 through age 22 for the existing PYS construct (phase S through phase SS). This variable allows me to summarize all the delinquent behaviors perpetrated by the participant from early adolescence through pre-adulthood. This scale is a continuous variable: the higher the score the more serious the participant's level of delinquency. Descriptive statistics show that the minimum is 0 and the maximum is 4.33. Three more variables were computed by collapsing information about the participants' delinquency by age group (13-15; 16-18; 19-22) by calculating the mean score over all the completed interviews. Thus, for the variable delinquency age-group 13-15, I have computed the mean over all the completed interviews between phase A and phase E; for delinquency age-group 16-18 I have computed the mean over all the completed interviews from phase G through phase K; and for delinquency age-group 19-22 I have computed the mean over all the completed interviews from phase U through phase SS¹⁸.

An analysis of the trend in number of levels of delinquency (by age-group) among the participants (computed by using weighted data) has indicated that delinquency was higher between age 13 and 18 (it slightly increased from age 13-15 to age 16-18) and started declining thereafter, reaching its lowest level at age 25. This was true for both the original PYS Oldest Sample of 506 boys and the sub-sample of 329 participants interviewed on their relationship with a current partner at age 23-25. This trend confirms existing developmental theories in the literature, which point out that delinquent/antisocial behavior escalates during late adolescence (or during teenage years) and subsequently enters a phase of "desistance" (see for instance, Loeber and Le Blanc, 1990).

Number of sexual partners: Every year the PYS participants were interviewed about their sexual activity and about their sexual partners. For this dissertation, I have selected the construct "Number of heterosexual partners in the past year" (Loeber et al., 1998). However, for the statistical analysis I have combined information into three 3 different scales, one for each age-group considered in the analysis (age 13-15; age 16-18; and age 19-22) by taking the mean over all the completed interviews within each age group. Following the same criterion I have also created a continuous variable that summarizes about the participant's number of sexual partners from age 13 to age 22 (computing the mean of all the completed interviews in phase A through phase SS); the higher the score the larger the numbers of female sexual partners¹⁹. Descriptive

¹⁸ A few missing cases were identified for the delinquency scales. More specifically, for the sub-sample of 329 participants interviewed on their relationship with an intimate partner, the variable delinquency age 13-15 carried no missing cases, the variable delinquency age 16-18 and the variable delinquency age 19-22 each carried 3 missing cases. Missing cases for the variable delinquency might be due to skipped interview or refusal to answer questions. A complete list of missing cases for all the variables employed in the analysis is reported in table 4.5.

¹⁹ A few missing cases for the variable "number of sexual partners" were identified among the 329 participants involved in a current romantic relationship by age 23-25. More specifically, the variable number of sexual partners

indicated that, while for the original PYS Oldest Sample of 506 participants the minimum number of heterosexual partners is 0 and the maximum is 29, for the sub-sample of 329 participants interviewed on their relationship with a current partner (age 23-25) the minimum is 0 but the maximum number is 24. However, an analysis of the distribution of sexual partners through the life-span, has indicated that, for both the original Oldest sample (N=506) and the sub-sample of 329 participants interviewed on their relationship with a current partner, the number of heterosexual partners are low during early adolescence (age 13-15), reach a peak during late adolescence (age 16-18) and tend to decrease during pre-adulthood.

Control Variables

The control variables in the analysis comprise three demographic factors (race, SES, and age at entering the relationship) and the variable alcohol use. I will describe how each variable was computed/recoded and I provide a summary of the descriptive analysis for each variable. Furthermore, I report a comparison between the sub-sample of 329 participants interviewed on their relationship with their current partner and the PYS Oldest Sample of 506 males from which it was drawn.

Race: In the PYS Oldest Sample (N=506), 213 participants are Whites, 281 are African Americans, 1 participant is Hispanic, 2 are Asians, and 9 are Biracial. In order to simplify the analysis I have recoded the original PYS race variable into a dichotomy that takes value 0 when participants reported being white and value 1 for all the non-White participants. For the PYS Oldest sample of 506 participants, 294 are non-White (58.1%) and the remaining 212 participants are White (41.9%). Among the 329 participants interviewed on their relationship

age 13-15 carries no missing cases, the variables number of sexual partners age 16-18 and number of sexual partners age 19-22 each carries 3 missing cases. The missing cases for the variable number of sexual partners might be due to either skipped interviews or refusal to report. A complete list of missing case for each variable included in the statistical analysis is reported in this Chapter (Table 4.5).

with a current intimate partner by age 23-25, 181 are non-White (55.5%) and the remaining 148 are White (44.5%).

Participants' SES: In the PYS constructs the participants' SES was computed by recoding the job categories (from 0 to 880 in the original code-book) according to the Hollingshead Scale (1975), which includes all groups of job-income categories from "farmer labor or menial worker" to "executive or major professional". Job prestige and educational level are also included in the construct (Loeber et al., 1998). For the analysis I have computed a scale that combines information for participants age 23 and 25; the variable was computed by taking the mean over all the completed interviews from phase U through phase Z. The SES scale was then dichotomized and its value is 0 for Low SES and 1 for High SES. Frequencies show that by age 23-25, three quarters of the participants within the group of 329 males interviewed on their relationship with an intimate partner had a Normal-High SES, while only two thirds of participants among the 506 males of the original Oldest Sample had a Normal-High SES. In Chapter 5 I report the statistical differences between the group of participants included in the analysis (329) and the remaining group of the Oldest Sample not included (n=117).

Age at beginning of romantic relationship: Participants who reported having a current partner were interviewed about their age when they and their partner first started considering themselves as a couple. The question was asked in retrospective in the Intimate Partner Abuse Questionnaire (Loeber et al., 1998); therefore, only the 329 participants who were involved in an intimate relationship at the time of the interview (age 23-25) were asked this question²⁰. In Phase U (age 23 years) the age distribution shows that, on average, participants reported that they and

²⁰ An analysis of frequencies shows that the N for the variable "age at beginning of romantic relationship" is actually 324 (5 participants are considered missing). The missing cases might refer to participants who refused to answer the question or answered "don't know". Since the question was asked in retrospective, it is possible that some of the participants could not remember at what age they and their partner had started considering themselves a couple.

their partner started considering themselves as a couple at around 21 years. The minimum for phase U is 13 years and the maximum is 25 years. The standard deviation in the distribution is 2.18 years. In Phase Z (age 25) participants reported starting to consider themselves a couple at around age 22 on average. The minimum for the distribution in phase Z is 13 and the maximum 27 years. The standard deviation for this distribution is 2.9 years. In order to combine information from both phase U and phase Z, I have computed a scale as the mean value across the phases. Missing values were limited to 1 for each phase per participant; in this way if the participant was interviewed in phase U but not in phase Z (or vice versa) the answer provided in one phase was taken into consideration. If the participant was not interviewed in either phases, then he was counted as a missing case, and if the participants were interviewed in both phases the two scores (age numbers) were averaged and the mean value was specified in the scale. The total N for the two phases combined is 324.

*Alcohol use*²¹: In the PYS participants were asked about their alcohol consumption from age 13 to 25. For the analysis I make use of a construct computed by the PYS researchers: "frequency of alcohol use". This construct includes consumption of beer, wine, hard liquor, or any other type of alcohol used during the year prior to the interview. I create a new scale to combine the participant's information about the frequency of alcohol use from age 13 through 25 by computing the mean over all the completed interviews. A descriptive analysis has indicated that, among the PYS participants, alcohol use was very low till age 17; it increased sharply from age 18 to 21 to decrease thereafter. This was true for both the Oldest Sample of 506 boys and the sub-sample of 329 participants interviewed on their relationship with an intimate partner (age 23-25). However, descriptive statistics indicated some differences in alcohol consumption between

²¹ This excludes drinking alcohol at festive occasions with parental knowledge when participants were minors (Loeber et al., 1998).

the two samples. More specifically, for the sub-sample of 329 participants interviewed on their relationship with an intimate partner, the mean of alcoholic beverages during the year prior to the interview is 41 and the standard deviation is 39. Differently, for the sample of 506 participants, the mean is 37 and the standard deviation is 39. This might indicate that some differences in frequency of alcohol use exist between the group of participants interviewed on their relationship with a current partner (n=329) and the group of participants not included in the analysis (n=177). In Chapter 5 I will assess the statistical differences between the two groups.

Further details about the data

Weight: As specified earlier in this Chapter, delinquent/antisocial individuals (or "high-risk" participants) were over-sampled in the PYS. In order to make sure that the analysis is conducted using a representative sample and that all the estimates are unbiased, I apply a weight to the distribution. The variable weight (w) takes 2 different values: .65 and 1.36. It was computed by researchers of the PYS to make sure that the sampling reflects the population at large. Table 4.4 explains how the weight is distributed among participants of the PYS in the analysis.

Weight value	Oldest sample (N= 506)		Partner Abuse sample (N= 329)	
-	Freq.	0⁄0	Freq.	%
.65	257	50.8	106	32.1
1.36	249	49.2	223	67.9
Total	506	100.0	329	100.0

 Table 4.4: Distribution of weight among PYS participants

Note: The weight was constructed by the PYS researchers; weighting is a standard procedure that is used to correct for over-sampling and it allows researchers to develop statistical analyses using unbiased estimators (Rosnow and Rosenthal, 1989). A comparison between the Oldest Sample of 506 males and the sub-sample of 329 participants interviewed on their relationship with an intimate current partner (age 23-25) is reported in the table.

Missing values: in the PYS missing cases are due to skipped interviews, refusals to answer questions, failure to remember information ("Don't Know") or computer errors.

The multivariate analysis developed for this dissertation focuses exclusively on the group of participants who were involved in a romantic relationship for at least a month at the time of the interview (N= 329); the remaining 177 participants from the PYS Oldest Sample, that is, those interviewed about a past partner and those interviewed about a casual partner, are counted as a separate group and won't be included in the multivariate statistical analysis (i.e. they are not considered as missing cases).

The missing cases for the multivariate statistical analysis depend on the missing cases reported for each single variable employed in the multivariate analysis. Table 4.5 specifies the missing cases for all the variables included in the multivariate analysis. Because all the empirical models adopted for the analysis use different explanatory variables (risk factors of abuse), the missing cases might vary model by model according to the variables included in the analysis.

Human subjects: The PYS project was funded by the National Institute of Mental Health (NIMH) and the Office of Juvenile Justice and Delinquency Prevention (OJJDP). The research program was approved by the University of Pittsburgh Institutional Review Board (IRB). As an employee of the PYS research center, I had the opportunity to use the data as they were collected by expert trained interviewers. Being part of the Western Psychiatric Institute and Clinic (University of Pittsburgh Medical School) every researcher of the PYS who has access to data collected from human subjects has to be certified for responsible use of such data. A HIPPA Researchers Privacy Requirements Module associated with the Educational and Certification Program in Research & Practice Fundamentals was issued as formal authorization that allowed me to investigate and analyze the data from human subjects of the PYS. I do not intend in any

way to use information to me available to identify names or features of the individuals who participated in the study.

4.5 Summary of the variables employed in the statistical analyses

Table 4.5 lists the variables that are included in the statistical analyses. For each variable I specify the name and the age group to which the variable refers. Due to the particularity of the empirical models (which will be explained in detail in the next section), some of the explanatory variables were computed in two different ways, first grouping all the possible interviews (all phases from age 13 to age 22), and then age-group by age-group (13-15; 16-18; and 19-22); (see for instance the variables delinquency the variable number of sexual partners). Also, the table reports the total number of observations for each variable and its missing cases.

The N for intimate partner abuse is 329, that is, the total number of PYS participants in the "Oldest sample" (N=506) who were involved in a relationship with an intimate partner by age 23-25 and that agree to talk about that relationship. However, as discussed above, the missing cases vary across the independent variables: the variable interpersonal conflict has a total of 16 missing cases, the variable parental bonding shows 1 missing case, delinquency and number of sexual partners for the age-group 16-18 and for the age group 19-22 carry 3 missing cases. Among the control variables, the variable age at the beginning of the relationship has 5 missing cases. Missing cases might be due to skipped interviews, refusal to answer the question, failure to remember ("don't know"), or computer error. The count of missing cases is important for the multivariate analysis. As a matter of fact, each variable carries its own missing cases within the empirical model affecting the total count for each statistical procedure. Hence, the maximum number of possible missing cases for the multivariate analysis is 34 out of the 329 participants

included in the analysis. The 177 participants who did not have an intimate partner at the time of the interview won't be included in the count.

Variable name	Age-group	Туре	Ν	Missing cases
<u>Dependent Variable</u>				
Intimate partner abuse	(23-25)	Dichotomous	329	16
Explanatory Variables				
Interpersonal conflict	(23-25)	Continuous	313	0
Experience of corporal punishment	(13-15)	Continuous	329	0
Parental bonding	(13-15)	Continuous	328	1
Negative attitudes toward women	(23-25)	Continuous	329	0
Academic competence	(13-15)	Continuous	329	0
Delinquency	(13-22)	Continuous	329	0
Delinquency by age group				
	(13-15)	Continuous	329	0
	(16-18)	Continuous	326	3
	(19-22)	Continuous	326	3
Number of sexual partners	(13-22)	Continuous	329	0
Number of sexual partners by age g	-			
	(13-15)	Continuous	329	0
	(16-18)	Continuous	326	3
	(19-22)	Continuous	326	3

Table 4.5: Summary of the variables included in the analysis

Variable name	Age-group	Туре	Ν	Missing cases
Control variables				
Race	13 Screening	Dichotomous	329	0
SES	(23-25)	Continuous	329	0
Age at beginning of relationship		Continuous	324	5
Alcohol use	(13-25)	Continuous	329	0
Total				34

Table 4.5: Summary of the variables included in the analysis (continue)

Note: The table summarizes the variables employed in the statistical analysis, specifying the age-group at which information was gathered, the variable type, the N count for each variable, and its missing cases. The N reported represent the actual N for the PYS distribution of participants who were interviewed on their relationship with a current partner at age 23-25.

4.6 Hypotheses and plan for the statistical analysis

In this section I specify the hypotheses that will guide the empirical investigation. Hypotheses were derived from the literature, based on controversies on the causes of intimate partner abuse as discussed in Chapter 3. The hypotheses reflect the two empirical strategies specified in the introduction. First, I investigate risk factors of intimate partner abuse by category of influence (family, socio-cultural, and individual/personal influences). Second, I investigate which factors considered by age-group (13-15 or early adolescence; 16-18 or later adolescence; 19-22 or pre-adulthood) influence the likelihood of being abusive during adulthood (age 23-25). For each hypothesis, I anticipate the plan for the empirical models that will allow me to verify whether my statements are correct. I will develop the statistical analysis in Chapter 5 and also in Chapter 5 I will explain the technical advantages of using these particular statistical models.

Investigating risk factors of abuse by category of influence

The first hypothesis is: Males who experience conflict within the relationship with an intimate partner (age 23-25), have experienced frequent corporal punishment in the family of origin (age 13-15), or have had weak bonds with parents/caretakers during early adolescence (age 13-15) are more likely to be abusive against their intimate partners in adulthood (age 23-25). This hypothesis addresses the major issues/concerns discussed by proponents of the family violence perspective (Straus, 1991, 2005; Mills, 2003; Straus and Yodanis, 1996) and provides a theoretical basis to analytically investigate whether a family influence affects the individual's propensity of abusing an intimate partner in adulthood. In Chapter 5 I will test this hypothesis by using a binary logistic regression where the dependent variable is intimate partner abuse (age 23-25) and the explanatory variables are all the family risk factors for abuse (interpersonal conflict with one's partner (age 23-25), experience of corporal punishment (age 13-15), and parental bonding during adolescence (age 13-15)). Four control variables will be also included in the analyses: race, SES (age 23-25), age at the beginning of the relationship, and frequency of alcohol use (age 13-25). In addition, I will focus on the role of race as risk factor for adult intimate partner abuse; thus, I will re-run the same empirical model by separating the two racial groups (White and non-White). This strategy was suggested by PYS researchers who had previously analyzed the causes of violent/aggressive behaviors among the participants in the study by separating the two racial groups (see for instance the work by Loeber, Stouthamer-Loeber and Wei, 2002).

<u>The second hypothesis</u> is: *Males involved with an intimate partner who have negative attitudes toward women and toward relationships with women (age 23-25) are more likely to abuse their intimate partners in adulthood (age 23-25).* This hypothesis reflects an important

issue that has concerned mainstream feminist theorists for almost fifty years (Dobash and Dobash, 1998). Although, the influence of one's beliefs, values and attitudes has been the focus of a number of research studies in feminist literature, quantitative/empirical papers have neglected this problem, mainly because of lack of available data. This hypothesis provides me with a theoretical basis for analyzing whether a socio-cultural influence exists within the causes of male intimate partner abuse. In Chapter 5 I will test this hypothesis by using a binary logistic regression model. For this model, the dependent variable is intimate partner abuse (age 23-25) (dichotomy) and the explanatory variable is attitudes toward women and relationship with women (age 23-25). Like in the previous model, four control variables are included in the analysis: race, SES, age at the beginning of the romantic relationship, and frequency of alcohol use (age 13-25). Furthermore, I will focus on the role of the variable race in the analysis of socio-cultural factors. Thus, I will re-run the empirical model to test the second hypothesis by separating the two racial groups (White and non-White).

The third hypothesis is: *Males involved with an intimate partner who have a low* academic competence (age 13-15), are delinquent (age 13-22), or have had large numbers of female sexual partners (age 13-22) are more likely to perpetrate abuse against their intimate partners (age 23-25). This hypothesis summarizes the main controversies on the causes of intimate partner abuse that have been indicated by life-course researchers (Moffitt and Caspi, 1999; Farrington 1994). It provides me with a theoretical basis for investigating whether individual/personal influences affect the probability of being abusive against an intimate partner in adulthood (age 23-25). In Chapter 5 I will test this hypothesis by using a binary logistic regression where the dependent variable is intimate partner abuse age 23-25 and the explanatory variables (individual/personal risk factors) are: academic competence (age 13-15), delinquency (age 13-22), and number of sexual partners (age 13-22). Four control variables will be also included in the analysis: race, SES (age 23-25), age at the beginning of the relationship, and frequency of alcohol use (age 13-25). In Chapter 5 I will also develop an analysis of the individual/personal risk factors by separating the two racial groups (White and non-White). This strategy seems to be necessary to explore in depth how intimate partner abuse in adulthood varies among the participants by race (Wei, 1999).

The fourth hypothesis is: Risk factors of abuse range among different categories of influence (family, socio-cultural, and individual influences). I hypothesize that family risk factors are more likely than socio-cultural or individual risk factors to influence one's likelihood of being abusive against an intimate partner in adulthood. This hypothesis summarizes all the main controversies on the causes of intimate partner abuse discussed in the literature by proponents of the family violence, mainstream feminist, and life-course perspectives and reflects the dominant view in sociology/criminology literature that experiences in the family during early adolescence tend to shape an individual's ability to manage interpersonal relationships (Siegel, 2006). Furthermore, this hypothesis provides a theoretical basis for comparison of that influence that each category of influence have on the individual's propensity of perpetrating abuse against an intimate partner in adulthood. In Chapter 5 I will test this hypothesis by using a multi-level binary logistic regression where the variables are analyzed by block. The dependent variable for the model is intimate partner abuse (age 23-25). The first equation (or block) includes all the control variables (race, SES, age at the beginning of the relationship, and frequency of alcohol use). The second equation, which represent the family influence block, includes the family risk factors, namely, interpersonal conflict with partner (age 23-25), experience of corporal punishment (age 13-15), and parental bonding during adolescence (age 13-15). The third

equation, the socio-cultural block, includes the factor attitudes toward women and relationships with women (age 23-25). The fourth equation, the individual/personal block, includes the factors academic competence (age 13-15), delinquency (age 13-22), and numbers of sexual partners (age 13-22).

Investigating risk factors of intimate partner abuse by age-group

The fifth hypothesis is: Personal characteristics, experiences, and behaviors that manifest early in life (during early adolescence) tend to influence the probability of being abusive against intimate partners during adulthood (measured at 23-25 years) more than experiences and behaviors that manifest later in life (age groups 16-18 and 19-22). This hypothesis reflects the suggestions provided by both family violence theorists and life-course theorists who suggest that experiences/behaviors that manifest early in life may influence adult behaviors within intimate contexts (Straus, 2005; Moffitt and Caspi, 1999; Mills, 2003). Furthermore, this hypothesis provides a theoretical basis for analyzing risk factors of intimate partner abuse by age-group (early adolescence or age 13-15, late adolescence or age 16-18, preadulthood or age 19-22, and adulthood or age 23-25)²². In Chapter 5 I will test this hypothesis using a multilevel binary logistic regression, where the variables for each age group are added to the model by block (or group of factors). The first block includes all the control variables (race, SES, age at the beginning of the relationship, and frequency of alcohol use). The second block includes all the risk factors of adult intimate partner abuse that were measured between age 13 and 15 (academic competence, experience of corporal punishment, parental bonding, delinquency, number of sexual partners). The third block includes all the risk factors measured between age 16 and 18 that are used for this dissertation (delinquency and number of sexual

²² The rational behind the 4 age-groups has been explained in both Chapter 1 and Chapter 3. The four age groups represent 4 developmental phases/stages and they were inspired by Erik Erikson readings on stages of development (1994).

partners). The fourth block includes all the risk factors measured between age 19 and 22 (delinquency and number of sexual partners).

In the next Chapter I will develop the statistical analysis and I will test all the hypotheses above specified. I will discuss each empirical model and I will report the findings. After testing each hypothesis, I will construct a model that would take into consideration only risk factors of abuse that were statistically significant in the models used to test the 5 hypotheses. I will refer this model simply as the "final model". The final model will help to summarize the main findings of the statistical analyses.

CHAPTER V

STATISTICAL ANALYSIS

5.1.: Empirical investigation

In this Chapter I develop the statistical analysis. First, I compare participants included in the analysis (those interviewed on their relationship with a current partner) to participants who are not included in the analysis because they did not have a current partner or simply did not agree to talk about their intimate relationship. Second, I investigate the bivariate association between each risk factor of intimate partner abuse (explanatory variables) and intimate partner abuse (dependent variable). Finally, I develop the multivariate analysis at the core of this empirical investigation. The multivariate analysis includes an analysis of risk factors of intimate partner abuse by category of influence (family, socio-cultural, and individual/personal influences) and an analysis of precursors of intimate partner abuse considered by age-group (age 13-15; age 16-18; and age 19-22). The empirical models designed for this investigation allow me to test the five hypotheses that I have specified in Chapter 4.

5.2 Comparing the "current partner-group" to the other participants in the sample

Among the PYS participants in the "Oldest sample" (N=506), a total of 329 males (65.1%) were interviewed about their relationship with a current partner (a partner with whom they were in a romantic relationship for at least a month prior to the time of the interview). It is important to compare these 329 participants (current-partner group) to the 177 participants (34.9%) excluded from the analysis (group of participants who did not report having a current partner at the time of the interview by age 23-25). In particular, it is appropriate to verify that the group of participants excluded from the analysis is not different from the group of participants included in the analysis with respect to aspects taken into consideration for this research, that is, race, SES, attitudes toward women, parental bonding, corporal punishment, academic

competence, delinquency, and number of sexual partners. Differences about interpersonal conflict and age at entering the relationship will not be investigated here because only participants who were interviewed on a current partner were asked questions about these two matters.

By using a statistical procedure that allows me to investigate the mean differences between the two groups of participants (independent sample t-test), I test the differences between the current partner-group and the non-current partner groups' experiences of corporal punishment (age 13-15), parental bonding (age 13-15), attitudes toward women and relationships with women (age 23-25), academic competence (age 13-15), delinquency (age 13-22, age 13-15, age 16-18, and age 19-22), number of sexual partners (age 13-22, age 13-15, age 16-18, and age 19-22), SES (age 23-25), and alcohol use (age 13-25). Furthermore, I test racial differences between the current partner-group and the non-current partner group of participants by using the Chi-square test²³. The results of the analysis are indicated in Table 5.1.

²³ The variable race is a dichotomous variable, therefore differences between the two groups can be assessed by using the Chi-square statistics.

			ent	Othe	r		Mean difference		
Comparison Varial	die(s)	N	Mean	Ν	Mean	Test	Coefficient	Significance	
Explanatory Variable	es								
(Frequent) Corporal	punishment (13-15)	329	.24	177	.26	T-test	.72	.474	
(Low) Parental bond	ling (13-15)	328	.85	177	.80	T-test	-1.8	070	
(Negative)Attitudes	toward women (23-25)	329	4.1	126	3.9	T-test	-1.3	.188	
(Low) Academic con	mpetence (13-15)	329	.78	177	.73	T-test	-1.9	.054	
(High) Delinquency	(13-22)	329	.001	177	.96	T-test	86	.077	
	(13-15)	329	.90	177	1.0	T-test	1.2	.217	
	(16-18)	326	.88	165	1.2	T-test	2.8	.005	
	(19-22)	326	.82	154	.60	T-test	97	.332	
(High)Number of set	xual partners (13-22)	329	2.2	177	2.9	T-test	1.0	.050	
	(13-15)	329	.99	177	1.6	T-test	2.1	.040	
	(16-18)	326	3.3	162	3.6	T-test	.60	.560	
	(19-22)	326	2.4	153	3.9	T-test	1.8	.070	

Table 5.1: Comparison between the "current partner-group" and other participants in the sample

Curr	ent	Other			Mean difference		
Ν	Mean	Ν	Mean	Test	Coefficient	Significance	
329		177		Chi-square	2.5	.115	
329	.80	127	.64	T-test	-4.0	.000	
329	41.3	177	29	T-test	3.4	.001	
	N 329 329	329 329 .80	N Mean N 329 177 329 .80 127	N Mean N Mean 329 177 329 .80 127 .64	N Mean N Mean Test 329 177 Chi-square 329 .80 127 .64 T-test	N Mean Test Coefficient 329 177 Chi-square 2.5 329 .80 127 .64 T-test -4.0	

Table 5.1: Comparison between the "current partner-group" and other participants in the sample (continue)

Note: The table compares the group of 329 participants interviewed on their relationship with a current partner (age 23-25) to the group of 177 participants who reported not having an intimate partner at the time of the interview (age 23-25). The tests used for this computation are the independent sample t-test (for continuous variables) and the Chi-square test (for dichotomous variables). The variables "interpersonal conflict" and "age at entering the relationship" were not included in this analysis because participants were asked about the two matters only if interviewed on their relationship with a current partner. Coefficients are considered significant when p-value is < .050. A weight was used through the analysis. The table shows the N for each variable included in the analysis. The missing cases for each variable may be due to skipped interviews, refusal to answer the question, or computer error. The missing cases for each variable are also reported in Chapter 4 (section 4.5).

The comparison of means between the two groups (as reported in Table 5.1) indicates that participants who reported being involved in a relationship with a current partner by age 23-25, were, on average, less delinquent during late adolescence (age 16-18) than participants who reported not having a romantic relationship at the time of the interview (age 23-24). The difference in levels of delinquency between the two groups is statistically significant with pvalue < .010. Also, participants involved in an intimate relationship by age 23-25 reported having, on average, a smaller number of heterosexual partners throughout their life span (age 13-22) and especially during early adolescence (age 13-15) than participants who were not involved in a current romantic relationship by age 23-25. The analysis reported on Table 5.1 shows that the difference in number of sexual partner between the two groups is statistically significant with p-value < or =.05. Also, participants involved in a romantic relationship by age 23-25 reported having a higher SES in adulthood than those who reported not having a romantic relationship at the time of the interview (age 23-25). As reported on Table 5.1, the difference in SES between the two groups is statistically significant with p-value < .001. Finally, the analysis shows that participants involved in a romantic relationship with a current partner at the time of the interview (age 23-25) reported consuming alcohol more frequently than participants who did not have a romantic partner at the time of the interview (age 23-25). Table 5.1 shows that the difference in alcohol consumption between the two groups is statistically significant with p-value < or = .001.

The Chi-square test for the racial differences between the two groups, White and non-White (as reported on Table 5.1) indicates that there are not statistically significant racial differences between the two compared groups.

In summary, the results of the analysis developed to compare participants who reported being involved in a romantic relationship by age 23-25 and those who reported not having a

current romantic relationship at the time of the interview, show that some statistically significant differences can be observed between the two groups. This confirms what the results of the descriptive analysis developed in Chapter 4 had anticipated²⁴. I will exclusively focus on the subsample of 329 participants interviewed on their relationship with an intimate partner (age 23-25) for the multivariate analysis.

5.3 Bi-variate associations (between intimate partner abuse and each risk factor)

In this part of the analysis I verify the statistical significance of the association between the dependent variable and each risk factor (or independent variable) taken into consideration for this research. The dependent variable intimate partner abuse (age 23-25) is dichotomous and the explanatory variables are continuous; therefore, I compute the bi-variate correlations using the Pearson's coefficient. Only one of the control variables included in the analysis is recoded as a dichotomy (race); hence, to test the bivariate association between intimate partner abuse and race, I compute the cross-tabulation and use the Chi-square statistics. Table 5.2 summarizes the results of the analysis.

²⁴ Chapter 4 reports a comparison between the group of 329 participants from the PYS Oldest sample interviewed on their relationship with a current partner and the 506 males of the original sample (PYS Oldest sample).

Explanatory Variable	Test	Value	Significance	N	
(High) Interpersonal Conflict (23-25)		Correlation	01	.831	313
Frequent Corporal pu	Correlation	.12	.004	329	
(Low) Parental bondi	ng (13-15)	Correlation	11	.043	328
(Negative) Attitudes	toward women (23-25)) Correlation	.16	.003	329
(Low) Academic com	Correlation	06	.231	329	
(High) Delinquency ((13-22)	Correlation	.16	.004	329
	(13-15)	Correlation	.07	.211	329
	(16-18)	Correlation	.01	.115	326
	(19-22)	Correlation	.23	.006	326
(High) Number of sex	xual partners (13-22)	Correlation	.15	.005	329
	(13-15)	Correlation	.09	.117	329
	(16-18)	Correlation	.12	.024	326
	(19-22)	Correlation	.10	.071	326
(Non-White) Race		Chi-square	10.5	.001	329
(Low) SES (23-25)	Correlation	10	.063	329	
(Younger) Age at ent	ering the relationship	Correlation	16	.003	324
(High) Frequency of	alcohol use	Correlation	.07	.181	329

Note: The table shows the bivariate statistical associations between the independent variable intimate partner abuse and each risk factor for abuse. The test used for the analysis is specified in the second column. A weight was applied to the analysis. Coefficients are considered significant if p-value < .050. The N for each variable is reported in the last column of the table. The sub-sample of participants interviewed on their relationship with an intimate partner comprises in total 329 males. There are a few missing cases for the variables interpersonal conflict, parental bonding, delinquency, number of sexual partners, and age at entering the relationship. Missing cases might be due to skipped interviews, refusal to answer the question, or computer error.

Table 5.2 shows that there is a positive statistically significant correlation between the variable experience of corporal punishment (age 13-15) and intimate partner abuse (age 23-25), which indicates that the higher the frequency of corporal punishment received during early adolescence the higher the participant's likelihood of being an adult intimate partner abuser. Also, there is a negative statistically significant correlation between parental bonding during early adolescence (age 13-15) and adult intimate partner abuse; this indicates that the lower the level of bonding to parents during early adolescence, the higher the likelihood of abusing an intimate partner in adulthood. Table 5.2 also indicates that there is a positive statistically significant correlation between attitudes toward women/relationship with women (age 23-25) and adult intimate partner abuse (age 23-25). This means that the more a participant has negative attitudes toward women and relationships with women, the more likely he is to abuse an intimate partner in adulthood. Also, there is a positive statistically significant association between delinquency (age 13-22) and adult intimate partner abuse (age 23-25). When age-groups are taken into consideration, the analysis indicates that delinquent behavior in pre-adulthood (age 19-22) is statistically significantly associated with adult intimate partner abuse (age 23-25). This indicates that more delinquent individuals are also more likely to perpetrate abuse against their intimate partners. There is also a positive statistically significant association between number of sexual partners (age 13-22) and adult intimate partner abuse. When age groups are taken into consideration, the analysis shows that having more heterosexual partners during late adolescence (age 16-18) might increase the likelihood to perpetrate abuse against intimate partners in adulthood. This might suggest that more promiscuous individuals tend to abuse their intimate partners more often than those who are less promiscuous.

The bivariate statistical associations also show that both race and age at entering the romantic relationship are negatively associated to intimate partner abuse. More specifically, being non-White increases the likelihood of being abusive against intimate partners; and the younger the age at which participants have entered the romantic relationship, the more likely they are to perpetrate abuse against their intimate female partners (age 23-25).

The variables interpersonal conflict with one's partner, academic competence during early adolescence, delinquency (age 13-15 and age 16-18), number of sexual partners (age 13-15 and age 19-22), SES, and frequency of alcohol use are not significantly correlated to the dependent variable intimate partner abuse (age 23-25) with p-value > .05.

The analysis of bi-variate associations has been useful to better understand how each explanatory variable included in the analysis (risk factor of intimate partner abuse) is linked to the dependent variable intimate partner abuse. In the next section I will focus on the multivariate statistical analysis and I will test the 5 hypotheses developed in Chapter 4.

5.4 Multivariate analysis: Testing hypotheses

The multivariate analysis is divided into three different parts. In the first part I investigate risk factors of intimate partner abuse by category of influence (family, socio-cultural, and individual/personal influences) and I test hypotheses 1 through 4 as specified in Chapter 4. In the second part of the multivariate analysis I investigate the role of risk factors of intimate partner abuse by age-group (age 13-15, age 16-18, and age 19-22), and I test hypothesis 5 as specified earlier in the previous Chapter. In the third part of the analysis I design a final model by taking into consideration only those risk factors of intimate partner abuse statistically significant in part 1 and part 2 of the multivariate analysis.

5.4.1 Part 1: Analysis of risk factors of intimate partner abuse by category of influence

In this first part of the multivariate analysis, risk factors of intimate partner abuse are investigated in association with the dependent variable by category of influence (family, socio-cultural, individual/personal influences). The empirical models used to develop this first part of the analysis explore how factors considered by category of influence affect the likelihood of perpetrating abuse against an intimate partner in adulthood (age 23-25).

Family influence (Testing hypothesis 1)

In Chapter 4 I hypothesized that males who have experienced conflict within an intimate relationship (age 23-25), have experienced frequent corporal punishment in the family of origin (age 13-15), or have had weak bonds with parents/caretakers (age 13-15) are more likely to abuse an intimate partner in adulthood (age 23-25). This hypothesis refers to the risk factors identified by proponents of the family violence perspective (discussed in Chapter 3) that have been the focus of major controversies. In order to test this hypothesis I use a binary logistic regression model. The binary logistic regression model is the most appropriate empirical model because it allows me to assess whether the odds of the observed values of the dependent variable (intimate partner abuse) are determined by the observed values of the explanatory variables (risk factors of abuse). Furthermore, this model allows me to use a dependent variable that has been recoded as a dichotomy.

In the model the dependent variable is "intimate partner abuse" perpetrated against a current partner (age 23-25) and the explanatory variables are all the risk factors categorized as family influence, i.e., the probability of intimate partner abuse (age 23-25) = Exp (β X) / [1 + exp (β X)] where X includes an intercept (α), interpersonal conflict (age 23-25), experience of

corporal punishment (age 13-15), parental bonding (age 13-15), four control variables (race,

SES, age at entering the romantic relationship, and alcohol use).

Table 5.3 shows the results of the statistical analysis.

Significance 95% C.I. Exp (B
4 .420 (.86; 1.1)
.020 (1.2; 6.3)
0 .043 (.17; .97)
.006 (1.2; 3.6)
0 .806 (.39; 2.1)
5 .003 (.77; .95)
.185 (.99; 1.0)

Table 5.3: Family influence model

N = 311

Pseudo R-square = .100

Note: The variables corporal punishment (age 13-15), and parental bonding (age 13-15) are statistically significant with p-value < .050; also, the control variables race and age at the beginning of the relationship are significant at p-value < .010. A weight was used throughout the analysis. The analysis focuses only on the sub-sample of 329 participants interviewed on their relationship with a current partner (age 23-25). In total 18 cases are identified as missing, due to skipped interview, refusal to answer, or failure to remember (like in the case of the variable "age at the beginning of the romantic relationship that was investigated in retrospective).

Table 5.3 reports the findings of the analysis conducted by using a binary logistic regression model to test hypothesis 1 as specified in Chapter 4. The analysis is limited to the 329 participants from the PYS "Oldest Sample" that were interviewed on their relationship with a current partner by age 23-25. However, 18 missing cases are evident from the N reported in Table 5.3. In Chapter 4, Table 4.5 specified the missing cases for each variable included in the analysis. Table 5.3 shows that participants' experience of corporal punishment, and bonding with parents/caretakers are statistically significant in the analysis even when controlling for race, SES, age at entering the relationship, and frequency of alcohol use. Among the control variables race and age at the beginning of the relationship are statistically significant.

The B coefficient for the variable corporal punishment is positive and it is significant (p-value < .050). Hence, the analysis shows that the more frequent corporal punishment the participant received during early adolescence (age 13-15), the more likely to perpetrate abuse against an intimate partner during adulthood (age 23-25). Also, the 95% confidence interval for the exponential B shows that increasing the frequency of corporal punishment by 1 unit multiplies the odds of abusing an intimate partner by a value between 1.12 and 6.3.

The B coefficient for the variable parental bonding is negative and it is significant with pvalue < than .05, which indicates that the weaker the bond with one's parents during early adolescence, the more likely the participant is to abuse an intimate partner in adulthood (age 23-25). Thus, as I had hypothesized in Chapter 4, participants who experienced frequent corporal punishment in the family of origin (age 13-15) and those who suffered from weak bonding with parents/caretakers were more likely to perpetrate abuse against an intimate partner during adulthood (age 23-25). Contrary to my hypothesis, the variable interpersonal conflict (age 23-25) is not significant in the analysis.

Among the control variables, age at the beginning of the relationship and race are statistically significant. More specifically, the B coefficient for the variable age at the beginning of the relationship is negative and it is significant with p-value < .010. This indicates that being younger at the beginning of the relationship increases the likelihood of abusive behaviors against an intimate partner in adulthood (age 23-25). The B coefficient for the variable race is positive and is also significant with p-value < .010. The 95% confidence interval for the exponential B shows that being non-white multiplies the odds of abusing a partner in adulthood (age 23-25) by a value between 1.2 and 3.6. Surprisingly, the variable SES is not significant. As indicated by statistical findings, race and SES are moderately correlated; therefore it is possible that the effect of race tends to wash out the effect of SES in the multivariate analysis. More specifically, race and SES are negatively correlated, that is, non-White people are more likely than White people to have a low SES; the Pearson's coefficient for the association between the two is -.328 and it is significant at p < .001. This confirms existing sociological theories that explain how race and SES are intertwined and it is impossible to understand the one without observing the other (Sokoloff and Pratt, 2005). The variable frequency of alcohol use included in the analysis as control variable is not statistically significant.

As already specified, Table 5.3 clearly shows that non-White participants are more likely than White participants to be involved in abusive intimate relationships. In order to better understand the role of the variable race in the analysis, I have tested the same model separating the two racial groups (White and non-White), computing a logistic regression for each group. This method was also suggested by literature developed by PYS researchers (see for instance Wei, 1998).

Table 5.4 summarizes the findings of the comparison between the two models considered by racial group.

	Whit	e Participants		non-V	non-White Participants		
Variable	В	Significance	C.I. Exp (B)	В	Significance	C.I. Exp (B)	
Explanatory Variables							
(High) Interpersonal conflict (23-25)	.04	.018	(.86; 1.2)	12	.100	(.77; 1.0)	
(Frequent) Corporal punishment (13-15)	88	.306	(.08; 2.2)	1.9	.001	(2.3; 21.3)	
(Low) Parental bonding (13-15)	-1.3	.050	(.07; 1.0)	96	.125	(.11; 1.3)	
Control Variables							
(Low) SES (23-25)	-2.1	.018	(.02; .70)	.65	.205	(.70; 5.2)	
(Young) Age at beginning of relationship	12	.162	(.74; 1.0)	19	.005	(.72; .94)	
(High) Frequency of alcohol use	01	.866	(.99; 1.0)	.01	.064	(1.0; 1.1)	
N = 142				N = 1	69		
Pseudo R-square = .080				Pseudo R-square = .135			

Table 5.4: Family influence model – Analysis by racial group

Note: The table shows the different results for the two models that compare the 329 participants interviewed on their relationship with an intimate partner by racial group. The B Coefficients are considered significant when their p-value is < .050. A weight was used throughout the analysis. The total N for the two groups is 311 (same as per Table 5.3). There are 18 missing cases, which might be due to skipped interviews, refusal to answer questions, or computer error. The missing cases for each explanatory variable included in the analysis are specified in table 4.5.

Table 5.4 shows that for the White racial group, participants who experienced conflict with their intimate partner (age 23-25), those who had weak bonds with parents/caretakers and those with a lower SES were more likely to become abusive in adulthood (age 23-25). More specifically, the B coefficient for the variable interpersonal conflict with partner (age 23-25) is positive and it is significant with p-value < .050. The analysis indicates that the higher the level of interpersonal conflict with one's intimate partner the more likely a White participant is to perpetrate abuse against his intimate partner (age 23-25). The B coefficient for the variable parental bonding is negative, but it is only marginally significant (p-value = .050). Among the control variables, the participant' SES is the only statistically significant one. The B coefficient for the variable SES is negative and significant with p-value < .05. This indicates that the lower a White participant' SES, the higher his likelihood to perpetrate abuse against an intimate partner in adulthood (age 23-25). In total 142 participants were included in the White racial group.

In contrast, for the non-White participants, the statistical results show that participants who experienced frequent corporal punishment in the family of origin (age 13-15) and those who entered the relationship at a younger age were more likely to become abusive against their intimate female partner in adulthood (age 23-25). More specifically, the B coefficient for the variable corporal punishment is positive and it is significant with p-value < .001. This shows that the more frequent a non-White participant's experience of corporal punishment during early adolescence (age 13-15) the higher his likelihood to perpetrate abuse against an intimate partner in adulthood (age 23-25). The 95% confidence interval for the exponential B shows that by increasing the frequency of corporal punishment by 1 unit multiplies the odds of abusing an intimate partner by a value between 2.3 and 21.3. The B coefficient for the variable age at the beginning of the relationship is negative and it is significant with p-value < .010. This indicates

that the younger a non-White participant is at the beginning of the romantic relationship, the more likely he is to perpetrate abuse against an intimate partner in adulthood (age 23-25). In total 169 participants were included in the non-White group.

<u>Socio-cultural influence</u> (Testing hypothesis 2)

Earlier in this dissertation (see Chapter 4) I have hypothesized that males involved with an intimate partner who have negative attitudes toward women and toward relationships with women (age 23-25) are more likely to be abusive against their intimate partners in adulthood (age 23-25). The statement summarizes the main risk factor of intimate partner abuse identified by proponents of the mainstream feminist perspective. Although thoroughly investigated in field work studies, the variable attitudes toward women/relationships with women has been often neglected in quantitative/empirical papers. To test this hypothesis I use a binary logistic regression model where the dependent variable is intimate partner abuse perpetrated against a current partner (age 23-25) and the explanatory variables are all the risk factors identified as socio-cultural influence, i.e., probability of intimate partner abuse (age 23-25) = Exp (β X) / [1 + $\exp (\beta X)$ where X includes an intercept (α), attitudes toward women and relationships with women (age 23-25), and four control variables (race, SES, age at entering the romantic relationship, and alcohol use). The binary logistic regression is the most appropriate empirical model because it allows me to investigate the participant's likelihood of being abusive against an intimate partner in adulthood (age 23-25) with a model that uses a dichotomous dependent variable (intimate partner abuse).

Table 5.5 summarizes the findings of the binary logistic regression used to investigate the socio-cultural influence model.

Variables in the Equation	В	Significance	95% C.I. Exp(B)
Explanatory Variables			
(Negative) Attitudes toward women (23-25)	.41	.017	(1.1; 2.1)
<u>Control Variables</u>			
(Non-White) Race	.65	.013	(1.1; 3.2)
(Low) SES	45	.264	(.29; 1.4)
(Young) Age at beginning of relationship	14	.006	(.79; .96)
(High) Frequency of alcohol use	.01	.449	(.99; 1.0)
N = 324			
Pseudo R-square = .079			

Table 5.5: Socio-cultural influence model

Note: The variable attitudes toward women (23-25) and race are significant with p-value < .050 and age at the beginning of relationship is statistically significant at p-value < .010. A weight was used throughout the analysis. The N for the analysis is 324, which shows that 5 of the 329 cases for the analysis are considered missing. The missing cases might be due to skipped interviews, refusal to answer the questions, or computer error. Chapter 4 (section 4.5) specifies that the variable age at the beginning of the relationship carries 5 missing cases.

Table 5.5 shows the results of the statistical analysis for the socio-cultural influence model that includes a total of 324 participants from the sub-sample 329 males interviewed on their relationship with an intimate partner (age 23-25). The analysis shows that 5 participants were considered missing. In Chapter 4 (table 4.5) it was specified that the variable age at the beginning of the relationships carries 5 missing cases. Table 5.5 indicates that participants who had negative attitudes toward women and relationships with women (age 23-25) were more likely than participants who did not to abuse an intimate female partner (age 23-25). More

specifically, the B coefficient for the variable attitudes toward women is positive and it is significant with p-value < .050. This indicates that the more negative the attitudes toward women the more likely a participant is to abuse an intimate partner in adulthood (age 23-25). The existing literature on intimate partner abuse has for many years explained that sexist values and gendered expectations among males and females in society are acquired within one's network (family, workplace, recreational circles, and community at large) and they reflect the characteristics of the environment in which the individual grew up (Dobash and Dobash, 1998). Although field work has provided a great contribution in explaining how cultural values affect men's relationships with women in intimate contexts, researchers have so far struggled to find a way to measure this association using a quantitative type of information (empirical studies). The PYS provides the unique opportunity to verify the role of the variable attitudes toward women and relationships with women in this research. In the model, the association between the explanatory variable attitudes toward women (age 23-25) and the dependent variable intimate partner abuse (age 23-25) is controlled by the three demographics (race, SES, and age at entering the relationship) and by the variable frequency of alcohol use. As in the previous model, race and age at entering the relationship are statistically significant. And like in the empirical model used to test family risk factors, non-White participants were more likely than their White peers to abuse their female intimate partner during adulthood (age 23-25). The B coefficient for the variable race is positive and it is significant with p-value < .050. Also, participants who entered the romantic relationship at a young age were at higher risk of developing abusive behaviors against their intimate partner in adulthood (age 23-25). The B coefficient for the variable age at the beginning of the romantic relationship is negative and it is significant with p-value < .010. The findings of this model confirm my hypothesis.

As I have already mentioned, Table 5.5 has indicated that race is a statistically significant risk factor for intimate partner abuse in adulthood. In order to better understand the role of race in determining the participants' probability to abuse an intimate partner I develop two different binary logistic regressions that allow me to compare the two racial groups (White and non-White). This methodology was also suggested by the literature developed by PYS researchers (Wei, 1998). Table 5.6 summarizes the findings for the two logistic regressions that analyze the socio-cultural model for the White and non-White groups.

White	e Participants	icipants			non-White Participants		
B	Significance	C.I. Exp (B)	В	Significance	C.I. Exp (B)		
.80	.058	(.97; 5.1)	.31	.102	(.94; 2.0)		
-1.8	.034	(.03; .87)	06	.893	(.39; 2.3)		
13	.135	(.75; 1.0)	15	.017	(.76; .97)		
00	.999	(.98; 1.0)	.00	.286	(.99; .01)		
				N = 177			
Pseudo R-square = $.061$							
	B .80 -1.8 13	.80 .058 -1.8 .034 13 .135	B Significance C.I. Exp (B) .80 .058 (.97; 5.1) -1.8 .034 (.03; .87) 13 .135 (.75; 1.0)	B Significance C.I. Exp (B) B .80 .058 (.97; 5.1) .31 -1.8 .034 (.03; .87) 06 13 .135 (.75; 1.0) 15 .00 .999 (.98; 1.0) .00	B Significance C.I. Exp (B) B Significance .80 .058 (.97; 5.1) .31 .102 -1.8 .034 (.03; .87) 06 .893 13 .135 (.75; 1.0) 15 .017 00 .999 (.98; 1.0) .00 .286 N = 177		

Table 5.6: Socio-cultural influence model - Analysis by racial group

Note: The table compares the results of the logistic regression for the White and non-White groups. The B coefficients are considered significant when their p-values are < .050. A weight was used throughout the analysis. The total N for the two regressions is 324 (same as table 5.5), which indicates that 5 cases are considered missing in the analysis. There were 5 missing cases for the variable age at beginning of the relationship as specified in Chapter 4 (table 4.5). Only the 329 participants interviewed on their relationship with an intimate partner (age 23-25) are considered for the analysis.

Table 5.6 shows that for the White racial group, the participants who had a lower SES in adulthood (age 23-25) were more likely than participants with a higher SES to perpetrate abusive behaviors against their intimate partner (age 23-25). The B coefficient for the variable SES is negative and it is significant with p-value < .050. Instead, the analysis for the non-White racial group indicates that the younger the participant the more likely he is to perpetrate abuse against an intimate partner in adulthood (age 23-25). The B coefficient for the variable age at entering the romantic relationship is negative and it is significant with p-value < .050. In the original socio-cultural risk factors model, where the variable race is included as a control variable, individuals with negative attitudes toward women and relationships with women were more likely than those who did not have negative attitudes to perpetrate abusive behaviors against their intimate partner (age 23-25). However, when the two racial groups are considered separately the influence of the variable "attitude toward women" is not significant for the non-White group and it is only marginally significant for the White group (with p-value slightly greater than .050). This might be due to the small samples size once the two groups are split. In total 148 White participants and 176 non-White participants were included in the analysis of socio-cultural risk factors of intimate partner abuse. The five missing cases from the 329 group of participants interviewed on their relationship with an intimate partner (age 23-25) are all within the non-White group. As specified in Chapter 4, the variable age at the beginning of the relationship carries 5 missing cases, due to either skipped interviews, refusal to answer the question, or failure to remember information (in this case the question was asked in retrospective).

Individual/personal influence (Testing hypothesis 3)

In Chapter 4 I have hypothesized that males with a lower academic competence (age 13-15), those who were delinquent (age 13-22), and those who had larger numbers of female sexual partners (age 13-22) were more likely to abuse their intimate partner during adulthood (age 23-25). This statement attempts to investigate the risk factors of intimate partner abuse as suggested by proponents of the life-course perspective. Furthermore, this hypothesis provides me with a theoretical basis for exploring whether individual/personal characteristics (such as, academic competence, delinquency, and number of sexual partner) influence an individual propensity of abusing an intimate partner in adulthood (age 23-25). In order to test this hypothesis I use a binary logistic regression where the dependent variable is "intimate partner abuse" perpetrated against a current partner and the explanatory variables are all the risk factors categorized as individual/personal influence, i.e., probability of intimate partner abuse (age 23-25) = Exp (β X) $/ [1 + \exp(\beta X)]$ where X includes an intercept (α), academic competence (age 13-15), delinquency (age 13-22), number of sexual partners (age 13-22), and four control variables (race, SES, age at the beginning of the relationship, and alcohol use). Like for the previous hypotheses, the binary logistic regression seems to be the most appropriate empirical model because it allows me to investigate the individual's probability of abusing a partner and it also allows me to employ a dependent variable that is dichotomous (0 non abuse, 1 abuse).

Table 5.7 (below) reports the statistical findings of the binary logistic regression for the individual/personal influence model.

Variables in the Equation	В	Significance	95% C.I. Exp (B)
Explanatory Variables			
(Low) Academic competence (age 13-15)	.01	.995	(.38; 2.7)
(High) Delinquency (age 13-22)	.35	.062	(.98; 2.0)
(High) Number of sexual partners (age 13-22)	.06	.231	(.96; 1.2)
Control Variables			
(Non-White) Race	.61	.025	(1.1; 3.1)
(Low) SES (age 23-25)	27	.517	(.33; 1.8)
(Young) Age at beginning of relationship	15	.003	(.78; .95)
(High) Frequency of alcohol use	.01	.942	(.99; 1.0)
N = 324			
Pseudo R-square = .083			

Table 5.7: Individual /personal influence model

Note: The table shows the results of the logistic regression to test individual/personal risk factors of intimate partner abuse (age 23-25). The variable delinquency (13-22) and the variable race are statistically significant at p-value < .050 while the variable age at the beginning of the relationship is significant at p-value < .010. A weight was used throughout the analysis. The N for the analysis is 324, which indicates that 5 of the 329 participants interviewed on their relationship with an intimate partner (age 23-25) are considered missing (due to skipped interviews or refusal to answer the question). The analysis exclusively focuses on the 329 participants of the Oldest Sample interviewed on their relationship with a current intimate partner at the time of the interview (age 23-25).

Table 5.7 shows that none of the explanatory variables included in the analysis are statistically significant in the individual/personal influence model. This does not confirm my initial hypothesis and it also contrasts with the results of other life-course studies on intimate partner abuse (Moffitt and Caspi, 1999; Gorman-Smith et al., 2001; Woodward et al., 2002; Farrington, 1994; 1997). However, no conclusion can be drawn at this point of the statistical analysis; the association between individual/personal risk factors of adult intimate partner abuse needs to be further investigated.

Among the control variables in the model race and age at the beginning of the relationship are statistically significant. More precisely, the B coefficient for the variable race is positive and it is significant with p-value < .050. This indicates that non-White participants are more likely than White participants to abuse their intimate partners in adulthood. The 95% confident interval for the exponential B indicates that when the variable race takes value 1 (that is, non-White) multiplies the odds of being abusive against an intimate partner by a value between 1.1 and 3.1. The B coefficient for the variable age at entering the relationship is negative and it is significant with p-value < .010. The analysis shows that the younger the participant was at the beginning of the romantic relationship, the more likely he is to abuse an intimate partner in adulthood (age 23-25). The variable frequency of alcohol use is not statistically significant.

Although it was found that non-White participants are more likely than their White peers to abuse their intimate partners in adulthood (age 23-25), it is important to further investigate the role of race in the analysis. I test the same individual/personal influence model by using two separate binary logistic regressions, one for the White group and one for the non-White group. As earlier specified this strategy was suggested by literature developed by PYS researchers (see

for instance Wei, 1998). Table 5.8 shows the results of the two logistic regressions individual/personal influence model specifying the analysis by racial group.

White Participants				non-White Participants			
Variable	B	Significance	C.I. Exp (B)	В	Significance	C.I. Exp (B)	
Explanatory Variables							
(Low) Academic competence (13-15)	13	.870	(.17; 4.3)	.24	.708	(.37; 4.4)	
(High) Delinquency (13-22)	.51	.111	(.89; 3.1)	.32	.195	(.85; 2.2)	
(High) Number of sexual partners (13-22)	11	.513	(.64; 1.3)	.08	.161	(.97; 1.2)	
<u>Control Variables</u>							
(Low) SES (23-25)	-1.5	.099	(.04; 1.3)	.03	.959	(.40; 2.6)	
(Young) Age at beginning of relationship	15	.065	(.73; 1.0)	16	.012	(.76; .97)	
(High) Frequency of alcohol use	00	.858	(.99; 1.0)	.00	.709	(.99; .01)	
N = 148				N = 1	76		
Pseudo R-square = .064				Pseudo R-square = .073			

Table 5.8: Individual/personal influence model – Analysis by racial group

Note: The table shows the results of the analysis for the individual/personal influence model by racial group (White and non-White). The coefficients are considered significant when their p-values are < .050. A weight was applied throughout the analysis. The actual N is 148 for the White group and 176 for the non-White group. The missing cases are 5 as in the previous model (see table 5.7). Missing cases might be due to either skipped interviews, refusal to answer the question, or computer error. The analysis takes into consideration the sub-sample of 329 (from the original 506 PYS sample) participants interviewed on their relationship with an intimate partner (age 23-25).

Table 5.8 shows that for the White group, none of the explanatory variables included in the analysis is actually significant. However, for the non-White group the analysis shows that participants who entered the relationship at a younger age were more likely to become abusive against their intimate partner by age 23-25. The B coefficient for the variable age at the beginning of the romantic relationship is negative and it is significant with p-value < .050. This shows that the younger the participant was at the beginning of the relationship, the more likely he is to perpetrate abuse against his intimate partner in adulthood (age 23-25). The N is 148 for the White group and 176 for the non-White group. Five cases in the analysis are considered missing. The analysis exclusively focuses on the 329 males of the PYS Oldest Sample interviewed on their relationship with an intimate current partner.

<u>Combining all the risk factors together</u> (Testing hypothesis 4)

In Chapter 4 I have specified that risk factors for abuse against intimate partners range among different categories of influence (family, socio-cultural, and individual/personal influences). I have hypothesized that family risk factors are more likely than other risk factors (socio-cultural or individual risk factors) to influence one's likelihood to abuse an intimate partner in adulthood. This statement summarizes what can be perceived as the dominant view on violent/aggressive behaviors in sociology/criminology literature (Siegel, 2006). More specifically, it is believed that experiences within the family of origin, especially during adolescence, tend to shape adult human behavior and the ability to manage interpersonal relationships (Siegel, 2006). In order to verify whether this statement is correct, I test a binary logistic regression where the dependent variable is "intimate partner abuse" against a current partner, and the explanatory factors are all the risk factors included in the three previous models (family influence, socio-cultural influence, and individual/personal influence models). The binary logistic regression model uses the Maximum Likelihood Estimator which, in this case, would allow me to assess whether the odds of the observed values of intimate partner abuse are determined by the observed values of all the risk factors of abuse included in the analysis. Differently from the previous regressions used for the multivariate analysis, in this model the explanatory variables are included by block, i.e., probability of intimate partner abuse = Exp $(\beta_1X_1 + \beta_2X_2 + \beta_3X_3 + \beta_4X_4) / [(1 + \exp \beta_1X_1 + \beta_2X_2 + \beta_3X_3 + \beta_4X_4)]$ where X₁ includes the four control variables (race, SES, age at entering the relationship, and alcohol use), X₂ includes interpersonal conflict (age 23-25), experience of corporal punishment (age 13-15), and parental bonding, X₃ includes attitudes toward women and relationships with women (age 23-25), X₄ includes academic competence (age 13-15), delinquency (age 13-22), and number of sexual partners (age 13-22). Furthermore, the Chi-square test (used in the Block model) allows me to understand how each group of risk factors of abuse contributes to explain the variance within the model. Table 5.9 reports the findings of the statistical analysis for this model.

Equations	Control Variables	influence	ocio-cultural influence	Individual/Personal influence				
Variables	(1)	(2)	(3)	(4)				
Race	.78** (1.3; 3.7)	.75** (1.3; 3.7)	.72** (1.2; 3.5)	.66* (1.1; 3.4)				
(Low) SES (age 23-25)	41 (.30; 1.4)	10 (.39; 2.1)	15 (.37; 1.9)	14 (.34; 2.1)				
(Young) Age at beginning of relationship	12* (.80; .97)	15** (.78; 95)	15** (.78; .95)15** (.77; .95)				
(High) Frequency of alcohol use	.00 (.99; 1.0)	.00 (.99; 1.0)	.00 (.99; 1.0)	.00 (.99; 1.0)				
Chi-square = 22.190 (5 degrees of freedom) p-value <.001								
Pseudo R-square = .068								
(High) Interpersonal conflict (age 23-25)		04 (.86; 1.0)	05 (.85; 1.0)	04 (.86; 1.1)				
(Frequent) Corporal punishment (age 13-15)		1.0* (1.2; 6.3)	.97* (1.1; 6.2)	.87* (1.0; 5.7)				
(Low) Parental bonding (age 13-15)		90* (.17; .97)	89* (.17; .99)	79 (.18; 1.1)				
Chi-square improvement = 10.67 (3 degrees o	f freedom) p-value < .010)						
Pseudo R-square = $.100$								

 Table 5.9: Analysis of risk factors of intimate partner abuse by category of influence – Logistic regression (N = 311)

Equations	Control Variables	Family influence	Socio-cultural influence	Individual/Personal influence				
Variables	(1)	(2)	(3)	(4)				
	5)		2(*(1,0,2,0))	22(0.00, 1, 0)				
(Negative) Attitudes toward women (age 23-2	.36* (1.0; 2.0)	.33 (.98; 1.9)						
Chi-square improvement 4.403 (1 degree of fr	eedom) p-value < .050							
Pseudo R-square = .112								
(Low) Academic competence (age 13-15)			.10 (.41; 3.0)					
(High) Delinquency (age 13-22)			.18 (.81; 1.8)					
(High) Number of sexual partners (age 13-22)		.05 (.95; 1.2)						
Chi-square improvement 2.204 (3 degrees of freedom) p-value > .100								
Pseudo R-square .118								
N = 311								

Table 5.9: Analysis of risk factors of intimate partner abuse by category of influence – Logistic regression (N = 311) (continue)

Note: For each equation the B value is reported with level of significance identified by (*). The p-value is * for alpha = .05; ** for alpha = .01; and, *** for alpha = .001. The value reported into parentheses is the 95% C.I. for each equation. The N for the analysis is 311. Only the 329 participants interviewed on their relationships with a current intimate partner (age 23-25) were considered. However, the table shows that 18 cases were missing in the analysis.

Table 5.9 shows the empirical findings for the model used to test hypothesis number 4 (as specified in Chapter 4, section 4.6). The analysis takes into consideration information gathered among the 329 participants of the PYS Oldest Sample interviewed on their relationship with a current intimate partner (age 23-25). However, the table shows that 18 of the 329 participants are considered missing in the analysis. The missing cases for each variable included in the statistical analysis are specified in Chapter 4 (Table 4.5).

Table 5.9 indicates that for the first block of variables (control variables) the Chi-square is 22.190 with 5 degrees of freedom²⁵ and it is significant with p-value < .001. The variable race and the variable age at entering the romantic relationship are statistically significant. The B coefficient for the variable race in the model is positive and it is significant with p-value < .010. This indicates that non-White participants are more likely than White participants to be abusive against an intimate partner in adulthood (age 23-25). The B coefficient for the variable age at the beginning of the relationship is negative and it significant with p-value < .010. That is, the younger the participant was at the beginning of the romantic relationship, the more likely he is to abuse his intimate partner in adulthood (age 23-25). The variables SES and frequency of alcohol use are not statistically significant in the analysis. These results confirm the findings of the analysis conducted separately for the three models (family influence, socio-cultural influence, and individual/personal influence models) reported in Tables 5.3, 5.5, and 5.7. When the second block of variables is added (family risk factors) the Chi-square value increases to 32.861; thus, the Chi-square improvement is approximately equal to 11 and it is statistically significant with pvalue < .010 (with three degrees of freedom). Among the explanatory variables, corporal punishment and parental bonding are both statistically significant. The B coefficient for corporal punishment is positive and it is significant with p-value < .050. This shows that the more

²⁵ The number of degrees of freedom for equation number 1 includes the four control variables and the constant.

frequent the corporal punishment during early adolescence (age 13-15) the more likely the participant is to perpetrate abuse against an intimate partner in adulthood (age 23-25). The B coefficient for the variable parental bonding during early adolescence (age 13-15) is negative and it is significant with p-value < .050. This indicates that the weaker the bond between the participant and his parents/caretakers during early adolescence (age 13-15) the more likely the participant is to perpetrate abuse against an intimate partner in adulthood (age 23-25). These findings confirm the results reported in table 5.3, 5.5, and 5.7 where each model was tested separately. Among the control variables race and age at the beginning of the romantic relationship are the only two statistically significant factors. As in the previous equation, the analysis shows that non-White participants and those who were younger at the beginning of the romantic relationship are more likely to be abusive against their intimate partners in adulthood (age 23-25). When socio-cultural risk factors are added to the model (third equation), the Chisquare value becomes 37.264, improving by 4.40, significant with p-value < .050 with (1 degree of freedom). The variable attitudes toward women is statistically significant. The B coefficient for attitudes toward women is positive and it is significant with p-value < .050. This indicates that participants who reported having negative attitudes toward women are more likely than those who did not to be abusive against their intimate partners in adulthood (age 23-25). This confirms the results of the analysis reported in 5.5 where the empirical model for the sociocultural risk factors was tested separately. The variables corporal punishment and parental bonding during adolescence are also statistically significant. This indicates that, even when socio-cultural factors are added to the model, the analysis still shows that participants who experienced corporal punishment and those who had a weak bond with parents/caretakers during early adolescence (age 13-15) were more likely to be abusive against an intimate partner in

adulthood (age 23-25). Among the control variables, race and age at the beginning of the romantic relationship are still statistically significant, indicating that non-White participants and those who reported being younger at the beginning of the relationship are more likely to be abusive against an intimate partner in adulthood (age 23-25). When the individual/personal risk factors are added to the model (equation 4), the Chi-square value becomes 39.468, improving by only 2.2; this improvement is not statistically significant with p-value > .100 (3 degrees of freedom). The results show that none of the individual/personal risk factors is significant in this model. This confirms the results reported in Table 5.7 where the individual/personal risk factors model was tested separately. However, in the same equation, the factor corporal punishment (age 13-15) and the control variables race and age at the beginning of the relationship are statistically significant. The B coefficient for the variable corporal punishment is positive and it is significant with p-value < .050. The 95% confidence interval for this factor indicates that by adding 1 unit to the frequency of corporal punishment during early adolescence (age 13-15) multiplies the odds of being abusive against an intimate partner in adulthood (age 23-25) by a value between 1.0 and 5.7. As in the previous equations, the B coefficient for race is positive and it is significant with pvalue < .050. Finally, the B coefficient for age at the beginning of the romantic relationship is negative with p-value < .010. These results confirm that non-White participants, those who were younger at the beginning of the relationship with an intimate partner and those who received frequent corporal punishment during early adolescence (age 13-15) were more likely to be abusive against intimate partners in adulthood (age 23-25).

In summary, Table 5.9 shows that the family risk factors are the most statistically significant in the analysis. As a matter of fact, when the family influence block is added, the Chi-square improvement is significantly higher than that of socio-cultural and individual/personal

factors. Furthermore, the variable experience of corporal punishment (age 13-15) is significant throughout the analysis, within each block added to the model. Perhaps, having experienced frequent corporal punishment during the age 13-15 produces a devastating effect and increases the chance that boys would abuse their intimate partners later in life (during adulthood). These statistical findings confirm my initial hypothesis. Among the control variables, race and age at the beginning of the relationship are consistently significant in the model. This confirms what was found in the previous models, that is, non-White participants and those who were younger at the beginning of the romantic relationship were more likely to be abusive against an intimate partner in adulthood (age 23-25).

5.4.2 Part 2: Analysis of risk factors of intimate partner abuse by age-group

In this part of the multivariate analysis, I investigate how factors considered by age-group (age 13-15; age 16-18; age 19-22) can influence the likelihood of abusing an intimate partner in adulthood (age 23-25). In this part of the analysis I test hypothesis 5 that I have specified in Chapter 4 as: Personal characteristics, experiences, and behaviors that develop and manifest early in life (during early adolescence age13-15) tend to influence the probability of being abusive during adulthood (age 23-25) more than personal characteristics, experiences, and behaviors that manifest later in life (during later adolescence and pre-adulthood, age groups 16-18 and 19-22). This hypothesis was suggested by both proponents of the family violence and the life-course perspectives, who have pointed out that it is important to investigate causes of intimate partner abuse among factors that manifest early in life (during early childhood or adolescence). In order to verify whether my hypothesis was correct, I test a binary logistic regression where the dependent variable is "intimate partner abuse" perpetrated against a current

partner and the explanatory variables are included by block, i.e., probability of intimate partner abuse = Exp $(\beta_1 X_1 + \beta_2 X_2 + \beta_3 X_3 + \beta_4 X_4) / [(1 + \exp \beta_1 X_1 + \beta_2 X_2 + \beta_3 X_3 + \beta_4 X_4]$ where X_1 includes the four control variables (race, SES, age at entering the relationship, and alcohol use), X₂ includes academic competence (age 13-15), experience of corporal punishment (age 13-15), parental bonding (age 13-15), delinquency (age 13-15), and number of sexual partners (age 13-15), X₃ includes delinquency (age 16-18), and number of sexual partners (age 16-18), and X₄ includes delinquency (age 19-22) and number of sexual partners (age 19-22). This empirical model seems to be the most appropriate to test hypothesis 5. As a matter of fact, the binary logistic regression allows me to assess whether the odds of the observed values of the dependent variable intimate partner abuse are determined by the observed values of all the explanatory variables (risk factors of abuse) included in the analysis. Furthermore, the binary logistic regression allows me to use a dependent variable that is dichotomous (abuse or non-abuse). The Chi-square test applied in the binary logistic regression with block analysis allows me to estimate the improvement of the model for each group of variables added to the model itself. This would explain how much of the variance each group of risk factors is able to explain. In this model each group of factors corresponds to an age-group. The age-group chosen for the analysis represent different developmental stages in the lives of participants. As it was specified earlier in this dissertation, age 13-15 (early adolescence) is the age at which an individual wants to discover his "identity" and a role within his social network (school, circle of friends, and neighborhood); age 16-18 (late adolescence) is the age at which an individual feels and behaves more independently, following hormonal changes and sexual drive and avoiding parental supervision; age 19-22 (pre-adulthood) is the age at which an individual attempts to loosen the ties with family members, starts being more financially independent; age 23-25 (adulthood) is the age at

which an individual starts feeling less "self-absorbed", looks for a more steady romantic relationship and attempts to settle $down^{26}$.

Table 5.10 shows the results of the analysis developed by using the age-group model.

²⁶ The developmental stages were inspired by Erik Erikson's reading on on stages of development (1994).

Equations	Control Variables (1)	Age group 13-15 (2)	Age-group 16-18 (3)	Age-group 19-22 (4)
Variables				
Race	.69** (1.2; 3.3)	.66** (1.1; 3.3)	.61* (1.1; 3.2)	.42 (.87; 2.7)
(Low) SES (age 23-25)	51 (.27; 1.3)	22 (.34; 1.9)	28 (.32; 1.8)	25 (.32; 1.9)
(Young) Age at beginning	13** (.79; .96)	15** (.78; .95)	15** (.78; 95)	18*** (.75; .93)
(High) Frequency of alcohol use	.00 (.99; 1.0)	.00 (.99; 1.0)	.00 (.99; 1.0)	00 (.99; 1.0)
Model Chi-square = 21.738 (p <.001)				
Pseudo R-square = .066				
-(Low) Academic competence (13-15)		01 (.37; 2.7)	.03 (.38; 2.8)	.10 (.39; 3.2)
(Frequent) Corporal punishment (13-15)		.84* (1.0; 5.4)	.81 (.97; 5.3)	.90* (1.0; 5.9)
(Low) Parental bonding (13-15)		77 (.18; 1.1)	72 (.19; 1.2)	83 (.17; 1.1)
(High) Delinquency (13-15)		.06 (.77; 1.4)	.05 (.72; 1.5)	.10 (.75; 1.6)
(High) Number of sexual partners (13-15)		.02 (.86; 1.2)	.02 (.85; 1.2)	.00 (.84; 1.2)
Model Chi-square Improvement = $8.33 (p > $.100)			
Pseudo R-square = .090				

Table 5.10: "Risk factors of intimate partner abuse by age group" model – Logistic regression (N = 317)

Equations	Control Variables (1)		Age-group 16-18 (3)	Age-group 19-22 (4)	
Variables					
(High) Delinquency (16-18)			00 (.72; 1.4)	44* (.44; .95)	
(High) Number of sexual partners (16-18)			.02 (98; 1.1)	.03 (.98; 1.1)	
Model Chi-square improvement = .951 (p >	.100)				
Pseudo R-square = $.093$					
(High) Delinquency (19-22)				.87*** (1.6; 3.7)	
(High) Number of sexual partners (19-22)				.04 (.95; 1.1)	
Model Chi-square improvement = 19.6 (p <	.001)				
Pseudo R-square = $.147$					
N=317					

Table 5.10: "Risk factors of intimate partner abuse by age group" model – Logistic regression (N = 317) (continue)

Note: For each equation the B value is reported with level of significance identified by a (*). The p-value is * for alpha = .05; ** for alpha = .01; and, *** for alpha = .001. The value reported into parentheses is the 95% C.I. for each equation. The N for this regression is 317. The analysis takes into consideration only the sub-sample of 329 participants interviewed on their relationship with an intimate partner (age 23-25). In total 12 cases are identified as missing cases, due to skipped interviews, refusal to answer questions, or computer error. The missing cases for each variable included in the statistical analysis are specified in Chapter 4 (section 4.5)

Table 5.10 reports the findings of the statistical analysis for risk factors of intimate partner abuse by age-group. The analysis uses information gathered from the 329 participants of the PYS Oldest Sample that were interviewed on their relationship with their current intimate partner at age 23-25. The N for the model is actually 317. In total 12 observations were identified as missing cases in the analysis.

As Table 5.10 indicates, when the first equation (represented by the control variables) is tested in the "risk factors of intimate partner abuse by age-group" model, the Chi-square is 21.738 and it is significant with p-value < .001. Both the variable race and age at entering the romantic relationship are statistically significant, and this confirms the statistical results of the analysis of risk factors considered by category of influence. The B coefficient for race is positive and it is significant with p-value < .010. This indicates that non-White males are more likely than White males to be abusive against an intimate partner in adulthood (age 23-25). The B coefficient for the variable age at entering the romantic relationship is negative and it is significant with p-value < .050. This shows that the younger the participant was at the beginning of the romantic relationship, the more likely he is to perpetrate abuse against his intimate partner in adulthood (age 23-25). The variables SES and alcohol use are not significant in the regression. When the second equation is added (factors for age-group 13-15) the Chi-square improvement is 8.33 for 5 degrees of freedom and it is not significant with p-value > .100. None of the explanatory variables included in the second equation are statistically significant. However, among the control variables, race and age at the beginning of the relationship at statistically significant. The B coefficient for the variable race is positive and it is significant with p-value < .010. This indicates that non-White participants are more likely than White participants to be abusive against an intimate partner in adulthood (age 23-25). The B coefficient for the variable

age at the beginning of the romantic relationship is negative and it is significant with p-value < .010. This shows that the younger the participant was at the beginning of the romantic relationship, the more likely he is to abuse an intimate partner in adulthood (age 23-25). When the third equation is added to the analysis (age group 16-18) the robustness of the model Chisquare improves by .952 for 2 degrees of freedom and it is not significant with p-value > .100. None of the two variables included in the analysis are statistically significant at this stage. However, the control variable race and age at the beginning of the relationship are still statistically significant .These findings indicate that non-White participants and those who were younger at the beginning of the romantic relationship are more likely to be abusive against an intimate partner in adulthood (age 23-25). When the fourth equation is added to the model (age 19-22) the robustness of the Chi-square improves by 19.6 for two degrees of freedom and it is significant with p-value < .001. The variable delinquency as measured at age 19-22 is strongly statistically significant. The B coefficient for the variable delinquency 19-22 is positive and it is significant with p-value < .001. The 95% confidence interval for the exponential B indicates that increasing the participant's level of delinquency by one unit multiplies the odds of abusing an intimate partner in adulthood (age 23-25) by a value between 1.6 and 3.6. Corporal punishment (age 13-15) and delinquency (age 16-18) are also statistically significant. The B coefficient for the variable corporal punishment (age 13-15) is positive and it significant with p-value < .050. The 95% confidence interval for the exponential B of the variable corporal punishment indicates that by adding 1 unit to the frequency of corporal punishment in early adolescence (age 13-15) multiplies the odds of being abusive against an intimate partner in adulthood (age 23-25) by a value between 1.0 and 5.9. The B coefficient for the variable delinquency age 16-18 is negative and it is significant with p-value < .05. However, no conclusions can be drawn from this result

about the link between adolescent delinquency (age 16-18) and intimate partner abuse (age 23-25). Indeed, it would be erroneous to consider delinquency age 16-18 as a "protective" factor for abuse. Among the control variables only age at the beginning of the relationship is significant. The B coefficient for the variable age at the beginning of the relationship is negative and it is significant with p-value < .001. This indicates that the younger the participant was at the beginning of the romantic relationship with partner, the more likely he is to perpetrate abuse against his intimate partner during adulthood (age 23-25).

The Chi-square analysis for the different age-groups indicates that behaviors/experiences/ personal characteristics that develop or manifest early in life (age 13-15) are neither more nor less important than behaviors/experiences/personal characteristics that manifest or develop later in life (age 16-18; and age 19-22). For instance, experience of corporal punishment (that is more likely to happen during early adolescence) is an important risk factor of adult intimate partner abuse. However, delinquency during pre-adulthood (age 19-22) seems to have a stronger impact on the likelihood of abusing an intimate partner than delinquency during early adolescence (age 13-15). These statistical results contrast with existing findings from longitudinal studies where it was found that delinquent/antisocial behavior at age 15 is one of the major risk factors for intimate partner abuse (Moffit and Caspi, 1999) and that early onset of delinquent/antisocial behavior makes certain individuals more likely to abuse an intimate partner (Woodward et al., 2002). Also, contrary to the Dunedin Study (Moffit and Caspi, 1999), this research shows that race is an important risk factor of intimate partner abuse, especially among delinquent males, and non-Whites are more likely than Whites to abuse their intimate partners.

5.4.3 Part 3: Final model

The final model for this dissertation is organized by taking into consideration only factors that showed statistical significance in the analysis of risk factors considered by category of influence (family, socio-cultural, and individual/personal influences) and in the analysis of risk factors by age-group.

Throughout the analysis the variable race (recoded as 0 White and 1 non-White) has been strongly statistically significant, indicating that non-White participants are more likely than White participants to perpetrate abuse against an intimate partner by age 23-25. In order to better understand the association between race and intimate partner abuse in the final empirical model, I run two separate equations, one for White participants and one for non-White participants²⁷. Each of these equations is a binary logistic regression where the dependent variable is "intimate partner abuse" perpetrated against a current partner and the explanatory variables are all the statistically significant factors in the analysis of risk factors considered by category of influence and in the analysis by age group. For each racial group I test the following empirical model: probability of intimate partner abuse is = exp $(\beta X) / [(1 + exp \beta X)]$ where X includes an intercept (α), the variable corporal punishment (age 13-15), parental bonding (age 13-15), delinquency (age 19-22), three control variables (SES, age at the beginning of the romantic relationship, and alcohol use). Also in this case, the binary logistic regression model seems to be the most appropriate because it allows me to assess the odds of being abusive within intimate contexts through the observed values of the explanatory variables (risk factors of abuse) with a dependent variable recoded as a dichotomy.

Table 5.11 reports the results of the two logistic regressions developed for the final model.

²⁷ This strategy was also suggested by literature developed by PYS researchers (see Wei, 1998).

	White Participants				non-White Participants			
Variable B	Significance		C.I. Exp (B)	B	Significance	C.I. Exp (B)		
Control Variables								
(Low) SES (age 23-25)	-1.6	.074	(.04; 1.1)		.49	.335	(.60; 4.4)	
(Young) Age at beginning of the relationship	17	.046	(.71; .99)		18	.007	(.74; .95)	
(High) Frequency of Alcohol use	00	.684	(.99; 1.0)		.00	.705	(.99; 1.0)	
Explanatory Variables								
(Frequent) Corporal punishment (age 13-15)	79	.342	(.09; 2.3)		1.8	.002	(1.9; 18.1)	
(Low) Parental bonding (age 13-15)	-1.0	.137	(.09; 1.4)		93	.150	(.11; 1.4)	
(High) Delinquency (age 19-22)	.69	.021	(1.1; 3.6)		.46	.019	(1.1; 2.3)	
N = 147					N = 173			
Pseudo R-square = .103					Pseudo R-square = .142			

Table 5.11: Final Model - Analysis of risk factors of intimate partner abuse by racial group

Note: The table summarizes the findings for the final model of the dissertation that takes into consideration only risk factors of intimate partner abuse that were found to be statistically significant in the first two parts of the multivariate statistical analysis (analysis by category of influence and analysis by age-group). The B coefficients are considered significant when their p-values are < .050. A weight was used throughout the analysis. The actual N for the analysis is 320 (that is, N= 147 for the White group and N= 173 for the non-White group). There are 9 missing cases in the analysis. As specified in Chapter 4 (section 4.5) the variable age at the beginning of the relationship carries 5 missing cases, the variable parental bonding carries 1 missing case, and the variable delinquency (age 19-22) carries 3 missing cases.

Table 5.11 shows the results of the statistical analysis for the empirical final model that includes two separate logistic regressions, one for the White group of participants (with N=147) and another one for the non-White group of participants (N=173). The total N for the two regressions is 320. There are 9 missing cases in the analysis.

Findings reported on table 5.11 show that for the White group, the variable age at the beginning of the relationship and delinquency age 19-22 are the only significant factors in the analysis. The B coefficient for the variable age at the beginning of the relationship is negative and it is significant with p-value < .050. This indicates that the younger a participant was at the beginning of the romantic relationship, the more likely he is to be abusive against his partner in adulthood (age 23-25). The B coefficient for the variable delinquency (age 19-22) is positive and it is significant for p-value < .050. This indicates that the more delinquent a participant was during pre-adulthood, the more likely to abuse an intimate partner later in adulthood (age 23-25).

For the non-White group, the factors age at the beginning of the relationship, corporal punishment, and delinquency (age 19-22) are all statistically significant in the binary logistic regression model. More specifically, the B coefficient for the control variable age at the beginning of the relationship is negative and it significant with p-value < .010. This shows that the younger the participant was at the beginning of the relationship, the more likely he is to perpetrate abuse against his intimate partners during adulthood (age 23-25). The B coefficient for the variable corporal punishment is positive and it is significant with p-value < .010. The 95% confidence interval indicates that increasing the frequency of corporal punishment (age 13-15) by just 1 unit multiplies the odds of being abusive in adulthood (age 23-25) by a value between 1.9 and 18.1. This shows that among the non-White participants, the more frequently corporal punishment was received during early adolescence, the more likely the participant is to abuse an

intimate partner during adulthood (age 23-25). The B coefficient for the variable delinquency (age 19-22) is positive and it is significant with p-value .050. The higher the level of delinquency during pre-adulthood (age 19-22), the more likely the participant is to abuse his intimate partners in adulthood (age 23-25).

A comparison between the two racial groups highlights that non-White abusive males in the sample were more likely than White abusive males to have experienced corporal punishment during adolescence (based on the differences between the significance levels and the odds in the two models). No other relevant differences between the two groups can be discussed at this point. The variables parental bonding, SES, and alcohol use were not significant for either of the two models.

In the next Chapter I will discuss the findings, the implications for future research, and the limitations of the study.

CHAPTER VI

DISCUSSION

6.1 Discussing the findings

This study has investigated risk factors of intimate partner abuse among 329 young males from the Oldest Sample (N=506) of the PYS who reported being involved in a romantic relationship by age 23-25. By drawing upon three theoretical perspectives (family violence, mainstream feminist, and life-course perspectives), risk factors for abuse were first analyzed by category of influence (family, socio-cultural, and individual/personal influences) and then by age groups (age 13-15, or early adolescence; age 16-18, or late adolescence; and age 19-22, or preadulthood).

Similar to other longitudinal studies (see for instance Moffitt and Caspi, 1999) the prevalence of intimate partner abuse was high: one third of participants reported having perpetrated some form of violence/abuse against their intimate partner.

Compared to other participants in the Oldest Sample, the 329 males interviewed on their relationship with an intimate partner by age 23-25 were less delinquent during adolescence (age 16-18), less sexually promiscuous (especially during early adolescence, age 13-15), more wealthy (age 23-25), and more inclined to consume alcohol from adolescence through adulthood (age 13-25).

Descriptive statistics have indicated that the independent variable intimate partner abuse (age 23-25) was strongly correlated to several of the risk factors included in the analysis, especially experience of corporal punishment (age 13-15), negative attitudes toward women (age 23-25), delinquency (19-22), number of sexual partners (age 13-22), race, and age at entering the romantic relationship.

Five hypotheses have guided the multivariate analysis. In particular, the first 4 hypotheses were part of the analysis by category of influence and the last hypothesis specifically referred to the analysis of risk factors by age-group.

The hypothesis that males who experienced conflict within an intimate relationship (age 23-25), have experienced frequent corporal punishment in the family of origin (age 13-15), or have had weak bonds with parents/caretakers during early adolescence (age 13-15) are more likely to abuse an intimate partner in adulthood (age 23-25) was only partially confirmed by the multivariate analysis. It was found that interpersonal conflict with partner (age 23-25) was not a significant factor in the analysis. Although family violence theorists have pointed out that it seems naturally to assume that abuse/violence most often arises from disagreement/interpersonal conflict (Straus and Yodanis, 1996), this is not always the case. This finding confirms Lisa Brush's theory on intimate partner abuse (Brush, 1990). While explaining the limitations of the Conflict Tactic Scale, Brush (1990) points out that most of the violence among intimate partners does not come from disagreements. As hypothesized, the variable experience of corporal punishment (age 13-15) was strongly associated with adult intimate partner abuse (age 23-25). This finding addresses a gap in the literature on intimate partner abuse. Although Straus and Yodanis (1996) found a causal link between adolescent corporal punishment and later spouse assault, they used only retrospective data from the National Family Violence Survey. Straus (1990; 2005) has suggested that researchers should investigate the link between adolescent corporal punishment and adult spouse abuse by using prospective studies. The PYS data set has allowed me to implement this suggestion and confirm the earlier result. Similarly, it was found that a weak bonding with parents/caretakers during early adolescence (age 13-15) might lead some individuals to become abusive against their intimate partners. No other longitudinal study

on intimate partner abuse had provided these results, although it was suggested by family violence theorists that lack of attachment to parents/caretakers while growing up can translate into an increased risk for later partner violence/abuse (Straus, 2005). By following the example of PYS researchers (Wei, Loeber, and Stouthamer-Loeber, 2002) the hypothesis was re-tested separating the two racial groups (White and non-White). The findings showed that risk factors for abuse do differ by racial group. More specifically, the analysis indicated that among the White participants, those who experienced conflict within their relationship with an intimate partner (age 23-25) and those who had a weak bond with parents/caretakers, were more likely to abuse their partner (age 23-25). Instead, among the non-White participants, those who had more frequent experiences of corporal punishment during early adolescence were far more likely than those who did not to abuse an intimate partner (age 23-25).

The hypothesis that males who have negative attitudes toward women and toward relationships with women (age 23-25) are more likely to abuse their intimate partners (age 23-25) was confirmed by the multivariate analysis. This finding is important because very few empirical studies have so far investigated how having certain attitudes affects the likelihood to abuse an intimate partner (Dobash and Dobash, 1998). Although it is common to investigate how participants view certain issues in field work studies, it is very difficult to operationalize abstract concepts like attitudes in empirical data sets (Johnson and Sigler, 2000). Beliefs and attitudes are learned in one's social networks (family, circle of friends, school) and they are internalized at the level of personality (Dobash and Dobash, 1998). These results make us reflect on the values in our culture and society at large. Why are many individuals learning that it is ok to be disrespectful and dominant within intimate relationships? Early educational programs should also address these issues and focus on more egalitarian principles about gender roles and

expectations. When the hypothesis was re-tested by separating the White and non-White participants, the analysis did not completely confirm the results of the analysis developed by taking into consideration all the 329 males involved in a relationship with a current partner (age 23-25). The variable negative attitudes toward women was only marginally significant for the White group, but it was not significant for the non-White group. However, as earlier specified, this might be due to the small sample size once the two groups are split.

The hypothesis that males with a lower academic competence (age 13-15), those who were delinquent (age 13-22), and those who had larger numbers of female sexual partners (age 13-22) were more likely to abuse their intimate partner during adulthood (age 23-25) was not confirmed by the multivariate statistical analysis. These results contradict the findings of other longitudinal studies. More specifically, previous studies had suggested that a low academic competence, might in some cases, be a precursor of adult intimate partner abuse. Therefore, the association between early adolescent academic competence and adult intimate partner abuse needs to be further investigated. Also, Moffit and Caspi (1999) and Woodward et al. (2002) had found that early delinquent behavior was associated to adult intimate partner abuse. And, Farrington (1994) had found that males who were sexually promiscuous were also more likely to be abusive against intimate partners. When the hypothesis was re-tested by separating the two racial groups (White and non-White) the statistical results did not change.

The hypothesis that family risk factors are more likely than other risk factors (sociocultural and individual/personal risk factors) to influence one's likelihood to abuse an intimate partner in adulthood was confirmed by the multivariate analysis. The hypothesis was tested by using a multilevel logistic regression model; the ranking among the three different categories of influence was based on both a comparison among the significance level of each risk factor taken

separately and the Chi-square improvement test used to assess the robustness of the model after adding each group of factors. In particular, the analysis confirmed that experiencing frequent corporal punishment and a low parental bonding during early adolescence might significantly increase the odds of being abusive against an intimate partner in adulthood. The evidence on the role of family factors in determining abusive behavior suggests that parenting education programs may help to mitigate children's proclivity to perpetrate abuse within intimate contexts later in life. In particular, advising parents about potential detrimental effects of corporal punishment and insubstantial bonding with children may act to reduce these behaviors.

Hypotheses 1 though 4 allowed me to investigate risk factors of intimate partner abuse by category of influence. In summary, this part of the analysis has indicated that a multi-contextual framework of analysis is useful to confirm and compare the importance of the different factors. My statistical findings strongly support the hypothesis that family factors are more important than socio-cultural or individual factors in affecting an individual's likelihood to abuse an intimate partner.

The last hypothesis specified for this dissertation guided the analysis of risk factors of intimate partner abuse as considered by age group. In particular, I had stated that personal characteristics, experiences, and behaviors that manifest early in life (during early adolescence) tend to influence the probability of abusing an intimate partner during adulthood (age 23-25) more than experiences and behaviors that manifest later in life (age groups 16-18 and 19-22). The multivariate analysis did not confirm my initial hypothesis. It was found that adolescent experience of corporal punishment (age 13-15) is a very important factor in the analysis, confirming the results of the analysis by category of influence. A strong association between delinquency in pre-adulthood (age 19-22) and adult intimate partner abuse (age 23-25) was also

found. Although prevalence rates in delinquency among the PYS participants were higher during age 13-15 than during 19-22, the statistical analysis has strongly indicated that delinquency during pre-adulthood is a much important factor in the analysis (based on a comparison of the B coefficients and 95% confidence intervals) than delinquency during early adolescence. My results contrast with the findings of previous longitudinal studies on intimate partner abuse, which have indicated that very early onset in delinquent behavior (Woodward et al, 2002) and delinquency at age 15 (Moffit and Caspi, 1999) are the most important risk factors for intimate partner abuse. However, it is important to point out that, in contrast to those studies, this research takes into consideration both violent and non-violent types of delinquency (using the PYS measure of "number of levels of delinquency"). The finding that experiences in the family of origin during early adolescence (age 13-15) and delinquency during pre-adulthood (age 19-22) are the most significant risk factors in the empirical analysis, indicating that there is not a specific age-group that is "most relevant" in the analysis. The ranking among factors considered by age-groups in the multilevel logistic regression was based on both the B coefficients for each factor and the Chi-square difference test after each group of factors was added to the model.

Throughout the analysis, four control variables were included in each empirical model: race, SES, age at the beginning of the relationship, and frequency of alcohol abuse. It was found that non-White participants were more likely than White participants to become abusive against their intimate partners. However, the analysis developed by separating the two racial groups pointed out that, although minority males tend to be more often involved in conflictual relationships and tend to be more abusive than their White peers, they are also more often exposed to frequent corporal punishment during early adolescence. This suggests that, even if race is strongly directly associated to intimate partner abuse, other factors (like experience of corporal punishment during adolescence) tend to exacerbate the association between the two. These findings suggest that it is not sufficient to address dating violence through prevention programs/educational programs that target poor minority communities, but it is important to start educating parents about the deleterious effects of corporal punishment, especially among minorities. What brings minority parents to hit their children more often than White parents? Is it part of their culture/customs or is it rather the last resort to exercise control and supervision in communities that do not offer much to youth? As Sokoloff and Pratt (2005) point out, often researchers try to justify certain facts through culture; but culture cannot always explain it all. Perhaps we need to pay more attention to certain patterns of marginalization that originate within a stratified social structure, that tend to affect minority men, women, and children, creating opportunities for social and economic marginalization. These findings on race as risk factor for intimate partner abuse add new evidence to existing controversies in the literature on intimate partner abuse. As reported in Chapter 3, conflicting findings have been reported by major crosssectional and longitudinal studies about the role that racial differences play in the analysis of risk factors for intimate partner abuse (Straus, 1980, 1991; Bachman and Saltzman, 1995; Moffit and Caspi, 1999).

The strong effect of race in the statistical analysis may have washed out the effect of SES. However, in the first part of the analysis, in both the family risk factors model and the socio-cultural risk factors model (where risk factors of intimate partner abuse were considered by category of influence), the statistical findings indicated that being poor increases the chance to abuse a partner for Whites but not for non-Whites. The analysis of the interaction between race and SES needs to be further investigated in studies on risk factors of intimate partner abuse.

Throughout the analysis, this study has consistently found that individuals who entered the romantic relationship at a young age were more likely to become abusive later in life, during adulthood. This was especially true for non-White participants. As in the case of race, these findings contrast existing literature in the field (Frieze, 2005; Rennison, 2001, Cate et al., 1982; Gorman-Smith et al., 2001).

The control variable frequency of alcohol abuse was found not significant throughout the statistical analysis. This contrasts with findings from other studies that pointed to an association between alcohol use/abuse and intimate partner abuse (Dobash and Dobash, 1998; Gelles, 1974).

The final model developed for this dissertation, which was designed by taking into consideration only factors that were significant in the analysis of risk factors by category of influence and in the analysis by age-group, pointed out that among non-White participants, those who entered the romantic relationship at a young age, those who received corporal punishment frequently during early adolescence, and those who were delinquent during pre-adulthood (age 19-22) were far more likely than other participants to become abusive against their intimate partner in adulthood (age 23-25). Similarly, among the White participants, those who entered the romantic relationship at younger age and those who where delinquent during pre-adulthood were more likely to be abusive against their wife/girlfriend in adulthood (age 23-25). Differently from their non-White peers, the White participants involved in conflictual relationship, did not report significant experiences of adolescent corporal punishment.

Because experience of corporal punishment is one of the most important risk factors of adult intimate partner abuse (especially among non-Whites), one can think that the roots of abusive behaviors within intimate contexts are found primarily in the family of origin. The family is a microcosm of our society and the family mirrors our socio-cultural and structural

problems. Non-White families (especially African Americans) are often confined to marginalized neighborhoods and networks that lack resources, income, appropriate schooling, and recreational activities (Carbado, 1999). They are often marked by poverty, degradation, crime, and discrimination. Perhaps we ought to pay more attention to the needs that minority communities have. Hence, we might find that prevention programs, which are needed to buffer the devastating effects of intimate partner abuse, can target the same groups of people and communities that are at risk of other forms of crime.

6.2 Strengths and limitations of the study

A major strength of this dissertation is that it provided an analysis of risk factors that are focus of major controversies in the literature and that needed to be further investigated. Furthermore, by drawing upon three theoretical perspectives (family violence, mainstream feminist, and life-course perspectives), this dissertation has provided two different types of analysis: first, I have investigated risk factors of intimate partner abuse by category of influence (family, socio-cultural, and individual/personal influences) including both past and concurrent factors; then, I have developed an analysis of precursors of intimate partner abuse by age-groups (age 13-15, or early adolescence; age 16-18 or late adolescence; and age 19-22 or pre-adulthood) where only past factors were included in the analysis. This was made possible by the unique features of the PYS; indeed this study includes nearly all the variables suggested by the literature and it also provides researchers with the opportunity to use information gathered during the course of development.

In contrast with most existing studies on intimate partner abuse, this research uses an intimate partner abuse scale that was designed to take into consideration a large number of

behaviors that include not only physical (violent and sexual) abuse but also non-physical controlling behaviors. This is a major innovation in a field where researchers have mostly focused on the use of the original version of the Conflict Tactic Scale, in spite of all the major criticisms and limitations explained by experts in the field (see Chapter 3).

By drawing upon the literature on violence developed by researchers of the PYS (Loeber et al., 2003; Wei, 1999; Wei et al, 2002), this investigation has differentiated between two racial groups, White and non-White, providing the reader with a better understanding of how risk factors of abuse are distributed across racial categories.

Although the PYS is one of the most important longitudinal studies in the US, it also presents some limitations; one of them is that interviews on intimate partner abuse only involved the participants and not their romantic partners (wives/girlfriends). It would certainly be very interesting to compare information gathered from both partners. Also, it is relevant to specify that although the PYS intimate partner abuse questionnaire is very rich compared to other instruments employed in other studies, questions are limited to descriptions of behaviors perpetrated by the participants against their partners (and vice-versa), and do not include details about the situation or the context in which the abuse occurred. In spite of the high level of violence described in several of the PYS questions on intimate partner abuse, the questionnaire does not allow the researcher to investigate whether the victim was injured as a consequence of the abuse. Furthermore, the questionnaire does not include questions that would help researchers to identify who initiated the violence and for what reason. Other limitations refer to the information on the participants' cultural background and beliefs about masculinity, male dominance, and abuse against women. It is certainly relevant to be able to assess the individual's ideological background when investigating the social problem of intimate partner abuse.

The major limitation of the empirical analysis refers to the fact that it would have been more efficient to investigate precursors of intimate partner abuse by using dynamic modeling. Indeed, through dynamic statistical models it is possible to observe how specific transitions during the life-span influence an individual's behaviors. The limitations of the empirical models are also due to the limitations of the PYS data set, which provides information on intimate partner abuse that are comparable only over two different points in time (two phases). Dynamic modeling requires that behaviors are observed for a larger number of years or over a larger number of phases. Another limitation of the analysis is that it does not take into consideration information on the participants' victimization of abuse by an intimate partner. The reason for excluding this variable is based on a technical/statistical problem that was explained in Chapter 4. In addition, the limitations of this research are the same limitations of most studies based on information gathered through survey/empirical methods. By making use of open ended questions, qualitative methods of research can best describe certain details of the situation in which abuse occurs and specific features of individual behaviors that are very relevant to our understanding of the social problem of intimate partner abuse.

6.3 Implications for further research

The analysis of this research has pointed out the importance of making use of all the possible suggestions provided by different theoretical perspectives in order to understand the nature and the causes of intimate partner abuse. Certainly, many other investigations are needed to better interpret the role that certain factors play in determining one's likelihood to abuse an intimate partner. For instance, it is important to investigate why abuse more often occurs among non-Whites. A better understanding of how one's cultural background makes certain individuals

more prone than others to perpetrate abuse against women in general and against intimates in particular needs to be pursued. However, culture will not explain it all. An analysis of the socioeconomic structure is needed in order to investigate at what level the marginalization that non-Whites experience in our society affects their likelihood to become involved in dysfunctional relationships. Qualitative investigations conducted by well cited sociology scholars (see for instance Carbado, 1999) has caused me to doubt about the efficiency of using a variable like the SES to explain the social and economic marginalization of individuals from the non-White communities. There are dynamics of marginalization that follow from one's past experiences (like experience of incarceration) that affect an individual for life. The lack of power and control might not be captured by empirical measures of one's socio-economic status; perhaps, these kinds of dynamics can be best explored by making use of qualitative research.

Researchers should also attempt to understand how perpetration of abuse and experience of abuse by a partner are associated with one another; in other words, it is important to investigate the dual aspect of abuse by interviewing both partners. Perhaps the principal investigators of the PYS might one day decide to also gather information about the occurrence of aggression/abuse from both partners.

Other studies on intimate partner abuse should also attempt to understand the contribution of delinquency to abuse/violence within intimate contexts. Although it seems clear that deviant/delinquent individuals tend to be more prone to abuse their intimate partners, we need to further investigate how specific situational/contextual factors intervene to mediate the association between delinquency and abuse.

It would be very helpful to gather information on intimate partner abuse over several phases (years) so that it would be possible to capture changes within the relationship(s) that

would help us to better explain the nature and the causes of intimate partner abuse. The PYS researchers have investigated abuse within intimate relationships at the age 23-25 years. Other relevant longitudinal studies have focused on intimate partner abuse among couples of younger age (Magdol et al, 1997; Gorman-Smith et al., 2000). It would be interesting to observe abuse within intimate contexts over time in order to understand whether changes in level of commitment, transitions in one's life-span (such as parenthood or divorce), or simply maturation influence participants' behaviors toward intimate partners.

APPENDICES

APPENDIX A

SUPERVISION/INVOLVEMENT QUESTIONNAIRE

Both parents and participants were administered versions of the Supervision/Involvement Scale. Here below I report some of the questions included in the original questionnaire based on the PYS literature (Loeber et al., 1998).

Questions as were addressed to caretakers:

- 1. When was the last time that you discussed with you son his plans for the coming day?
- In the past six months, about how often have you discussed with your son his plans for the coming day?
- 3. When was the last time you talked with your son about what he had actually done during the day?
- 4. In the past six months, about how often have you talked with your son about what he had actually done during the day?
- 5. Does your son have a set time to be home on school nights?
- 6. Does your son have a set time to be home on weekend nights?
- 7. If your son did not come home by the time that was set, would you know?
- 8. If you or another adult are not at home, does your son leave you a note or call you to let you know where he is going?
- 9. Do you know who the companions of your son are when he is not at home?
- 10. When you are not at home, does your son know how to get in touch with you?
- 11. When your son is out, do you know what time he will be home?
- 12. Is it important to you to know what your child is doing when he is outside of the home?
- 13. Where does your child usually go after school?
- 14. When is he usually in the evening?
- 15. Where is he usually on weekends?

- 16. How many of his friends do you know?
- 17. When you and your son are both at home, do you know what he is doing?
- 18. Does your son help to plan family activities?
- 19. Does your son get involved in family activities?
- 20. Do you find time to listen to your son when he wants to talk to you?
- 21. Do you and your son do things together at home?
- 22. Does your son go with members of the family to movies, sports events, or other outings?
- 23. Does your son go with members of the family to church, synagogue, or Sunday school?
- 24. How often do you have a friendly chat or talk with your son?
- 25. Does your son help you?
- 26. Does your son prefer to be with his friends rather than with the family?
- 27. Do you talk with your son about how he is doing in school?
- 28. On the average, how much time each day are you together with your child on weekdays, that is, when you and your child are both awake?
- 29. And, on weekend days?
- 30. On weekdays, how much of that time are you doing something together, like making something, playing a game, talking or going out together?
- 31. And, on weekend days?
- 32. In general, are these activities enjoyable?

APPENDIX B

SELF-REPORTED DELINQUENCY QUESTIONNAIRE

In this Appendix I report some of the questions included in the SRD questionnaire (Loeber et al., 1998).

In the past six months:

- 1. Have you run away from home?
- 2. Have you skipped classes without an excuse?
- 3. Have you lied about age to get into someplace or buy something?
- 4. Have you hitchhiked where it was illegal to do so?
- 5. Have you carried a hidden weapon?
- 6. Have you been rowdy, unruly in a public place so that you got into trouble?
- 7. Have you begged money or things from strangers?
- 8. Have you drunk in a public place?
- 9. Have you purposely damaged other people's property?
- 10. Have you purposely set fire to a house, building, car or other property?
- 11. Have you avoided paying for things?
- 12. Have you broken into a building?
- 13. Have you stolen or tried to steal things worth \$5 or less?
- 14. Have you stolen or tried to steal things worth \$5 to \$50?
- 15. Have you stolen or tried to steal something worth \$50 to \$100?
- 16. Have you stolen or tried to steal something worth \$100 or more?
- 17. Have you taken something from a store without paying for it?
- 18. Have you snatched someone's purse/wallet or picked someone's pocket?
- 19. Have you taken something from a car?
- 20. Have you bought/sold/held stolen goods?

- 21. Have you gone joyriding with someone's vehicle without permission?
- 22. Have you stolen or tried to steal a vehicle/motorcycle?
- 23. Have you used checks illegally?
- 24. Have you used other's credit card or money without their permission?
- 25. Cheated by selling something worthless?
- 26. Attacked someone with a weapon or with the idea to hurt them?
- 27. Hit someone to hurt?
- 28. Used a weapon/force/strong arm to get money or things from people?
- 29. Thrown objects such as rocks or bottles at people?
- 30. Have you been involved in gang fights?
- 31. Have you been paid for sex?
- 32. Have you physically hurt or threatened to hurt someone to have sex?
- 33. Have you tried to have sex with someone against their will?
- 34. Have you sold marijuana or hashish?
- 35. Have you sold hard drugs (heroin, cocaine, or LSD)?
- 36. Have you been arrested or picked up by police for other than minor traffic offenses?

APPENDIX C

BIBLIOGRAPHY

- Achenbach, T.M. (1978), "The Child Behavior Profile: I. Boys Aged 6-11". Journal of Consulting and Clinical Psychology" (46): 478-488.
- Achenbach, T.M., and Edelbrock, C.S. (1983), "Manual for the Child Behavior Checklist and Revised Child Behavior Profile". Burlington, VT: University of Vermont, Department of Psychiatry.
- Achenbach, T.M., and Edelbrock, C.S. (1987), "Manual for the Youth Self-Report and Profile". Burlington, VT: University of Vermont, Department of Psychiatry.
- Afoldy, G. (1987), "Storia Sociale dell'Antica Roma", Ed. Il Mulino, Bologna.
- Agnew, R. (1985), "Social Control and Delinquency: A Longitudinal Test", Criminology 23: 47-61.
- Allen, B. (1992), "Terrorism Tales: Gender and the Fictions of Italian National Identity" in Italica 69 (2): 161-176.
- Archer, J. (2000a), "Sex Differences in Aggression Between Heterosexual Partners: A Meta-Analytic Review". Psychological Bulletin, 126(5): 651-680.
- Archer, J. (2000b), "Sex Differences in Physical Aggression to Partners: A Reply to Frieze (2000), O'Leary (2000), and White, Smith, Koss, and Figueredo (2000). Psychological Bulletin, 126(5): 697-702.
- Austin, J. and Dankwort J. (1998), "A Review of Standards for Batterer Intervention Programs", In Applied Research Forum, National Electronic Network for Violence Against Women.
- Bachman, R. and Saltzaman, L. E. (1995), "Violence Against Women: Estimates from the Redesigned Survey", Bureau of Justice Statistics, Special Report, August 1995, NCJ-154348.
- Bard, M. (1973), "The role of law enforcement in the helping system". In J.R. Snibbe and H. M. Snibbe (Eds.), "The Urban Policeman in Transition" A psychological and Sociological Review (pp. 407-420). Springfield, IL: Charles C. Thomas.
- Baron, R.M. and Kenny, D.A. (1986), "The Moderator-Mediator Variable Distinction in Social Psychological Research: Conceptual, Strategic, and Statistical Considerations". Journal of Personality and Social Psychology, 6:1173-1182.
- Berns, N. (2004), "Framing the Victim: Domestic Violence, Media, and Social Problem", Aldine Transaction Publisher.

Blumstein, A., Cohen, J, Roth J.A., Visher, C.A. (eds.) (1986), "Criminal Careers and 'Career Criminals'. Washington, D.C.: National Academy Press.

Blundell, S. (2001), "Women in Ancient Greece".

Brofenbrenner, U. (1979), "The Ecology of Human Development: Experiments by Nature and Design." Cambridge University Press.

Browne, A. (1987), "When Battered Women Kill", New York: the Free Press.

Brule, P. (2003), "Les Femmes Grecques a l'epoque classique", Edinburgh, UP.

- Brush, L.D. (1990), "Violent Acts and injurious outcomes in married couples: Methodological Issues in the National Survey of Family and Households". Gender and Society, 4(1): 56-67.
- Brush, L.D. (1999), "Woman Battering and Welfare Reform: The view from a Welfare-to-Work Program", Journal of Sociology and Social Welfare, Sept. Vol. XXVI, No. 3 (pp. 49-60).
- Brush, L.D. (2001), "Poverty, Battering, Race, and Welfare Reform: Black-White Differences in Women's Welfare-to-Work Transitions", Journal of Poverty, Vol. 5 (1), pp.67-88.
- Bureau of Justice Statistics (1994), "Violent Crime", Selected Findings, NCJ-147486 <u>http://www.ojp.usdoj.gov/bjs</u>
- Bureau of Justice Statistics (2005), "Family Violence Statistics: Including Statistics on Strangers and Acquaintances", <u>http://www.ojp.usdoj.gov/bjs</u>
- Bureau of Justice Statistics (2006), "Intimate Partner Violence Declined between 1993 and 2004"; Press Release of the Department of Justice, Office of Justice Program, <u>http://www.ojp.usdoj.gov/bjs/pub/press/ipvpr.htm</u>
- Busby, D.M. (1991), "Violence in the Family", in Family Research. A Sixty-Year Review, 1930-1990"; Volume 1, Stephen J. Bahr Ed., Lexington Books.
- Buzawa, E.S. and Buzawa, C.G. (2003), "Domestic Violence: The Criminal Justice Response", Third Edition: Sage Publications.

- Capaldi, D.M. and Clark, S. (1998), "Prospective Family Predictors of Aggression Toward Female Partners for At-Risk Young Men". Developmental Psychology, 34(6): 1175-1188.
- Capaldi, D.M. and Crosby, L. (1997), "Observed and Reported Psychological and Physical Aggression in Young, At-Risk Couples". Social Development, 6(2):184-206.
- Carbado, D. W. (1999), "Black Men on Race, Gender, and Sexuality. A Critical Reader.
- Cate, R.M., Henton, J.M., Koval, J., Christopher, F.S., and Lloyd, S. (1982), "Premarital Abuse. Social Psychological Perspective", Journal of Family Issues, 3(1): 79-90.
- Centers for Disease Control and Prevention, Department of Health and Human Services, Injury Center (2003), "Intimate Partner Violence: Overview", http://www.cdc.gov.ncipc/factsheets/ipvfacts.htm
- Chapple, C.L. (2003), "Examining Intergenerational Violence: Violent Role Modeling or Weak Parental Controls". Violence and Victims, 18: 143-162.
- Chermack, S.T., Walton, M.A., Fuller, B.E., and Blow, F.C. (2001), "Correlates of Expressed and Received Violence Across Relationship Types Among Men and Women Substance Abusers", Psychology of Addictive Behavior, 15 (2): 140-151.
- Chesney-Lind, M. (1999), "Girls' Crime and Woman's Place. Toward a Feminist Model of Female Delinquency", in Pontell, H.N. (ed.), "Social Deviance: Readings in Theory and Research", Prentice Hall, Upper Saddle River, New Jersey.
- Chesney-Lind, M. (2006), "Patriarchy, Crime, and Justice: Feminist Criminology in an Era of Backlash", in Feminist Criminology, 1(1): 6-26.
- Coleman D.H. and Straus M.A. (1983), "Alcohol Abuse and Family Violence". In E. Gottheip, K. Druley, T. Skoloda, and H. Waxman (Eds.), "Alcohol, Drug Abuse, and Aggression" (pp. 104-124), Springfield, IL, Charles C. Thomas.
- Collins, P. H. (2000), "Black Feminist Thought: Knowledge, Consciousness, and the Politics of Empowerment.
- Daniels, C.R. (1997), "Feminists Negotiate the State. The Politics of Domestic Violence". University Press of America.
- Darroch, J.E. and Singh, S. (1999), "Why is Teenage Pregnancy Declining? The Roles of Abstinence, Sexual Activity and Contraceptive Use", The Alan Guttmacher Institute, Occasional Report No.1.
- Department of Health and Human Services, Centers for Disease Control and Prevention (CDCP National Center for Injury Prevention and Control (2003), "Costs of Intimate Partner

Violence Against Women in the United States". Atlanta, Georgia.

- Dixon, L. and Browne, K. (2003), "The Heterogeneity of Spouse Abuse: A Review". Aggression and Violent Behavior 8:107-130.
- Dobash, R.E. and Dobash, R.E. (1978), "Wives: The 'appropriate' victims of marital violence". Victimology 2(3/4):426-42.
- Dobash, R.E. and Dobash, R.E. (1979), "Violence Against Wives: A Case Against the Patriarchy". New York: Free Press.
- Dobash, R.E. and Dobash, R.P. (1979), "Violence Against Wives: A Case Against the Patriarchy", New York: Free Press.
- Dobash R. P., and Dobash R. E. (1986), "The Imprisonment of Women", Bladewell Publication.
- Dobash, R. E. and Dobash, R.P. (1992), "Women, Violence, and Social Change". London: Routledge.
- Dobash, R.E. and Dobash, R.P. (1998), Eds. "Rethinking Violence Against Women". Sage series on Violence Against Women.
- Dobash, R.P., Dobash, R.E., Wilson, M., and Daly, M. (2005), "The Myth of Sexual Symmetry in Marital Violence". In Violence Against Women, Renzetti, C. M. and R.K. Burgen (Editors).
- Downey, G. and Feldman, S. (1996), "Implications of Rejection Sensitivity for Intimate Relationships", Journal of Personality and Social Psychology, 70: 1327-1343.
- Downey, G., and Feldman, S. (1996), "Implications of Rejection Sensitivity for Intimate Relationships", Journal of Personality and Social Psychology, 70: 1327-1343.
- Downs, W.K., Smyth, N.J., and Miller, B.A. (1996), "The Relationship between childhood Violence and Alcohol Problems among Men who Batter: An Empirical Review and Synthesis". Aggressive and Violent Behavior, 1(4): 327-344.
- Dunford, F.W., Huizinga, D. and Elliott, D. (1989), "The Omaha Domestic Violence Police Experiment: Final Report to the National Institute of Justice and the City of Omaha, Boulder, CO: Institute of Behavioral Science.
- Dutton, D.G. (1998), "The Abusive Personality: Violence and Control in Intimate Relationships. New York: Guilford Press.
- Dutton, D.G. and Hart, S.D. (1992), "Risk Markers for Family Violence in a Federally Incarcerated Population". International Journal of Law and Psychiatry, 15: 101-112.

- Dutton, D.G. and Nicholls, T.L. (2005), "The gender paradigm in domestic violence research and theory: Part 1 The conflict of theory and data". Aggression and violent behavior, 10: 680-714.
- Edelbrock, C.S. and Achenbach, T.M. (1984), "The Teacher Version of the Child Behavior Profile: I. Boys Aged Six through Eleven". Journal of Consulting and Clinical Psychology (52): 207-217.
- Edleson, J.L. (1995), "Do Batterers' Programs Work?" Minnesota Center Against Violence and Abuse.
- Ehrensaft, M.K., Cohen, P., Brown, J., Smailes, E., Chen, H., and Johnson J.G. (2003), "Intergenerational Transmission of Partner Violence: A 20-Year Prospective Study", Journal of Consulting and Clinical Psychology, 71 (4): 741-753.
- Elder Glen H., Jr. (1985), "Perspective of the Life Course", in "Life Course Dynamics, Glen Elder H. Jr. Editor, Ithaca, NY: Cornell University Press.
- Elliott, D.S., Huizinga, D., and Ageton, S.S. (1985), "Explaining Delinquency and Drug Use", Beverly Hills, California: Sage Publications.
- Elliott, D.S., Wilson, W.J., Huizinga D., Sampson, R.J., Elliott, A., and Rankin, B., (1996), "The Effects of Neighborhood Disadvantage on Adolescent Development", Journal of Research in Crime and Delinquency 33: 389-426.
- Erikson, E.H. (1994), "Identity and the Life Cycle", W.W. Norton & Company, Inc. ISBN 0393311325.
- Ewing, C.P. (1987), "Battered Women Who Kill: Psychological self-defense as legal justification", Lexington, MA: Lexington Books.
- Ewing, C.P. (1997), "Fatal Families: the Dynamics of Intra-familial homicide, London: Sage Publications.
- Fagan, J. and Browne, A. (1994), "Violence between Spouses and Intimates: Physical Aggression between Women and Men in Intimate Relationships. In A.J. Reiss, Jr. and J.A. Roth (Eds.), Understanding and Preventing Violence: 3, Social Influences: 115-292. Washington, D.C.: National Academy Press.
- Fagan, J.A., Stewart, D.K., and Hansen, K.V. (1981), "Violent Men or Violent Husbands? Background Factors and Situational Correlates", in Finkelhor, D., Gelles, R., Hotaling, G., and Straus, M.A. (Eds.), "The Dark Side of Families: Current Family Violence Research. Beverly Hills, CA: SAGE.
- Fagan, J. and Wexler, S. (1987), "Crime at Home and in the Streets: the Relationship between Family and Stranger Violence". Violence and Victims, 2: 5-23.

- Fanworth, M. and M.J., Leiber (1989), "Strain Theory Revisited: Economic Goals, Educational Means, and Delinquency", American Sociological Review, 54: 263-74.
- Farrington, D.P. (1993), "Motivations for Conduct Disorder and Delinquency", Development and Psychopathology, 5: 225-241.
- Farrington, D.P. (1994), "Early Developmental Prevention of Juvenile Delinquency", Criminal Behavior and Mental Health, 4: 209-227.
- Farrington, D.P. (1994), "Childhood, Adolescent, and Adult Features of Violent Males. In L.R. Huesmann (Ed.), Aggressive Behavior: Current Perspectives: 215-240. N.Y.: Plenum.
- Farrington, D.P. (1997), "Human Development and Criminal Careers". In M. Maguire and R. Reiner (Eds.), The Oxford Handbook of Criminology (2nd Ed.). Oxford: Clarendon Press.
- Farrington, D.P. and West, D.J. (1990), "The Cambridge Study in Delinquent Development: A longterm follow-up of 411 London males. In H-J. Kerner and G. Kaiser (Eds.) Criminality: Personality, Behavior, Life History (115-138). Berlin: Springer-Verlag.
- Farrington, D.P. (1990), "Implications of Criminal Career Research for the Prevention of Offending", Journal of Adolescence, 13: 93-113
- Federal Bureau of Investigation (2004), "Crime in the United States, 2003", Washington D.C.: Government Printing Office.
- Ferraro, K.J. (1989), "Policing women battering". Social Problems, 36, 61-74.
- Ferraro, K.J. (2001), "Woman Battering: More than a Family Problem", in Renzetti, C. and Goodstein, L. (Eds.) "Women, Crime, and Criminal Justice", Los Angeles: Roxbury, 135-153.
- Follingstad, D.R., Bradley, R.G., Helff, C.M., and Laughlin, J.E. (2002), "A Model from Predicting Dating Violence: Anxious Attachment, Angry Temperament, and the Need for Relationship Control". Violence and Victims, 17: 35-47.
- Follingstad, D.R., Rutledge, L.I., Berg, B.J., Hause, E.S., and Polek, D.S. (1990), "The Role Of Emotional Abuse in Physically Abusive Relationships". Journal of Family Violence, 5: 107-120.
- Frieze, I.H. (2005), "Hurting the one we love. Violence in relationships". Thomas Wadsworth.
- Frieze, I.H. (2000), "Violence in Close Relationships Development of a Research Area: Comment on Archer (2000). "Psychological Bulletin, 126(5): 681-684.

- Gagne, P. (1998), "Battered Women's Justice: The Movement for Clemency and the Politics of Self-Defense".
- Gelles, R.J. (1974), "The Violent Home: A Study of Physical Aggression Between Husbands and Wives". Beverly Hills, CA, Sage.
- Gelles, R.J. (1980), "Violence in the Family: A Review of Research in the Seventies", in Journal of Marriage and the Family, November: 873-85.
- Gelles, R.J. and Cavanaugh, M.M. (2005), "Association is not Causation. Alcohol and other Drugs Do not Cause Violence", in D.R. Loseke, R.J. Gelles, and M.M. Cavanaugh Eds., "Current Controversies in Family Violence", pp. 175-190, Sage Publications.
- Gleason, W.J. (1997), "Psychological and Social Dysfunctions in Battering Men: A Review". Aggressive and Violent Behavior, 2(1): 43-52.
- Goetting, A. (1999), "Getting out: Life Stories of Women who Left Abusive Men", Columbia University Press.
- Goldestein, J.S. (2001), "War and Gender: How Gender Shapes the War System and Viceversa", Cambridge University Press.
- Gorman-Smith, D., Tolan, P.H., and Henry, D. (2000), "A Developmental-Ecological Model of the Relation of Family Functioning to Patterns of Delinquency", Journal of Quantitative Criminology, 16: 169-198.
- Gorman-Smith, D., Tolan, P.H., Sheidow, A.J., and Henry, D.B. (2001), "Partner Violence and Street Violence among Urban Adolescents: Do the Same Family Factors Relate?" Journal of Research on Adolescence, 11(3): 273-295.
- Gorman-Smith, D., Tolan, P.H., Zelli, A., and Hesmann, L.R. (1996), "The relation of Family Functioning to Violence among inner-city Minority Youths", Journal of Family Psychology, 10: 115-129.
- Gottfredson, M. and Hirschi, T. (1990), "A General Theory of Crime". Stanford, California: Stanford University Press.
- Greenberg, D.F. (1999), "The Weak Strength of Social Control Theory", Crime and Delinquency, 45(1): 66-81.
- Greenfeld, A. and Snell, T. (1999), "Women Offenders: Bureau of Justice Statistics, Special Report, Washington, D.C.: U.S. Department of Justice.
- Hamberger, L.K. and Guse, C.E. (2002), "Men's and Women's Use of Intimate Partner Violence in Clinical Samples". Violence Against Women, 8(11): 1301-1331.

- Harris, J.R. (1998), "The Nurture Assumption: Why Children Turn Out the Way they Do", New York Free Press.
- Hazelwood, R.R. and Warren, J.I. (2000), "The Sexually Violent Offender: Impulsive or Ritualistic?" Aggression and Violent Behavior, 5 (3): 267-279.
- Healey, K., Smith, C., and Sullivan, C. (1998), "Batterer Intervention: Program Approaches and Criminal Justice Strategies", U.S. Department of Justice, Office of Justice Programs, National Institute of Justice.
- Hicks, M.W. and Platt, M. (1970), "Marital Happiness and Stability: A Review of the Research in the Sixties". In Carlfred B. Broderick (ed.), "A Decade of Family Research and Action": Minneapolis, Minnesota: National Council of Family Relations.
- Hines, D.A. and Saudino, K.J. (2002), "Intergenerational Transmission of Intimate Partner Violence: A Behavioral Genetic Perspective", Trauma, Violence, and Abuse, 3(3): 210-225.
- Hirschi, T. (1969), "Causes of Delinquency". Berkeley and Los Angeles: University of California Press.
- Hollingshead, A.B. (1975), "Four Factor Index of Social Status". New Haven, CT: Unpublished Manuscript.
- Hotaling, G.T. and Sugarman, D.B. (1986), "An Analysis of Risk Markers in Husband to Wife Violence: The Current State of Knowledge. Violence and Victims, 1:101-124.
- Huizinga, D., Wylie Weiher, A., Espiritu, R., and Esbensen, F. (2003), "Delinquency and Crime: Some Highlights from the Denver Youth Survey". In Thornberry, T.P. and Krohn M.D. (eds.), "Taking Stock of Delinquency: An Overview of Findings from Contemporary Longitudinal Studies. New York: Kluwer/Plenum.
- Johnson, I. M. and Sigler, R. T. (2000), "Public perceptions. The Stability of the Public's Endorsements of the Definition and Criminalization of the Abuse of Women". Journal Of Criminal Justice, 28: 165-179.
- Johnson, M.P. (1995), "Patriarchal Terrorism and Common Couple Violence: Two Forms Of Violence Against Women", in Journal of Marriage and the Family, 57 (2): 283-294.
- Kantor, G.K., Jasinski, J.L., and Aldarondo, E. (1994), "Socio-cultural Status and Incidence of Marital Violence in Hispanic Families. Special Issue: Violence against Women of Color". Violence and Victims, 9: 207-222.
- Kaufman Kantor G. and Straus M.A. (1987), "The Drunken Bum Theory of Wife Beating". Social Problems, 34: 213-230.
- Kimmel, M.S. (2002), "Gender Symmetry in Domestic Violence. A Substantive and Methodological Research Review", Violence Against Women, 8 (11): 1332-1363.

- Krueger, R.F., Moffit, T.E., Caspi, A., Bleske, A., Silva, P.A. (1998), "Assortative Mating of Antisocial Behavior: Developmental and Methodological Implications", Behavior Genetics, 28 (3): 173-186.
- Kyriacou, D.N., Anglin, D., Tagliaferro, E., Stone, S., Tubb, T., Linden, J.A., Muelleman, R., Barton, E., and Krauss, J.F. (1999), "Risk Factors for Injury to Women from Domestic Violence". The New England Journal of Medicine, 341 (25):1892-1898.
- Le Blanc, M., and Loeber, R. (1993), "Precursors, Causes and the Development of Criminal Offending", in Hay, D. F., and Angold, A. (Eds.), "Precursors and Causes in Development and Psychopathology", Wiley & Son Ltd.
- Leonard, K.E. and Senchak, M. (1993), "Alcohol and Pre-Marital Aggression among Newlywed Couples". Journal Studies of Alcohol, 11: 96-108.
- Loeber, R. (1982), "The stability of antisocial and delinquent child behavior". Child Development (53): 1431-1446.
- Loeber, R. (1987), "Behavioral Precursors and Accelerators of Delinquency". In W. Buikhuisen and S.A. Mednick (Eds.) Explaining Criminal Behavior: Interdisciplinary Approaches (51-67). Leiden, Netherlands: Brill.
- Loeber, R. and Dishion, T. (1983), "Early Predictors of Male Delinquency: A Review", Psychological Bulletin, 94: 68-99.
- Loeber, R. and Le Blanc M., (1990), "Toward a Developmental Criminology". In M. Tonry and N. Morris (Eds.) Crime and Justice, vol. 12 (375-473). Chicago: University of Chicago Press.
- Loeber, R. and Stouthamer-Loeber, M. (1986), "Family Factors as Correlates and Predictors of Juvenile Conduct Problems and Delinquency. In M. Tonry and N. Morris (Eds.) Crime and Justice, 7: 29-149. Chicago: University of Chicago Press.
- Loeber, R., Farrington, D.P., Stouthamer-Loeber, M., and van Kammen, W.B. (1998), "Antisocial Behavior and Mental Health Problems". Mahwah, NJ: Erlbaum.
- Loeber, R. Farrington, D.P., Stouthamer-Loeber, M., Moffit, T.E., Caspi, A., Raskin White, H., Wei, E.H., and Beyers J.M. (2003), "The Development of Male Offending: Key findings from fourteen years of the Pittsburgh Youth Study. In Thonberry, T.P., and Krohn, M.D., (eds.), "Taking Stock of Delinquency: An Overview of Findings from Contemporary Longitudinal Studies: New York: Kluwer/Plenum
- Loeber, R., Stouthamer-Loeber, M., Van Kammen, W.B., and Farrinton D.P. (1991), "Initiation, Escalation and Desistance in Juvenile Offending and their Correlates", Journal of Criminal Law and Criminology, 82: 36-82.
- Logan, T., Walker, R., and Leukefeld, C. (2001), "Rural, Urban Influenced, and Urban

Differences among Domestic Violence Arrestees in Kentucky". Journal of Interpersonal Violence, 16(2): 266-283.

- Mccoby, E.E., and Jacklin, C.N. (1974), "The Psychology of Sex Differences", Stanford: Stanford University Press.
- Magdol, L., Moffit, T.E., Caspi, A., Newman, D.L., Fagan, J., and Silva, P.A.(1997). Gender Differences in Partner Violence in a Birth Cohort of 21-year-olds: Bridging the gap between clinical and epidemiological approaches", Journal of Consulting and Clinical Psychology, 65: 68-78.
- Magdol, L., Moffit, T.E., Caspi, A., Silva, P.A. (1998), "Developmental Antecedents of Partner Abuse: A Prospective-Longitudinal Study", Journal of Abnormal Psychology, 107 (3): 375-389.
- Massey, D.S, and Denton, N.A. (1993), "American Apartheid: Segregation and the Making of the Urban Underclass, Cambridge, MA: Harvard University Press.
- Maxfield, M.G., and Babbie, E. (2001), "Research Methods for Criminal Justice and Criminology" Wadsworth, Third Edition.
- McGregor, H. (1990), "Domestic Violence: Alcohol and other Distractions a Grassroots". Alcohol and Crime. Australian Institute of Criminology Conference, April 1989.
- Miller, N. (1997), "Domestic Violence Legislation Affecting Police and Prosecutor Responsibilities in the United States: Inferences from a 50-State Review of State Statutory Codes. Alexandria, VA: Institute for Law and Justice.
- Miller, N. (2000), "A Legislative Primer on State Domestic Violence-Related Legislation: A law Enforcement and Prosecution Perspective. Alexandria, VA: Institute for Law and Justice.
- Miller, N. (2004), "Domestic Violence: A Review of State Legislation Defining Police and Prosecution Duties and Powers. Alexandria, VA: Institute for Law and Justice.
- Miller, S.L. (2005), "Victims as Offenders. The Paradox of Women's Violence in Relationships", Rutgers University Press.
- Mills, L.G. (2003), "Insult to Injury. Rethinking Our Responses to Intimate Abuse". Princeton University Press.
- Mirrlees-Black, C. (1999), "Domestic violence: findings from a new British Crime Survey self-completion questionnaire". Home Office Research Study No.191, London: Home Office.
- Moffit, T.E. (1993), "Adolescence-Limited and Life-Course-Persistent Antisocial Behavior: A Developmental Taxonomy", Psychological Review, 100 (4): 674-701.

- Moffit, T.E. (2003), "Life-Course Persistent and Adolescence-Limited Antisocial Behavior: A 10-Year Research Review and a Research Agenda". In Lahey, B.B., Moffit, T.E., and Caspi, A. (eds.), "Causes of Conduct Disorder and Juvenile Delinquency", New York: Guildford.
- Moffit, T.E., and Caspi, A. (1998), "Annotation: Implications of Violence between Intimate Partners for Child Psychologists and Psychiatrists", Journal of Child Psychology and Psychiatry, 39 (2): 137-144.
- Moffit, T.E., and Caspi, A. (1999), "Findings About Partner Violence From the Dunedin Multidisciplinary Health and Development Study". National Institute of Justice, Research in Brief. U.S. Department of Justice.
- Moffit, T.E., Caspi, A., Silva, P.A., and Stouthamer-Loeber, M. (1995), "Individual Differences in Personality and Intelligence are Linked to Crime: Cross-Context Evidence from Nations, Neighborhoods, Genders, Races, and Age-Cohorts". Current Perspective on Aging and the Life Cycle, 4:1-34.
- Moffit, T.E., Krueger, R.F., Caspi, A. and Fagan, J. (2000), "Partner Abuse and General Crime: How Are They The Same? How Are They Different?" Criminology, 38(1):199-232.
- Moos, R.H. and Moos, B.S. (1975), "Evaluating Correctional and Community Settings". In R.H. Moos (Ed), "Families":263-286. New York: Wiley.
- Morse, B.J. (1995), "Beyond the Conflict Tactic Scale: Assessing gender differences in partner Violence". Violence and Victims, 10(4): 257-272.
- Murphy, C.M., O'Farrell, T.J., Fals-Stewart, W., and Feehan, M. (2001), "Correlates of Intimate Partner Violence among Male Alcoholic Patients", Journal of Consulting and Clinical Psychology, 69: 528-540.
- Neidig, P.H., Friedman, D.H., and Collins, B.S. (1986), "Attitudinal Characteristics of Males Who Have Engaged in Spouse Abuse". Journal of Family Violence, 1: 223-233.
- Ohio Domestic Violence Network, website http://www.odvn.org; visited on September 13, 2006.
- O'Leary, K.D. (1988), "Physical Aggression Between Spouses: A Social Learning Theory Perspective". In V.B. Van Hasselt, R.L. Morrison, A.S. Bellack, and M. Hersen (Eds.), Handbook of Family Violence (11-55). New York: Plenum.
- O'Leary, K.D. (1993), "Through a Psychological Lens: Personality Traits, Personality Disorders, and Level of Violence. In R.J. Gelles and D.R. Loseke (Eds.), Current Controversies on Family Violence (7-30). Newbury Park: Sage.
- O'Leary, K.D., Malone, J., and Tyree, A.(1994), "Physical Aggression in Early Marriage: Prerelationships and relationship effects". Journal of Consulting and Clinical Psychology, 62: 594-602.

- O'Leary, K.D. (2000), "Are Women Really More Aggressive than Men in Intimate Relationships? Comment on Archer (2000)". Psychological Bulletin 126(5): 685-689.
- O'Leary, K.D., Malone, J. and Tyree, A. (1994), "Physical Aggression in Early Marriage: Pre-relationship and Relationship Effect". Journal of Consulting and Clinical Psychology, 62: 594-602.
- Osofsky, J.D. (1995), "Children who witness domestic violence: The invisible victims". SRCD Social Report Policy Report, 9:1-16.

Owen, B., (1998), "In the Mix: Struggle and Survival in a Women's Prison".

Pagelow, M.D. (1984), "Family Violence", Praeger Publishers.

- Patterson, G.R. and Stouthamer-Loeber M. (1984), "The Correlation of Family Management Practices and Delinquency". Child Development (55): 1299-1307.
- Pence E., and Paymar M. (1993), "Education Groups for Men who Batter. The Duluth Model", Springer Publishing Company, New York.
- Peeples, F., and Loeber R. (1994), "Do Individual Factors and Neighborhood Context Explain Ethnic Differences in Juvenile Delinquency?" Journal of Quantitative Criminology, 10: 141-158.
- Pleck, E. (1987), "Domestic Tyranny: the Making of Social Policy against Family Violence from Colonial Times to the Present". New York: Oxford University Press.
- RADAR: Respecting Accuracy in Domestic Violence Abuse Reporting (2006), Special Report, "VAWA: Threat to Families, Children, Men, and Women". Web-site www.mediaradar.org, visited on September 13, 2006.
- Rankin, B. H., & Quane, J. M. (2002). Social contexts and urban adolescent outcomes: The interrelated effects of neighborhoods, families, and peers on African-American youth. *Social Problems*, 49, 79–100.
- Raphael, J. (2000), "Saving Bernice. Battered Women, Welfare, and Poverty", Northwestern University Press, Boston.
- Rennison, C.M. (2001), "Intimate Partner Violence and Age of Victim, 1993-99". Bureau Of Justice Statistics, Special Report, NCJ 187635.
- Rennison, C.M. (2003), "Intimate Partner Violence, 1993-2001". Bureau of Justice Statistics, Crime Data Brief, NCJ 197838.
- Rennison, C.M. and Welchans, S. (2000), "Intimate Partner Violence", Bureau of Justice Statistics Special Report. U.S. Department of Justice, Office of Justice Programs.

- Richie, B. (1996), "Compelled to Crime. The Gender Entrapment of Battered Black Women". Routledge Edition, London.
- Rosnow, R.L. and R. Rosenthal (1989), "Statistical Procedures and the Justification of Knowledge in Psychological Science", in American Psychologist, Vol. 44 (10): 1276-1284.
- Sampson, R.J. (1987), "Urban Black Violence: the Effect of Male Joblessness and Family Disruption", American Journal of Sociology, 93 (2): 348-82.
- Sampson, R.J. and J.H. Laub (1990), "Crime and Deviance over the Life Course: the Salience of Adult Social Bonds", American Sociological Review, 55: 609-627.
- Schuller, R.A. and Vidmar, N. (1992), "Battered Woman Syndrome Evidence in the Courtroom". Law and Human Behavior, 16(3): 273-291.
- Schumacher, J.A., Feldban-Kohn, S., Smith Slep, A.M., and Heyman, R.E. (2000), "Risk Factors for Male-to-Female Partner Physical Abuse". Aggression and Violent Behavior 6: 281-352.
- Shaw, C. and McKay H.D. (1942), "Juvenile Delinquency and Urban Areas", Chicago: The University of Chicago Press.
- Sherman L.W. and Berk, R.A. (1984), "The Minneapolis Domestic Violence Experiment". Washington DC: Police Foundation.
- Shupe, A.S. and Hazelwood, L. (1987), "Violent Men: Violent Couples". Lexington MA: Lexington Books.
- Siegel, L.J. (2006), "Criminology", Thomson, Wadsworth, 9th Edition
- Simon, R.L., Lin, K., and Gordon, L.C. (2001), "Socialization in the Family of Origin and Male Dating Violence: A Prospective Study", Journal of Marriage and the Family, 60: 467-478.
- Skinner, H.A., Steinhauer, P.D., and Santa-Barbara, J. (1983), "The Family Assessment Measure." Canadian Journal of Community Mental Health, (2): 91-105.
- Smith Slep, A.M. and Heyman, R.E. (2001), "Where Do We Go from Here? Moving Toward An Integrated Approach to Family Violence". Aggression and Violent Behavior, 6:353-356.
- Sokoloff, N.J, and Pratt C. (Eds.) (2005), "Domestic Violence At The Margin: Readings on Race, Class, Gender, And Culture"; Foreword by B. Richie, Rutgers University Press.
- Spanier, G.B. (1976), "Measuring Dyadic Adjustment: New Scale for Assessing the Quality of Marriage and Similar Dyads". Journal of Marriage and the Family, 38: 15-28.
- Stanko, E.A. (1989), "Missing the Mark? Police Battering". In Hanmer, J. Radford and B. Stanko (Eds.), "Women Policing and Male Violence (pp. 46-9). London: Routledge and Kegan Paul.

- Stanton, B.F., Li, X., Black, M.M., Ricardo, I., Galbraith, J., Feigelman, S., and Kaljee, L. (1996), "Longitudinal Stability and Predictability of Sexual Perceptions, Intentions, and Behaviors Among Early Adolescent African-Americans". Journal of Adolescent Health, 18: 10-19.
- Steffensmeier, D., Zhong, H., Ackerman, J., Schwartz, J. and Agha, S. (2006), "Gender Gap Trends for Violence Crimes, 1980 to 2003", in Feminist Criminology 1(1): 72-98.
- Steinmetz, S.K. (1978), "The Battered Husband Syndrome". Victimology 2:499-509.
- Stocker, M., "Judith, Sexual Warrior: Women and Power in Western Culture.
- Stouthamer-Loeber, M. and Wei, E.H. (1998) "The Precursors of Young Fatherhood and Its Effect on Delinquency of Teenage Males", Journal of Adolescent Health, 22: 56-65.
- Stouthamer-Loeber, M., Loeber, R., Wei, E., Farrington, D.P., and Wikstrom, P.H. (2002), "Risk and Promotive Effects in the Explanation of Persistent Serious Delinquency in Boys", Journal of Consulting and Clinical Psychology, 70 (1): 111-123.
- Straus, M.A. (1991), "New theory and old canards about family violence research". Social Problems, 38 (2):180-197.
- Straus, M.A. (1990). "Ordinary Violence, Child Abuse, and Wife Beating". In M.A. Straus and R.J. Gelles (Eds.), "Physical Violence in American Families: Risk Factors and Adaptations to Violence in 8,145 Families":403-424. New Brunswick: Transaction.
- Straus, M.A. (1999), "The Controversy over Domestic Violence by Women: A Methodological, Theoretical, and Sociology of Science Analysis", in Arriaga, X.B. and Oskamp, B.,(Eds.)"Violence in Intimate Relationships", Thousand Oaks, CA: Sage Publications.
- Straus, M.A. (2005), "Children Should Never, Ever, Be Spanked No Matter What the Circumstances", in D.R. Loseke, R.J. Gelles, M.M. Cavanaugh Eds., "Current Controversies on Family Violence", pp. 137-158, Sage Publications.
- Straus, M.A. and Gelles, R.J. (1986), "Societal Change and Change in Family Violence from 1975 to 1985 as revealed by two National Surveys". Journal of Marriage and the Family, 48:465-479.
- Straus, M.A. and Yodanis, C.L. (1996). "Corporal Punishment in Adolescence and Physical Assaults on Spouses in Later Life: What Accounts for the Link?" Journal of Marriage and the Family, 58: 825-841
- Straus, M.A., Gelles, R., and Steinmetz, S. (1980), "Behind closed doors: Violence in the American family". New York: Anchor Books.
- The Alan Guttmacher Institute, 2002, "Teenagers' Sexual and Reproductive Health", Facts in Brief: New York and Washington. Web site: <u>www.guttmacher.org</u>.

- The National Survey of Families and Households, "Data description and documentation" by J.A. Sweet and L.L. Bumpass, University of Wisconsin. Web site: <u>http://www.ssc.wisc.edu/nsfh/home.htm</u>
- Thornberry, T.P., Lizotte, A.J., Krohn, M.D., Smith, C.A., and Porter, P.K (2003), "Causes and Consequences of Delinquency: Findings from the Rochester Youth Development Study. In Thornberry, T.P. and Krohn, M.D. (eds.), "Taking Stock of Delinquency: An Overview of Findings from Contemporary Longitudinal Studies. New York: Kluwer/Plenum.
- Thompson, M.P., Saltzman, L.E., Johnson, H. (2003), "A Comparison of Risk Factors for Intimate Partner Violence-Related Injury across Two National Surveys on Violence against Women". Violence against Women, 9(4): 438-457.
- Tjaden, P. and Thoennes, N. (1998), "Prevalence, incidence, and consequences of violence Against women: Findings from the National Violence against Women Survey". Washington, DC: US Department of Justice.
- Tjaden, P. and Thoennes, N. (2000), "Prevalence and consequences of male-to-female intimate partner violence as measured by the National Violence against Women Survey". Violence Against Women, 6: 142-161.
- Tolman, R. and Bennett, J. (1990), "A Review of Quantitative Research on Men who Batter". Journal of Interpersonal Violence, 5: 87-118.
- Tolman, P.H., Gorman-Smith, D., Huesmann, L.R., and Zelli, A. (1997), "Assessing Family Processes to Explain Risk for Antisocial Behavior and Depression among Urban Youth", Psychological Assessment, 9: 212-223.
- Tubman, J.G., Windle, M., and Windle, R. (1996), "Cumulative Sexual Intercourse Patterns among Middle Adolescents: Problem Behavior Precursors and Concurrent Health Risk Behaviors". Journal of Adolescent Health, 18: 182-191.
- Walby, S. and Allen, J. (2004), "2001 British Crime Survey". London: Home Office
- Walker, L. (1984), "The Battered Women Syndrome", New York: Springer.
- Walker, L., (1989), "Terrifying Love: Why Battered Women Kill and How Society Responds", New York: Harper and Row Publications, Inc.
- Waltz, J., Babcock, J.C., Jacobson, N.S., and Gottman, M.J. (2000), "Testing a Typology of Batterers", Journal of Consulting and Clinical Psychology, 68 (4): 658-669.

Weedom, C. (1999), "Feminism, Theory, and the Politics of Difference", Blackwell Publishers.

Wei, H.E. (1999) "Teenage Fatherhood and Pregnancy Involvement among Urban Adolescent Males: Risk Factors and Consequences", Unpublished Manuscript – PhD Dissertation, University of Pittsburgh, Pittsburgh (PA).

- Wei, H.E., Loeber, R., and Stouthamer-Loeber, M. (2002), "How many of the offspring born to teenage fathers are produced by repeat serious delinquents?" Criminal Behavior and Mental Health, 12: 83-98.
- Whitbeck, L.B., Yoder, K.A., Hoyt, D.R., and Conger, R.D. (1999), "Early Adolescent Sexual Activity: A Developmental Study", Journal of Marriage and the Family, 61: 934-946.
- White, H.R. and Chen, P.H. (2002), "Problems Drinking and Intimate Partner Violence". Journal Of Studies on Alcohol", 63:205-214.
- White, J.L., Moffit, T.E., Caspi, A., Bartusch, D.J., Needles, D.J., and Stouthamer-Loeber, M. (1994), "Measuring Impulsivity and Examining Its Relationship to Delinquency". Journal of Abnormal Psychology, 103 (2): 192-205.
- Widom, C.S. (1989), "Does violence beget violence? A critical examination of the literature", Psychological Bulletin, 106: 3-28.
- Widom, C.S. and Maxfield, M.G. (2001). "An Update on the "Cycle of Violence". National Institute of Justice in Brief (<u>http://www.ojp.usdoj.gov/nij</u>).
- Wilson, W.J (1987), "The Truly Disadvantaged: the Inner City, the Underclass, and Public Policy", Chicago: the University of Chicago Press.
- Wilson, W.J. (1996), "When Work Disappears: the World of the New Urban Poor", New York: Knopf.
- Wilson, M. and Daly, M. (1998), "Lethal and Non-lethal Violence Against Wives and the Evolutionary Psychology of Male Sexual Proprietariness". In "Rethinking Violence Against Women", Dobash, R. E. and Dobash, R.P. Editors. Sage Series on Violence Against Women.
- Wilt, M. and Bannon, J. (1977), "Domestic Violence and the Police: Studies in Detroit and Kansas City. Washington, DC: Police Foundation.
- Woodward, L.J., Fergusson, D.M. and Horwood, L.J. (2002), "Romantic Relationships of Young People with Childhood and Adolescent Onset Antisocial Behavior Problems", Journal of Abnormal Psychology.
- Wooldredge, J. and Thistlethwaite, A. (2003), "Neighborhood Structure and Race-Specific Rates Of Intimate Assault. Criminology, 41(2): 393-422.
- Zorza, J. (1992), "The Criminal Law of Misdemeanor Domestic Violence, 1970-1990". The Journal of Criminal Law and Criminology, 83(1):46-72.